UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of report (Date of earliest event reported): December 16, 2021

comScore, Inc.

(Exact name of registrant as specified in charter)

Delaware (State or other jurisdiction of incorporation) 001-33520 (Commission File Number) 54-1955550 (IRS Employer Identification No.)

11950 Democracy Drive Suite 600 Reston, Virginia 20190

(Address of principal executive offices, including zip code)

(703) 438–2000

(Registrant's telephone number, including area code)

N/A (Former name or former address, if changed since last report)					
following provisions:	·	sfy the filing obligation of the registrant under any of the			
 □ Written communications pursuant to Rule 425 under th □ Soliciting material pursuant to Rule 14a-12 under the E 	,				
□ Pre-commencement communications pursuant to Rule	,				
□ Pre-commencement communications pursuant to Rule	13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))			
Securities registered pursuant to Section 12(b) of the Act:					
Title of Each Class	Trading Symbol	Name of Each Exchange on Which Registered			
Common Stock, par value \$0.001 per share	SCOR	NASDAQ Global Select Market			
Indicate by check mark whether the registrant is an emerg chapter) or Rule 12b-2 of the Securities Exchange Act of 193 Emerging growth company □		Rule 405 of the Securities Act of 1933 (§230.405 of this			
If an emerging growth company, indicate by check mark if to revised financial accounting standards provided pursuant	the registrant has elected not to use to Section 13(a) of the Exchange A	e the extended transition period for complying with any new ct. \square			

Item 1.01 Entry into a Material Definitive Agreement.

On December 16, 2021, comScore, Inc. (the "Parent") and two newly formed, wholly owned subsidiaries of the Parent entered into an Agreement and Plan of Merger (the "Merger Agreement") with Shareablee, Inc. (the "Company"), two newly formed, wholly owned subsidiaries of the Company, and Shareholder Representative Services LLC, solely in its capacity as the representative, agent and attorney-in-fact of the stockholders of the Company. Pursuant to the Merger Agreement, on December 16, 2021 (the "Closing Date"), the transactions contemplated by the Merger Agreement (the "Transactions") were initiated and, as of the respective effective times of the Transactions, the Parent will acquire the Company for a total purchase price of up to \$45.0 million, subject to certain adjustments and achievement of certain conditions set forth in the Merger Agreement.

At closing, the aggregate consideration payable to the former holders of the Company's capital stock and warrant, and underlying certain equity awards that are to be assumed by the Parent, was approximately 9.0 million shares of the Parent's common stock. The number of shares of common stock issuable at closing was determined by dividing (x) \$36.4 million, plus the exercise price of options that are to be assumed by the Parent, less an estimated working capital adjustment, by (y) a price per share of Parent's common stock of \$4.12. The closing price of Parent's common stock on the Nasdaq Global Select Market on December 16, 2021 was \$3.14 per share.

Certain former holders of the Company's capital stock, warrant and equity award holders (the "Deferred Holders") may also receive up to an aggregate of \$8.6 million of deferred consideration (the "Deferred Amount") over three years after closing of the Transactions, subject to the satisfaction of certain conditions set forth in the Merger Agreement. The Deferred Amount is payable, at the Parent's option, in any combination of cash and the Parent's common stock, with any issuance of common stock to be based on the volume-weighted average trading price of the Parent's common stock for the ten (10) full trading days ending on and including the last business day prior to the applicable date of release of the Deferred Amount.

Following the First Release Date (as defined in the Merger Agreement), which shall be no earlier than the first anniversary of the Closing Date, subject to the satisfaction of certain terms and conditions in the Merger Agreement, the Parent will pay or issue (i) to each of the Deferred Holders (other than the Company's founder and former Chief Executive Officer, Tania Yuki), 50% of their pro rata share of the Deferred Amount and (ii) to Tania Yuki, 33% of her pro rata share of the Deferred Amount. Following the second anniversary of the Closing Date, the Parent will pay or issue (i) to each of the Deferred Holders (other than Tania Yuki), 50% of their pro rata share of the Deferred Amount and (ii) to Tania Yuki, 33% of her pro rata share of the Deferred Amount. Following the third anniversary of the Closing Date, the Parent will pay or issue to Tania Yuki, 33% of her pro rata share of the Deferred Amount. Notwithstanding anything to the contrary, if certain contractual consents are not received within ninety (90) days of the Closing Date, payment of the Deferred Amount shall be subject to additional performance conditions (based on achievement of specified revenue targets) and the Deferred Amount shall be earned or reduced in accordance with the procedures set forth in the Merger Agreement. Any payment or issuance of any Deferred Amount will be reduced by the amounts of any losses alleged under any outstanding indemnification claims at such time, and Tania Yuki will forfeit any rights to the Deferred Amount upon any material breach of her non-competition and non-solicitation agreement entered into in connection with the Transactions, in each case subject to the terms of the Merger Agreement.

The former holders of the Company's capital stock and warrant are not permitted to transfer, directly or indirectly, any of the shares of the Parent's common stock (other than shares issued with respect to the Deferred Amount) acquired by such holders pursuant to the Merger Agreement, other than (i) with respect to each such former holder of the Company's capital stock other than Tania Yuki, (A) 35% following the six (6) month anniversary of the Closing Date, (B) an additional 50% following the twelve (12) month anniversary of the Closing Date, and (C) the remaining 15% following the eighteen (18) month anniversary of the Closing Date, (ii) with respect to Tania Yuki, (A) 25% following the six (6) month anniversary of the Closing Date, (B) an additional 25% following the one (1) year anniversary of the Closing Date, (C) an additional 25% following the two (2) year anniversary of the Closing Date, and (D) the remaining twenty five percent (25%) following the (3) year anniversary of the Closing Date, and (iii) with respect to the former holder of the Company's warrant, 100% following the six (6) month anniversary of the Closing Date. The Deferred Holders are not permitted to transfer, directly or indirectly, any shares issued with respect to the Deferred Amount until the six (6) month anniversary of receipt of any such shares. The transfer restrictions are subject to customary exceptions for affiliate transfers, bona fide estate planning, and transfers to cover applicable taxes.

The Merger Agreement contains customary representations and warranties and covenants and agreements of the parties, and the consummation of the Transactions was subject to the satisfaction or waiver of certain closing deliverables, including, among others, execution of an employee non-competition and non-solicitation agreement by certain key employees of the Company. The Merger Agreement also contains customary indemnification obligations of each of the parties, subject to certain qualifications and thresholds as set forth in the Merger Agreement.

Itzhak Fisher, a member of the Parent's board of directors, is a former director, stockholder and equity award holder of the Company. Mr. Fisher is entitled to receive less than 3% of the aggregate consideration payable under the Merger Agreement.

The foregoing description of the Merger Agreement is qualified in its entirety by reference to the full text of the Merger Agreement, a copy of which is included as Exhibit 10.1 to this Current Report on Form 8-K and is incorporated herein by reference.

Item 3.02 Unregistered Sales of Equity Securities.

The description of the Transactions provided under Item 1.01 of this Current Report on Form 8-K is incorporated into this Item 3.02 by reference.

At closing of the Transactions, the aggregate consideration payable to the former holders of the Company's capital stock and warrant, and underlying certain equity awards that are to be assumed by the Parent, was approximately 9.0 million shares of the Parent's common stock.

The Parent may also elect to issue up to \$8.6 million of additional shares of the Parent's common stock with respect to the Deferred Amount (to the extent earned), based on the volume-weighted average trading price of the Parent's common stock for the ten (10) full trading days ending on and including the last business day prior to the applicable date of release of the Deferred Amount.

The issuance of securities to the former holders of the Company's capital stock and warrant was made, or will be made, in reliance upon an exemption from the registration requirements provided by Section 4(a)(2) of the Securities Act of 1933, as amended (the "Securities Act").

Item 7.01 Regulation FD Disclosure.

On December 17, 2021, the Parent issued a press release announcing the consummation of the Transactions. A copy of the press release is furnished as Exhibit 99.1 hereto and is incorporated herein by reference.

The information in this Item 7.01, including Exhibit 99.1 attached hereto, is being furnished and shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934 (the "Exchange Act"), or otherwise subject to the liabilities of that section, nor shall it be deemed incorporated by reference in any filing under the Securities Act or the Exchange Act, regardless of any general incorporation language in such filing.

Item 9.01 Financial Statements and Exhibits.

Description

(d) Exhibits.

Exhibit No.

Exilibit 110.	Description
10.1*	Agreement and Plan of Merger, dated December 16, 2021, by and among comScore, Inc., SS Media Holdco, LLC, SS Media Merger Sub, Inc., Shareablee, Inc., Shareablee Holdco, Inc., Shareablee Merger Sub, Inc. and Shareholder Representative Services LLC, as Stockholder Representative
99.1	Press release dated December 17, 2021
101.INS	XBRL Instance Document - the instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the Inline XBRL document.
101.SCH	Inline XBRL Taxonomy Extension Schema Document.
101.CAL	Inline XBRL Taxonomy Extension Calculation Linkbase Document.
101.DEF	Inline XBRL Taxonomy Extension Definition Linkbase Document.
101.LAB	Inline XBRL Taxonomy Extension Label Linkbase Document.
101.PRE	Inline XBRL Taxonomy Extension Presentation Linkbase Document.
104	Cover Page Interactive Data File - the cover page iXBRL tags are embedded within the Inline XBRL document

^{*} Schedules and similar attachments have been omitted pursuant to Item 601(a)(5) of Regulation S-K and Instruction 4 to Item 1.01 of Form 8-K. Specific terms in this exhibit (indicated therein by asterisks) have been omitted pursuant to Instruction 6 to Item 1.01 of Form 8-K because such terms are both not material and would likely cause competitive harm to the registrant if publicly disclosed. The registrant undertakes to furnish supplementally copies of any of the omitted schedules, exhibits and terms upon request by the U.S. Securities and Exchange Commission.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

comScore, Inc.

By: /s/ William P. Livek

William P. Livek

Chief Executive Officer and Executive Vice Chairman

Date: December 17, 2021

AGREEMENT AND PLAN OF MERGER
BY AND AMONG
COMSCORE, INC.,
SS MEDIA HOLDCO, LLC,
SS MEDIA MERGER SUB, INC.,
SHAREABLEE, INC.,
SHAREABLEE HOLDCO, INC.,
SHAREABLEE MERGER SUB, INC.,

AND

SHAREHOLDER REPRESENTATIVE SERVICES LLC, AS STOCKHOLDER REPRESENTATIVE

December 16, 2021

TABLE OF CONTENTS

			Page
ART	ICLE I	THE MERGERS AND LLC CONVERSION	<u>2</u>
	1.1	The Combination Transactions	<u>2</u>
	1.2	Effectiveness of the Shareablee Merger	<u>2</u>
	1.3	Effectiveness of the Shareablee LLC Conversion	<u>2</u>
	1.4	Effectiveness of the Initial Merger	<u>2</u>
	1.5	Effectiveness of the Final Merger	<u>2</u>
	1.6	Closing	<u>3</u>
	1.7	Closing Deliverables	<u>3</u>
	1.8	PPP Loan Matters	<u>3</u>
	1.9	Taking of Necessary Action; Further Action	<u>3</u>
ART	ICLE I	I EFFECT OF THE COMBINATION TRANSACTIONS ON THE CAPITAL STOCK	<u>3</u>
		OF THE COMPANY AND SHAREABLEE HOLDCO	
	2.1	Effect of the Combination Transactions on the Capital Stock of the Company, Shareablee Merger Sub, Shareablee Holdco and Parent Holdco LLC	<u>3</u>
	2.2	Effect of the Combination Transactions on the Treatment of Warrant of the Company and Shareablee Holdco	<u>4</u>
	2.3	Effect of the Combination Transactions on the Treatment of Equity Awards of the Company and Shareablee Holdco	<u>5</u>
	2.4	Allocation Schedule	<u>6</u>
	2.5	Surrender and Share Issuance Procedures	<u>7</u>
	2.6	Deferred Amount	<u>8</u>
	2.7 2.8	No Fractional Shares Deferred Amount Grants	10 10
	2.9	Withholding	<u>10</u>
	2.10	Post-Closing Reconciliation	<u>11</u>
ART	ICLE I	II REPRESENTATIONS AND WARRANTIES OF THE COMPANY PARTIES	<u>13</u>
	3.1	Organization	<u>13</u>
	3.2	Company Capital Structure	<u>14</u>
	3.3	No Other Subsidiaries; Ownership Interests	<u>16</u>
	3.4	Authority and Enforceability	<u>16</u>
	3.5	No Conflict	<u>17</u>
	3.6	No Prior Activities of Shareablee Holdco and Shareablee Merger Sub	<u>18</u>
	3.7	Company Financial Statements: No Undisclosed Liabilities	18

	3.8	Tax Matters	<u>19</u>
	3.9	Compliance with Securities Act	<u>21</u>
	3.10	Title to Properties; Absence of Liens; Condition and Sufficiency of Assets	<u>21</u>
	3.11	Intellectual Property	<u>22</u>
	3.12	Agreements, Contracts and Commitments	<u>29</u>
	3.13	Interested Party Transactions	<u>31</u>
	3.14	Company Authorizations	<u>31</u>
	3.15	Litigation	<u>31</u>
	3.16	Books and Records	<u>32</u>
	3.17	Environmental, Health and Safety Matters	<u>32</u>
	3.18	Brokers' and Finders' Fees	<u>32</u>
	3.19	Employee Benefit Plans and Compensation	<u>32</u>
	3.20	Insurance	<u>36</u>
	3.21	Compliance with Laws	<u>36</u>
	3.22	Export Control and Sanctions Laws	<u>36</u>
	3.23	Anti-Corruption Laws	<u>36</u>
	3.24	Due Diligence	<u>37</u>
	3.25	No Other Representations or Warranties	<u>37</u>
ART	ICLE I	V REPRESENTATIONS AND WARRANTIES OF THE PARENT PARTIES	<u>37</u>
	4.1	Organization	<u>37</u>
	4.2	Authority and Enforceability	<u>37</u>
	4.3	Capitalization	<u>38</u>
	4.4	No Conflict	<u>38</u>
	4.5	Litigation	<u>38</u>
	4.6	No Prior Activities of Parent Holdco LLC and Parent Merger Sub	<u>38</u>
	4.7	Financial Statements	<u>38</u>
	4.8	NASDAQ Compliance	<u>39</u>
	4.9	Brokers' and Finders' Fees	<u>40</u>
	4.10	Form S-8 Registration Statement	<u>40</u>
	4.11	Due Diligence	<u>40</u>
	4.12	No Other Representations or Warranties	<u>40</u>
ART	ICLE V	V STOCKHOLDER AND TAX MATTERS	<u>40</u>
	5.1	Company Board Recommendation	<u>40</u>
	5.2	Tax Matters	<u>40</u>

	5.3	Indemnification of Directors and Officers of the Company; Insurance	<u>41</u>
	5.4	Securities Law Exemption; Transfer Restrictions; Stop-Transfer Instructions, Securities Law Compliance	<u>42</u>
ART	ICLE '	VI ADDITIONAL AGREEMENTS	<u>44</u>
	6.1	Conduct of the Business of the Company	<u>44</u>
	6.2	Public Disclosure	<u>44</u>
	6.3	Commercially Reasonable Efforts	<u>44</u>
	6.4	Exchange Documentation	<u>44</u>
	6.5	Confidentiality	<u>44</u>
	6.6	Employee Benefits Covenant	<u>45</u>
ART	'ICLE	VII SURVIVAL OF REPRESENTATIONS AND WARRANTIES; INDEMNIFICATION	<u>45</u>
	7.1	Survival	<u>45</u>
	7.2	Indemnification	<u>46</u>
	7.3	Certain Limitations	<u>47</u>
	7.4	Claims for Indemnification; Resolution of Conflicts	<u>48</u>
	7.5	Third Party Claims	<u>49</u>
	7.6	Stockholder Representative	<u>50</u>
	7.7	Treatment of Payments	<u>51</u>
ART	ICLE '	VIII TERMINATION, EFFECT OF TERMINATION	<u>51</u>
	8.1	Termination	<u>51</u>
	8.2	Effect of Termination	<u>51</u>
ART	ICLE 1	IX GENERAL PROVISIONS	<u>52</u>
	9.1	Notices	<u>52</u>
	9.2	Interpretation	<u>52</u>
	9.3	Counterparts	<u>53</u>
	9.4	Entire Agreement; Assignment	<u>53</u>
	9.5	Severability	<u>53</u>
	9.6	Waiver	<u>53</u>
	9.7	Amendment	<u>53</u>
	9.8	Specific Performance	<u>54</u>
	9.9	Other Remedies	<u>54</u>
	9.10	Governing Law	<u>54</u>
	9.11	Exclusive Jurisdiction; Waiver of Jury Trial	<u>54</u>
	9.12	Rules of Construction	<u>54</u>
	9.13	Acknowledgments	<u>55</u>
	O 1 1	Dalassa her Campaner Staalthaldows and Campaner Warmanthaldor	

<u>57</u>

No Third Party Beneficiaries 9.15

Annex

A Certain Defined Terms

Schedules (Omitted)

- A Disclosure Schedule
 B Allocation Schedule
 C-1 Data Partners
 C-2 Data Consent Requirements
 D Escrow Payments
 E Revenue Hurdle Considerations
 F Revenue Policies and Procedures

AGREEMENT AND PLAN OF MERGER

This Agreement and Plan of Merger (this "Agreement") is made and entered into as of December 16, 2021, by and among comScore, Inc., a Delaware corporation ("Parent"), SS Media Holdco, LLC, a Delaware limited liability company and a direct wholly-owned Subsidiary of Parent ("Parent Holdco LLC"), SS Media Merger Sub, Inc., a Delaware corporation and a direct wholly-owned Subsidiary of Parent ("Parent Merger Sub"), Shareablee, Inc., a Delaware corporation (the "Company"), Shareablee Holdco, Inc., a Delaware corporation and a direct wholly-owned Subsidiary of the Company ("Shareablee Holdco"), Shareablee Merger Sub, Inc., a Delaware corporation and a direct wholly-owned Subsidiary of Shareablee Holdco ("Shareablee Merger Sub"), and Shareholder Representative Services LLC, a Colorado limited liability company, solely in its capacity as the representative, agent and attorney-in-fact of the Company Stockholders (the "Stockholder Representative," each a "Party" and collectively, the "Parties"). Capitalized terms used but not otherwise defined in this Agreement shall have the respective meanings ascribed to them in Annex A.

RECITALS

Whereas, the board of directors of Parent (the "*Parent Board*") has (a) determined that this Agreement and the transactions contemplated by this Agreement, on the terms and subject to the conditions set forth in this Agreement, and the issuance of Parent Common Stock to the Shareablee Holdco Stockholders and Shareablee Holdco Warrantholder pursuant to this Agreement (the "*Parent Stock Issuance*"), are advisable and in the best interests of Parent and its stockholders, and (b) approved, adopted and declared advisable this Agreement and the transactions contemplated hereby, including the Parent Stock Issuance;

Whereas, the board of directors of the Company (the "Company Board") has (a) determined that this Agreement and the transactions contemplated hereby, including the Shareablee Merger, the Shareablee LLC Conversion, the Initial Merger and the Final Merger (the "Combination Transactions"), are advisable and fair to, and in the best interests of, the Company and the Company Stockholders, (b) approved, adopted and declared advisable this Agreement and the transactions contemplated hereby, including the Combination Transactions, (c) directed that this Agreement be submitted to the Company Stockholders for its adoption immediately after the execution and delivery of this Agreement, and (d) recommended that this Agreement be adopted by the Company Stockholders;

Whereas, for U.S. federal income Tax purposes, it is intended that (i) the Shareablee Merger and the Shareablee LLC Conversion, taken together, qualify as a "reorganization" within the meaning of Section 368(a)(1)(F) of the Internal Revenue Code of 1986, as amended (the "*Code*"), (ii) the Initial Merger and the Final Merger, taken together, constitute an integrated plan described in Revenue Ruling 2001-46, 2001-2 C.B. 321 (the "*Integrated Transaction*"), (iii) the Integrated Transaction qualify as a "reorganization" within the meaning of Section 368(a) of the Code (clauses (i)-(iii), the "*Intended Tax Treatment*") and (iv) this Agreement be, and is hereby adopted as, a "plan of reorganization" for purposes of Sections 354, 361 and 368 of the Code and within the meaning of Treasury Regulations Section 1.368-2(g);

Whereas, immediately after the execution and delivery of this Agreement, it is anticipated that the Company shall deliver to Parent, Parent Holdco LLC and Parent Merger Sub the written consent of all of the Company Stockholders (the "Company Written Consent");

Whereas, in connection with the execution of this Agreement and as an inducement for Parent to enter into this Agreement, (a) the Key Employees have executed an Employee Non-Competition and Non-Solicitation Agreement, (b) each of the Company Stockholders has executed a Joinder Agreement and Suitability Documentation, and (c) Tania Yuki Mustapic has executed the Stockholder Non-Competition and Non-Solicitation Agreement; and

Now, Therefore, in consideration of the mutual agreements, covenants and other premises set forth herein, the mutual benefits to be gained by the performance thereof, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged and accepted, the Parties agree as follows:

ARTICLE I THE MERGERS AND LLC CONVERSION

1.1. The Combination Transactions.

- (a) **The Shareablee Merger**. At the Shareablee Merger Effective Time and subject to and upon the terms and conditions of this Agreement and the applicable provisions of the General Corporation Law of the State of Delaware (the "**DGCL**"), Shareablee Merger Sub shall be merged with and into the Company (the "**Shareablee Merger**"), the separate corporate existence of Shareablee Merger Sub shall cease, and the Company shall continue as the surviving corporation (the "**Shareablee Merger Surviving Corporation**") in the Shareablee Merger and shall succeed to an assume all the rights, privileges, immunities, properties, powers and franchises of Shareablee Merger Sub in accordance with the DGCL.
- (b) **The Shareablee LLC Conversion**. At the Shareablee Conversion Effective Time and subject to and upon the terms and conditions of this Agreement and the applicable provisions of the DGCL and the Limited Liability Company Act of the State of Delaware (the "**DLLCA**"), the Shareablee Merger Surviving Corporation shall be converted (the "**Shareablee LLC**"), for all purposes under the DGCL and DLLCA, shall be deemed to be the same entity as the Shareablee Merger Surviving Corporation.
- (c) **The Initial Merger**. At the Initial Merger Effective Time and subject to and upon the terms and conditions of this Agreement and the applicable provisions of the DGCL, Parent Merger Sub shall be merged with and into Shareablee Holdco (the "*Initial Merger*"), the separate corporate existence of Parent Merger Sub shall cease, and Shareablee Holdco shall continue as the surviving corporation (the "*Initial Merger Surviving Corporation*") in the Initial Merger and shall succeed to an assume all the rights, privileges, immunities, properties, powers and franchises of Parent Merger Sub in accordance with the DGCL.
- (d) **The Final Merger**. At the Final Merger Effective Time and subject to and upon the terms and conditions of this Agreement and the applicable provisions of the DGCL and DLLCA, the Initial Merger Surviving Corporation shall be merged with and into Parent Holdco LLC (the "*Final Merger*"), the separate corporate existence of the Initial Merger Surviving Corporation shall cease, and Parent Holdco LLC shall continue as the surviving corporation (the "*Final Surviving Company*") in the Initial Merger and shall succeed to an assume all the rights, privileges, immunities, properties, powers and franchises of the Initial Merger Surviving Corporation in accordance with the DGCL and DLLCA.
- (e) **Resulting Structure**. As a result of the foregoing transactions, (i) Parent Holdco LLC (or the "Final Surviving Company") shall be a direct wholly-owned Subsidiary of Parent, (ii) Shareablee LLC (or the "Shareablee Merger Surviving Corporation") shall be a direct wholly-owned Subsidiary of Parent Holdco LLC, (iii) Shareablee Merger Sub shall cease to exist, (iv) Parent Merger Sub shall cease to exist and (v) Shareablee Holdco shall cease to exist.

1.2. Effectiveness of the Shareablee Merger.

- (a) **Shareablee Merger Effective Time**. Subject to the provisions of this Agreement, as soon as practicable on the Closing Date, the applicable Parties shall cause the Shareablee Merger to be consummated by filing a certificate of merger (the "**Shareablee Merger Certificate**") with the Secretary of State of the State of Delaware, in accordance with the terms and conditions of the DGCL. The Shareablee Merger shall become effective at 11:58 p.m. New York City time on the Closing Date or at such other time as Parent and the Company shall agree and specify in the Shareablee Merger Certificate; **provided**, that such other time shall be before the Shareablee Conversion Effective Time (such time as the Shareablee Merger becomes effective being the "**Shareablee Merger Effective Time**"). The Shareablee Merger shall have the effects set forth in this Agreement, the Shareablee Merger Certificate and the applicable provisions of the DGCL.
- (b) **Organizational Documents**. The certificate of incorporation and bylaws of the Company in effect immediately prior to the Shareablee Merger Effective Time shall remain the certificate of incorporation and bylaws of the Shareablee Merger Surviving Corporation, until thereafter amended in accordance with applicable Law.
- (c) **Officers and Directors**. The officers and directors of the Company immediately prior to the Shareablee Merger Effective Time shall be the officers and directors of the Shareablee Merger Surviving

Corporation, until the earlier of their resignation or removal or until their respective successors are duly appointed or elected and qualified, as the case may be.

1.3. Effectiveness of the Shareablee LLC Conversion.

(a) **Shareablee Conversion Effective Time**. Subject to the provisions of this Agreement, as soon as practicable on the Closing Date, the applicable Parties shall cause the Shareablee LLC Conversion to be consummated by filing a certificate of conversion (the "**Shareablee Conversion Certificate**") with the Secretary of State of the State of Delaware, in accordance with the terms and conditions of the DGCL and DLLCA. The Shareablee LLC Conversion shall become effective at 11:59 p.m. New York City time on the Closing Date or at such other time as Parent and the Company shall agree and specify in the Shareablee Conversion Certificate; **provided**, that such other time shall be after the Shareablee Merger Effective Time and before the Initial Merger Effective Time (such time as the Shareablee LLC Conversion becomes effective being the "**Shareablee Conversion Effective Time**"). The Shareablee LLC Conversion shall have the effects set forth in this Agreement, the plan of conversion as agreed by the Parties, the Shareablee Conversion Certificate and the applicable provisions of the DGCL and DLLCA.

1.4. Effectiveness of the Initial Merger.

- (a) **Initial Merger Effective Time**. Subject to the provisions of this Agreement, as soon as practicable on the Closing Date, the Parties shall cause the Initial Merger to be consummated by filing a certificate of merger (the "**Initial Merger Certificate**") with the Secretary of State of the State of Delaware, in accordance with the terms and conditions of the DGCL. The Initial Merger shall become effective at 12:01 a.m. New York City time on the day immediately following the Closing Date or at such other time as Parent and the Company shall agree and specify in the Initial Merger Certificate; *provided*, that such other time shall be after the Shareablee Conversion Effective Time and before the Final Merger Effective Time (such time as the Initial Merger becomes effective being the "**Initial Merger Effective Time**").
- (b) **Organizational Documents**. The certificate of incorporation and bylaws of Shareablee Holdco in effect immediately prior to the Initial Merger Effective Time shall remain the certificate of incorporation and bylaws of Initial Merger Surviving Corporation as of the Initial Merger Effective Time, in each case until thereafter amended in accordance with applicable Law.
- (c) **Officers and Directors**. The officers and directors of Parent Merger Sub immediately prior to the Initial Merger Effective Time shall be the officers and directors of the Initial Merger Surviving Corporation, until the earlier of their resignation or removal or until their respective successors are duly appointed or elected and qualified, as the case may be.

1.5. Effectiveness of the Final Merger.

- (a) **Final Merger Effective Time**. Subject to the provisions of this Agreement, as soon as practicable on the Closing Date, the Parties shall cause the Final Merger to be consummated by filing a certificate of merger (the "**Final Merger Certificate**") with the Secretary of State of the State of Delaware, in accordance with the terms and conditions of the DGCL and DLLCA. The Final Merger shall become effective at 12:02 a.m. New York City time on the day immediately following the Closing Date or at such other time as Parent and the Company shall agree and specify in the Final Merger Certificate; *provided*, that such other time shall be after the Initial Merger Effective Time (such time as the Final Merger becomes effective being the "**Final Merger Effective Time**"). The Final Merger shall have the effects set forth in this Agreement and the applicable provisions of the DGCL and DLLCA.
- (b) **Organizational Document**. At the Final Merger Effective Time, the certificate of formation and limited liability company agreement of Parent Holdco LLC immediately prior to the Final Merger Effective Time shall remain the certificate of formation and limited liability company agreement of Parent Holdco LLC, in each case until thereafter amended in accordance with applicable Law.
- (c) **Officers and Managers**. The officers and managers of Parent Holdco LLC immediately prior to the Final Merger Effective Time shall be the officers and managers of the Final Surviving Company, until the earlier of their resignation or removal or until their respective successors are duly elected and qualified, as the case may be.

1.6. Closing. The closing of the Combination Transactions (the "*Closing*") shall take place on the date hereof or on such other date as Parent and the Company may agree in writing, subject to, and immediately following, receipt of the Company Written Consent. The date on which the Closing actually occurs shall be referred to herein as the "*Closing Date*."

1.7. Closing Deliverables.

- (a) At or prior to the Closing, as applicable, the Company shall deliver (or cause to be delivered) to Parent:
- (i) evidence reasonably satisfactory to Parent that any and all Company Employee Plans intended to include a Code Section 401(k) arrangement have been terminated pursuant to resolution of the Company (the form and substance of which shall have been subject to review and approval by Parent, which approval shall not be unreasonably withheld, conditioned or delayed), effective as of no later than the day immediately preceding the Closing Date;
- (ii) a duly executed resignation letter, from each of the officers and directors of the Company effective as of the Initial Merger Effective Time in their capacity as officers and directors;
 - (iii) a copy of the Company Written Consent and the Company Board Resolutions;
- (iv) (A) executed payoff letters (including Tax Forms) in forms reasonably satisfactory to Parent, in each case dated no more than three (3) Business Days prior to the Closing Date, with respect to all Indebtedness of the Company set forth on Section 3.2(k) of the Disclosure Schedule owed to the lender thereof and the amounts payable to such lender providing (other than the PPP Loan) for (x) the full and final satisfaction of such Indebtedness as of the Closing Date and (y) the termination and release of any Liens related thereto, and (B) an invoice (including Tax Forms) from each advisor or other service provider to the Company (other than any employee, director or officer of the Company), in each case dated no more than three (3) Business Days prior to the Closing Date, with respect to all Transaction Expenses estimated to be due and payable to such advisor or other service provider as of the Closing Date;
- (v) a properly executed statement, in accordance with U.S. Treasury Regulation Sections 1.897-2(h) and 1.1445-2(c)(3), certifying that Shareablee Holdco is not, and has not been, a "United States real property holding corporation" (as defined in Section 897(c)(2) of the Code) during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code, together with the required notice to the IRS and written authorization for Parent to deliver such notice and a copy of such statement to the IRS on behalf of Shareablee Holdco upon the Closing (the "FIRPTA Compliance Certificate");
- (vi) Joinder Agreements duly executed by each Company Stockholder and the Company Warrantholder (in each case in the forms agreed by the Parties), and, in each case, the Company, and all such Joinder Agreements shall be in full force and effect;
- (vii) Employee Non-Competition and Non-Solicitation Agreements duly executed by the Key Employees, and all such agreements shall be in full force and effect;
- (viii) Stockholder Non-Competition and Non-Solicitation Agreement duly executed by Tania Yuki Mustapic, and such agreement shall be in full force and effect;
- (ix) Evidence of termination of that certain Employment Agreement, dated September 1, 2016, by and between the Company and Tania Yuki Mustapic;
- (x) Evidence of termination of that certain Consulting Services Agreement, dated March 20, 2017, by and between the Company and Advanced Motion Technologies, Inc.;
- (xi) Certificate of good standing for the Company, dated as of a date not more than five (5) Business Days prior to the Closing Date; and
- (xii) evidence reasonably acceptable to Parent that the Company has deposited the PPP Loan Deposit Amount with the PPP Lender in accordance with <u>Section 1.8(a)</u>.
 - (b) At or prior to the Closing, Parent shall deliver (or cause to be delivered) to the Company:
- (i) a duly executed counterpart to the Exchange Agent Agreement from Parent and the Exchange Agent;
- (ii) Employee Non-Competition and Non-Solicitation Agreements, and all such agreements shall be in full force and effect;
- (iii) Stockholder Non-Competition and Non-Solicitation Agreement duly executed by Parent, and such agreement shall be in full force and effect; and
- (iv) Certificate of good standing for Parent, dated as of a date not more than five (5) Business Days prior to the Closing Date.

1.8. PPP Loan Matters.

(a) Prior to the Closing, the Company shall have deposited an aggregate amount of \$849,866.04 (the "**PPP Loan Deposit Amount**") in a deposit account pursuant to an escrow agreement (the "**PPP Loan Escrow Agreement**") by and

among Bank of America, N.A., in its capacity as the lender of the PPP Loan (the "**PPP Lender**"), the PPP Lender, in its capacity as the escrow agent depository, and the Company. Pursuant to the escrow agreement, the PPP Lender shall hold the PPP Loan Deposit Amount as a cash security pending consideration of the Company's application for forgiveness of the PPP Loan. From and after the Closing, no Party shall take any action with respect to the PPP Loan Deposit Amount, the escrow agreement or the PPP Loan forgiveness application, without the prior written consent of Parent and the Stockholder Representative. The PPP Loan Deposit Amount will remain in escrow with the PPP Lender and will only be released from escrow pursuant to the terms of the PPP Loan Escrow Agreement.

(b) Within five (5) Business Days of receipt by the Final Surviving Company of that portion of the PPP Loan, if any (such portion, the "Forgiven PPP Loan Amount"), released from the escrow pursuant to the terms of the PPP Loan Escrow Agreement, and only to the extent of the PPP Forgiven Loan Amount, the Final Surviving Company shall, or shall cause Parent on its behalf to, make the payments set forth on Schedule D (subject to and in accordance with Schedule D) (the "Escrow Payments"). The Final Surviving Company shall have no obligation to pay the Escrow Payments to the extent the Escrow Payments exceed the PPP Forgiven Loan Amount. In the event that the PPP Forgiven Loan Amount is in excess of the Escrow Payments, Parent, on behalf of the Final Surviving Company, shall promptly cause the Exchange Agent to distribute any such excess amount (the amount of such excess, minus any fees and expenses of the Exchange Agent, the "Excess Escrow Payments") to the As-Converted Stockholders based on their Pro Rata Share, and thereafter neither Parent nor the Final Surviving Company shall have any further obligation with respect to the PPP Forgiven Loan Amount.

1.9. Taking of Necessary Action; Further Action

If at any time after the Final Merger Effective Time, any further action is necessary or desirable to carry out the purposes of this Agreement and to vest the Final Surviving Company with full right, title and possession to all assets, property, rights, privileges, powers and franchises of the Company, Parent, Parent Holdco LLC, Shareablee LLC, and the officers and directors of Parent, Parent Holdco LLC and Shareablee LLC are fully authorized in the name of their respective corporations or otherwise to take, and shall take, all such lawful and necessary action.

ARTICLE II EFFECT OF THE COMBINATION TRANSACTIONS ON THE CAPITAL STOCK OF THE COMPANY AND SHAREABLEE HOLDCO

- 2.1. Effect of the Combination Transactions on the Capital Stock of the Company, Shareablee Merger Sub, Shareablee Holdco and Parent Holdco LLC.
- (a) **Conversion of Company Capital Stock**. As of the Shareablee Merger Effective Time, by virtue of the Shareablee Merger and without any action on the part of the Company, Shareablee Merger Sub, the holders of any shares of Company Capital Stock or any other Person, upon the terms and conditions of this Agreement (including upon and subject to the delivery of duly executed Exchange Documentation in the

manner set forth in Section 2.5(a)),, each (i) issued and outstanding share of Company Common Stock shall be converted into the right to receive one fully paid and nonassessable share of Shareablee Holdco Common Stock, (ii) each issued and outstanding share of Company Series A Preferred Stock shall be converted into the right to receive one fully paid and nonassessable share of Shareablee Holdco Series A Preferred Stock, and (iii) each issued and outstanding share of Company Series A-2 Preferred Stock shall be converted into the right to receive one fully paid and nonassessable share of Shareablee Holdco Series A-2 Preferred Stock. As of the Shareablee Merger Effective Time, all such shares of Company Capital Stock shall no longer be outstanding and shall automatically be canceled and retired and shall cease to exist.

- (b) **Treasury Stock**. At the Shareablee Merger Effective Time, by virtue of the Shareablee Merger, each share of Company Capital Stock held by the Company immediately prior to the Shareablee Merger Effective Time shall be cancelled and extinguished without any conversion thereof or the right to receive any consideration therefor. At the Initial Merger Effective Time, by virtue of the Initial Merger, each share of Shareablee Holdco Capital Stock held by Shareablee Holdco or any of its subsidiaries immediately prior to the Initial Merger Effective Time shall be cancelled and extinguished without any conversion thereof or the right to receive any consideration therefor.
- (c) **Conversion of Shareablee Merger Sub Capital Stock**. Each share of common stock of Shareablee Merger Sub issued and outstanding immediately prior to the Shareablee Merger Effective Time shall be converted into one fully paid and nonassessable share of common stock, par value \$0.001 per share, of the Shareablee Merger Surviving Corporation.
- (d) Conversion of Shareablee Holdco Capital Stock. As of the Initial Merger Effective Time, by virtue of the Initial Merger and without any action on the part of Parent, Parent Holdco LLC, Shareablee Holdco, any holder of interests in Shareablee Holdco or any other Person, upon the terms and conditions of this Agreement, including the provisions set forth in this Article VII, each share of Shareablee Holdco Capital Stock that is issued and outstanding immediately prior to the Initial Merger Effective Time shall be cancelled and extinguished, and shall be converted automatically, upon and subject to the delivery of duly executed Exchange Documentation in the manner set forth in Section 2.5(a), into (A) in accordance with Section 2.5, the right of (i) each Shareablee Holdco Series A-2 Preferred Stockholder to receive at the Closing, the Per Share Series A-2 Consideration, and (ii) each Shareablee Holdco Series A Preferred Stockholder and Shareablee Holdco Common Stockholder to receive at the Closing, the Per Fully Diluted Share Closing Consideration, and (B) the contingent right of such Shareablee Holdco Series A Preferred Stockholder to receive (x) subject to and in accordance with Section 2.10, the Per Share Purchase Price Holdback Amount and (y) subject to Article VII and in accordance with Section 2.6, the Per Share First Deferred Amount and the Per Share Second Deferred Amount, and, if applicable, the Per Share Third Deferred Amount.
- (e) **Conversion of Parent Merger Sub Capital Stock**. Each share of common stock of Parent Merger Sub issued and outstanding immediately prior to the Initial Merger Effective Time shall be converted into one fully paid and nonassessable share of common stock, \$0.001 par value per share, of the Initial Merger Surviving Corporation (the "Initial Merger Surviving Corporation Common Stock").
- (f) Conversion of Initial Merger Surviving Corporation. As of the Final Merger Effective Time, by virtue of the Final Merger and without any action on the part of the Initial Merger Surviving Corporation, Parent Holdco LLC or the holders of shares of Initial Merger Surviving Corporation Common Stock (i) each issued and outstanding share of Initial Merger Surviving Corporation Common Stock shall no longer be outstanding and shall automatically be canceled and retired and shall cease to exist, and no consideration shall be delivered in exchange therefor and (ii) each limited liability company interest of Parent Holdco LLC issued and outstanding immediately prior to the Final Merger Effective Time shall be converted into one limited liability company interest of the Final Surviving Company.

2.2. Effect of the Combination Transactions on the Treatment of Warrant of the Company and Shareablee Holdco.

(a) **Conversion of Company Warrant**. At the Shareablee Merger Effective Time, the Company Warrant that is outstanding as of immediately prior to the Shareablee Merger Effective Time shall, by virtue of the Shareablee Merger and without further action by the holder thereof, be converted into an award of the same type with respect to Shareablee Holdco on a one-for-one basis (with the same terms and exercise price, if applicable) and, following such conversion, shall be referred to as the "Shareablee Holdco Warrant."

Shareablee Holdco agrees to assume the obligations of the Company under the Company Warrant, and the Shareablee Holdco Warrant shall continue to be governed thereunder.

(b) **Conversion of Shareablee Holdco Warrant**. The Shareablee Holdco Warrant shall not be assumed, continued or substituted by Parent, Parent Holdco LLC, or Shareablee Holdco in connection with the Combination Transactions or the other transactions contemplated hereby. Immediately prior to the Initial Merger Effective Time, the Shareablee Holdco Warrant shall be cancelled and extinguished and, subject to the execution and delivery by the Shareablee Holdco Warrantholder of a Joinder Agreement and a properly completed Letter of Transmittal at or prior to the Closing, be converted automatically into (A) the right of the Shareablee Holdco Warrantholder to receive with respect to each share of Shareablee Holdco Common Stock subject thereto at the Closing, shares of Parent Common Stock equal to the Per Fully Diluted Share Closing Consideration minus a number of shares of Parent Common Stock equal to the aggregate exercise price payable thereunder divided by the Parent Stock Price, rounded up to the nearest whole share (the "Warrant Exercise Price"), and (B) the contingent right of the Shareablee Holdco Warrantholder to receive with respect to each share of Shareablee Holdco Common Stock (y) subject to and in accordance with Section 2.10, shares of Parent Common Stock equal to the Per Share Purchase Price Holdback Amount and (z) subject to Article VII and in accordance with Section 2.6, the Per Share First Deferred Amount and the Per Share Second Deferred Amount.

2.3. Effect of the Combination Transactions on the Treatment of Equity Awards of the Company and Shareablee Holdco.

(a) Treatment of Company Equity Awards.

- (i) Prior to the Shareablee Merger Effective Time, and subject to the prior review and approval of Parent, the Company shall take all actions necessary to effect the transactions anticipated by this Article II under the applicable Company Equity Plan and any Contract applicable to any Company Option and any Company RSU Award (whether written or oral, formal or informal), including delivering all required notices, obtaining all necessary approvals and consents, making all necessary assignments, and delivering evidence reasonably satisfactory to Parent that all necessary determinations by the Company Board or applicable committee of the Company Board to terminate all Company Options and Company RSU Awards (to the extent not assumed or converted as set forth in this Article II) in accordance with Section 2.3(b) have been made. Without limiting the foregoing, the Company shall take all actions necessary to ensure that the Company will not immediately prior to the Shareablee Merger Effective Time be bound by any options, stock appreciation rights, restricted stock units, warrants or other rights or agreements which would entitle any Person, other than Parent and its Subsidiaries, to own any capital stock of the Shareablee Merger Surviving Corporation or Final Surviving Company, or to receive any payment in respect thereof.
- (ii) At the Shareablee Merger Effective Time, each Company Option and Company RSU Award that is outstanding as of immediately prior to the Shareablee Merger Effective Time shall, by virtue of the Shareablee Merger and without further action by the holder thereof, be converted into an award of the same type with respect to Shareablee Holdco on a one-for-one basis (with the same terms and exercise price, if applicable) and, following such conversion, shall be referred to as "Shareablee Holdco Options" and "Shareablee Holdco RSU Awards," as applicable.

(b) Treatment of Shareablee Holdco Equity Awards.

(i) At the Initial Merger Effective Time, each Shareablee Holdco Option that is outstanding as of immediately prior to the Initial Merger Effective Time shall, by virtue of the Initial Merger and without any further action by Parent, Parent Holdco LLC, Shareablee Holdco or the holder thereof, be, in Parent's sole discretion, assumed by Parent and converted into, (I) a stock option of Parent that represents the right to acquire a number of validly issued, fully paid and non-assessable shares of Parent Common Stock, equal to the product of (A) the number of shares of Shareablee Holdco Common Stock subject to such Shareablee Holdco Option immediately prior to the Initial Merger Effective Time, *multiplied* by (B) the Per Fully Diluted Share Closing Consideration (each, a "*Converted Option*", and collectively, the "*Converted Options*"); *provided*, that any fractional share resulting from such multiplication shall be rounded up to the nearest whole share and (II) the contingent right to receive subject to Article VII and in accordance with Section 2.6, the Per Share First Deferred Amount and the Per Share Second Deferred Amount, and, if applicable, the Per Share Third Deferred Amount. Following the Initial Merger Effective Time, each Converted Option shall continue to be governed by the same material terms and conditions, including the vesting schedule, as were applicable immediately prior to the Initial Merger Effective Time to the Shareablee Holdco Option from which it was

converted or for which it is a substitute, in all cases subject to restrictions related to the issuance of shares under applicable Law; *provided*, *however*, that no Converted Option may be "early exercised" (i.e., each Converted Option may be exercised for shares of Parent Common Stock only to the extent the Converted Option is vested at the time of exercise pursuant to the applicable vesting schedule). The exercise price of each Converted Option shall be equal to (x) the exercise price of the Shareablee Holdco Option from which it was converted or for which it is a substitute, *divided* by (y) the Per Fully Diluted Share Closing Consideration, rounded up to the nearest whole cent. It is the intention of the Parties that the adjustments in this Section 2.3(b)(i) be performed in a manner that complies with or is exempt from Section 409A of the Code.

- (ii) At the Initial Merger Effective Time, each Shareablee Holdco RSU Award that is outstanding as of immediately prior to the Initial Merger Effective Time shall, by virtue of the Initial Merger and without any further action by Parent, Parent Holdco LLC, Shareablee Holdco or the holder thereof, be, in Parent's sole discretion, assumed by Parent and converted into, (I) a restricted stock unit of Parent covering a number of validly issued, fully paid and non-assessable shares of Parent Common Stock, equal to the product of (A) the number of shares of Shareablee Holdco Common Stock subject to such Shareablee Holdco RSU Award immediately prior to the Initial Merger Effective Time, multiplied by (B) the Per Fully Diluted Share Closing Consideration (each, a "Converted RSU Award", and collectively, the "Converted RSU Awards"); provided, that any fractional share resulting from such multiplication shall be rounded up to the nearest whole share and (II) the contingent right to receive subject to Article VII and in accordance with Section 2.6, the Per Share First Deferred Amount and the Per Share Second Deferred Amount, and, if applicable, the Per Share Third Deferred Amount. Following the Initial Merger Effective Time, each Converted RSU Award shall continue to be governed by the same material terms and conditions, including the vesting schedule, as were applicable immediately prior to the Initial Merger Effective Time to the Shareablee Holdco RSU Award from which it was converted or for which it is a substitute, in all cases subject to restrictions related to the issuance of shares under applicable Law.
- (iii) In connection with the Combination Transactions, Parent shall take all actions reasonably necessary to reserve for issuance a sufficient number of shares of Parent Common Stock for delivery upon exercise of such Converted Options and settlement of such Converted RSU Awards. With respect to the Converted Options and the Converted RSU Awards, Parent shall file with the SEC a registration statement on Form S-8 (or any successor form) relating to the Parent Common Stock issuable pursuant to the exercise of such Converted Options and settlement of such Converted RSU Awards as promptly as possible following the Closing Date and shall use commercially reasonable efforts to comply in all material respects with SEC reporting and other requirements necessary to keep the Form S-8 effective until all shares of Parent Common Stock issuable pursuant to the exercise of such Converted Options and the settlement of such Converted RSU Awards have been issued thereunder. Shares of Parent Common Stock that may become issuable pursuant to the exercise of such Converted Options and the settlement of such Converted RSU Awards and that at the time of such exercise or settlement, as applicable, are registered under an effective Form S-8 shall not constitute "restricted securities" under the Securities Act and shall not bear the Restrictive Legends set forth in Section 5.4(b).
- (c) **Issuance Cap**. Notwithstanding anything herein to the contrary, prior to the receipt of any applicable stockholder approval required by applicable listing standards of any stock exchange on which the Parent Common Stock is then listed, Parent shall not, under any circumstances, be required to deliver more than 19.99% of the number of shares of Parent Common Stock outstanding immediately prior to the Initial Merger Effective Time (such limitation, the "**Conversion Cap**"); provided, that if Parent would at any time be required to deliver hereunder (assuming all amounts payable hereunder are satisfied in shares of Parent Common Stock) a number of shares of Parent Common Stock exceeding the Conversion Cap but for the preceding sentence, Parent shall, at its option, (i) hold a meeting of its stockholders and use its commercially reasonable efforts to obtain the approval of its stockholders for such issuance in accordance with then-applicable stock exchange rules, or (ii) in lieu of delivering shares of Parent Common Stock in excess of the Conversion Cap (or upon a failure to obtain such approval), pay cash.

2.4. Allocation Schedule.

(a) Prior to the Closing, the Company shall prepare and deliver to Parent a spreadsheet (the "*Allocation Schedule*"), which shall be reasonably satisfactory to Parent, which Allocation Schedule shall be dated as of the Closing Date and shall set forth all of the information (in addition to the other required data and information specified therein) set forth on <u>Schedule B</u> hereto, as of immediately prior to the Closing.

- (b) The Allocation Schedule shall be accompanied by reasonably detailed back-up documentation for the calculations contained therein. The Company shall make available to Parent and its Representatives the work papers (subject to the execution of customary work paper access letters, if requested) and other books and records used in preparing the Allocation Schedule and reasonable access to employees of the Company as Parent may reasonably request in connection with its review of the Allocation Schedule, and will otherwise cooperate in good faith with Parent's and its Representatives' review and shall take into consideration in good faith any comments of Parent on the Allocation Schedule. Notwithstanding the foregoing, in no event will any of Parent's rights be considered waived, impaired or otherwise limited as a result of Parent not making an objection prior to the Closing or its making an objection that is not fully implemented in the Allocation Schedule, as applicable.
- (c) Each of the Parties acknowledges and agrees that, prior to any subsequent post-Closing distributions, the Stockholder Representative shall deliver to Parent an updated Allocation Schedule promptly following a request by Parent, and Parent and the Exchange Agent shall be entitled to conclusively rely upon the Allocation Schedule or an updated Allocation Schedule delivered by the Stockholder Representative, as applicable for all purposes hereof; *provided*, that the Stockholder Representative shall be entitled to conclusively rely upon the initial Allocation Schedule prepared by the Company in connection with the Stockholder Representative's preparation and delivery of any updated Allocation Schedule.
- (d) In no event will Parent, the Exchange Agent or any of their Affiliates have any liability to any person on account of issuances of Parent Common Stock or payments or distributions made in accordance with the Allocation Schedule or an updated Allocation Schedule delivered by the Stockholder Representative, as applicable. Parent, the Exchange Agent or any of their Affiliates shall be entitled to conclusively rely on the Allocation Schedule for purposes of this Agreement, including the issuance of any shares of Parent Common Stock as part of consideration for the Combination Transactions.

2.5. Surrender and Share Issuance Procedures.

- (a) At the Closing, the Company shall deliver to American Stock Transfer & Trust Company, LLC, as exchange agent (the "*Exchange Agent*"), the Allocation Schedule. At the Closing, Parent shall send, or cause the Exchange Agent to send, to each Company Stockholder and the Company Warrantholder pursuant to the Allocation Schedule, a letter of transmittal (the "*Letter of Transmittal*") and the Exchange Documentation applicable to such holder.
- (b) Subject to <u>Section 2.5(a)</u>, as promptly as practicable following receipt by the Exchange Agent of a Letter of Transmittal and applicable Exchange Documentation from a Company Stockholder or the Company Warrantholder, duly completed and validly executed in accordance with the instructions thereto (including, for the avoidance of doubt, any certificate with respect to any Company Capital Stock held by such Person in accordance with the procedures of the Exchange Agent), but for the avoidance of doubt in no event prior to the Initial Merger Effective Time, Parent shall cause the Exchange Agent to issue, in accordance with <u>Section 2.1(d)</u> or <u>Section 2.2(b)</u>, as applicable, to (i) each Shareablee Holdco Series A-2 Preferred Stockholder, the Per Share Series A-2 Consideration and (ii) each As-Converted Stockholder, the Per Fully Diluted Share Closing Consideration, in each case as set forth opposite such holder's name on the Allocation Schedule.
- (c) Subject to Section 2.5(a), in the event a Shareablee Holdco Stockholder is unable, under applicable Law, to provide the Suitability Documentation because such Person is not an Accredited Investor, following receipt by the Exchange Agent of a Letter of Transmittal and applicable Exchange Documentation from such Shareablee Holdco Stockholder, duly completed and validly executed in accordance with the instructions thereto, but for the avoidance of doubt in no event prior to the Initial Merger Effective Time, Parent shall pay, in accordance with Section 2.1(d), an amount in cash equal to such Shareablee Holdco Stockholder's Per Fully Diluted Share Closing Consideration, as set forth opposite such holder's name on the Allocation Schedule based on the Parent Stock Price.
- (d) At any time following the one (1) year anniversary of the Closing Date, the Shareablee Holdco Stockholders and Shareablee Holdco Warrantholder shall be entitled to look only to Parent as general creditors thereof with respect to any and all consideration that may be payable to such holders upon the due surrender of duly executed Exchange Documentation (if any) in the manner set forth in Section 2.5(a).
- (e) Notwithstanding anything to the contrary in this <u>Section 2.5</u>, none of Parent, the Exchange Agent, the Shareablee Merger Surviving Corporation, the Final Surviving Company nor any Party

shall be liable to any Person for any amount paid to a public official or Governmental Entity pursuant to any applicable abandoned property, escheat or similar Law. Any portion of the Total Merger Consideration that remains unclaimed immediately prior to the date on which it would otherwise become subject to any abandoned property, escheat or similar Law shall, to the extent permitted by applicable Law, become the property of Parent, free and clear of all claims or interest of any Person previously entitled thereto.

2.6. Deferred Amount.

- (a) In connection with the Closing, Parent shall reserve for issuance, out of its authorized but unissued shares of Parent Common Stock, the number of shares of Parent Common Stock equal to the Deferred Amount (assuming solely for purposes of such reservation that the Deferred Amount shall be paid entirely in shares of Parent Common Stock and using the Parent Stock Price as of the Closing Date to determine the number of shares to reserve); *provided*, that Parent reserves the right to satisfy any and all obligations of Parent to the Deferred Amount Holders with respect to the Deferred Amount in cash at its election. The Deferred Amount shall become payable to the Deferred Amount Holders, if at all, subject to the terms and conditions of this Agreement. The Parties agree that, Parent shall be treated as the owner of the Deferred Amount until and to the extent issued or paid to the Deferred Amount Holders.
- Subject to the other terms and conditions of this Agreement (Section 2.6(e) and Article VII, and solely in the case of Tania Yuki Mustapic, subject to Section 2.6(c)), as promptly as practicable following (i) the First Release Date, Parent shall, or shall cause the Exchange Agent to, issue or pay to each Deferred Amount Holder all or the applicable portion (which may be zero), as applicable, of the Per Share First Deferred Amount applicable to such Deferred Amount Holder in accordance with such Deferred Amount Holder's Pro Rata Share, (ii) the two (2) year anniversary of the Closing Date, Parent shall, or shall cause the Exchange Agent to, issue or pay to each Deferred Amount Holder all or the applicable portion (which may be zero), as applicable, of the Per Share Second Deferred Amount applicable to such Deferred Amount Holder in accordance with such Deferred Amount Holder's Pro Rata Share, and (iii) the three (3) year anniversary of the Closing Date, Parent shall, or shall cause the Exchange Agent to, issue to Tania Yuki Mustapic all or the applicable portion (which may be zero) of the Per Share Third Deferred Amount applicable to Tania Yuki Mustapic in accordance with Tania Yuki Mustapic's Pro Rata Share (as applicable in clauses (i), (ii) and (iii), the "*Release Date*"). Notwithstanding anything to the contrary in this Agreement, any payment or issuance of any Deferred Amount shall be reduced by the amounts of any Losses alleged under any outstanding Liability Claim or Third Party Claim as of the applicable Release Date (and if such Deferred Amount is elected by Parent to be paid in shares of Parent Common Stock under this Agreement, each share of Parent Common Stock shall be deemed to have a value equal to the Parent Stock Price as of the applicable Release Date); *provided* Parent shall, in accordance with this <u>Section 2.6(b)</u>, promptly pay or issue (and if such Deferred Amount is elected by Parent to be paid in shares of Parent Common Stock under this Agreement, each share of Parent Common Stock shall be deemed to have a value equal to the Parent Stock Price as of the applicable Release Date) the amount by which the applicable Deferred Amount was reduced if and to the extent the alleged Losses underlying such outstanding Liability Claim or Third Party Claim are finally determined by a non-appealable order of a court of competent jurisdiction not to be actual Losses of the Parent Indemnified Parties in accordance with Article VII.
- (c) Notwithstanding the foregoing <u>Section 2.6(b)</u>, Parent's obligation to pay or issue the Deferred Amount to Tania Yuki Mustapic shall immediately terminate, and Tania Yuki Mustapic agrees to forfeit any rights she may have with respect to the Deferred Amount, upon any material breach of the Stockholder Non-Competition and Non-Solicitation Agreement alleged by Parent in good faith (which is not cured within ten (10) Business Days following Tania Yuki Mustapic's receipt of written notice thereof; *provided* if Tania Yuki Mustapic is pursuing the cure of such alleged material breach in good faith at the expiration of such cure period, then such cure period shall be extended by ten (10) additional Business Days). If Tania Yuki Mustapic is finally determined by a non-appealable order of a court of competent jurisdiction not to be in breach of the Stockholder Non-Competition and Non-Solicitation Agreement or Parent withdraws the allegation of breach in full, Parent shall, in accordance with <u>Section 2.6(b)</u>, promptly pay or issue (and if such Deferred Amount is elected by Parent to be paid in shares of Parent Common Stock under this Agreement, each share of Parent Common Stock shall be deemed to have a value equal to the Parent Stock Price as of the applicable Release Date) to Tania Yuki Mustapic the amount deemed forfeited pursuant to the preceding sentence.
- (d) Notwithstanding anything to the contrary in this Agreement, if, prior to the ninety (90) day anniversary of the Closing Date (the "*Outside Delivery Date*"), either (x) Parent has not received all of the

Approved Data Consents or (y) all communications from the Data Partners questioning or raising concerns about the Company's use of such Data Partner's data, in any case that were raised prior to the Closing or were raised (whether prior to or after Closing) in connection with an audit initiated prior to the Closing, have not been fully remediated or resolved, the following shall apply:

- (i) As soon as reasonably practicable, and in any event within sixty (60) days after the end of the calendar year ended December 31, 2022, Parent shall prepare and deliver (or cause to be prepared and delivered) to the Stockholder Representative a statement setting forth in reasonable detail a calculation of Revenue, together with a determination of the Reduction Amount (the "2022 Statement"). Parent shall prepare such 2022 Statement in good faith and in accordance with Parent's normal and customary accounting methods, but subject to Schedule E and Schedule F.
- (ii) During the thirty (30) day period immediately following the Stockholder Representative's receipt of the 2022 Statement, Parent shall make available to the Stockholder Representative and its Representatives the books and records used in preparing the 2022 Statement (subject to the execution of customary work paper access letters, if requested) and provide reasonable access (on prior written notice and during business hours) to employees of Parent as the Stockholder Representative may reasonably request in connection with its review of the 2022 Statement, and shall otherwise reasonably cooperate in good faith with the Stockholder Representative and its Representatives in their review of the 2022 Statement. The 2022 Statement, including the determination of the Reduction Amount set forth therein, shall be final and binding on the Parties unless the Stockholder Representative delivers an objection notice in writing to Parent within thirty (30) days following the Stockholder Representative's receipt of the 2022 Statement (an "Objection Notice") setting forth in reasonable detail its objection(s) to the 2022 Statement and the Stockholder Representative's proposed adjustments to the 2022 Statement with reasonably detailed supporting documentation, including the proposed calculation of the Reduction Amount.
- During the thirty (30) days immediately following the delivery of an Objection Notice, if any, Parent and the Stockholder Representative agree to seek in good faith to resolve any disagreement that they may have with respect to the matters specified in the Objection Notice. If at the end of the thirty (30) day period, Parent and the Stockholder Representative have been unable to resolve any differences they may have with respect to the matters specified in the Objection Notice, Parent and Stockholder Representative shall refer any unresolved differences to the Independent Accountant. The Independent Accountant, acting as an expert and not an arbitrator, shall make a final determination as to referred unresolved differences within thirty (30) days after its engagement. In resolving any referred unresolved differences, the Independent Accountant may not assign a value greater than the highest value nor less than the lowest value for such referred unresolved difference claimed by either Party. The Independent Accountant's determination shall be based solely on presentation by Parent and the Stockholder Representative (i.e. not an independent review) and on the terms in this Agreement that are relevant to the calculation of the disputed item. The final determination of the Independent Accountant of the 2022 Statement, including the Reduction Amount, shall be binding on all Parties. The fees and expenses of the Independent Accountant shall be borne by Parent and the Stockholder Representative (on behalf of the Deferred Amount Holders) based on the inverse of the percentage that the Independent Accountant's resolution of the referred unresolved differences covered by such Objection Notice (before such allocation) bears to the total amount of such referred unresolved differences as originally submitted to the Independent Accountant (for example, if the total amount of such referred unresolved differences as originally submitted to the Independent Accountant equals \$1,000 and the Independent Accountant awards \$600 in favor of the Stockholder Representative's position, sixty percent (60%) of the fees and expenses of the Independent Accountant would be borne by Parent and forty percent (40%) of the fees and expenses of the Independent Accountant would be borne by the Stockholder Representative (on behalf of the Deferred Amount Holders)). Parent shall pay the expenses of the Independent Accountant, provided that the Deferred Amount Holders shall bear their portion of the expenses as set forth in the definition of "Deferred Amount".
- (iv) Parent acknowledges that the payment of the Deferred Amount serves as a material inducement for the Company and the Company Stockholders and Company Warrantholder to enter into this Agreement, and Parent covenants through December 31, 2022, that Parent will not take any action with the actual intent of reducing Revenue with the actual intent that the Deferred Amount is reduced as a result thereof. Notwithstanding the foregoing, the Parties agree that Parent shall not be responsible for any lost Revenue caused by the Company's failure or delay in obtaining the Approved Data Consents on or before the Outside Delivery Date.

- (v) After the determination of the Reduction Amount in accordance with this <u>Section 2.6(d)</u>, the Deferred Amount shall be reduced by the Reduction Amount for all purposes of this Agreement.
- (e) If at any time during the period between the Initial Merger Effective Time and, with respect to the issuance or payment of the Deferred Amount, on or prior to the issuance or payment of the Deferred Amount pursuant to the terms of this Agreement, any change in the outstanding shares of Parent Common Stock shall occur by reason of any reclassification, stock split or combination, exchange or readjustment of shares, or any stock dividend thereon with a record date during such period, any number or amount or class contained in this Agreement which is based on the price of Parent Common Stock or the number or class of shares of Parent Common Stock shall be equitably adjusted to the extent necessary to provide the Parties with the same economic effect with respect to the Deferred Amount as contemplated by this Agreement prior to such reclassification, stock split or combination, exchange or readjustment of shares, or stock dividend thereon.
- (f) In connection with the consummation of a Change of Control Event prior to the three (3) year anniversary of the Closing Date, Parent shall make appropriate provision so that the successors and assigns of Parent shall affirmatively assume the obligations of Parent with respect to the Deferred Amount. If, as a result of such Change of Control Event, shares of Parent Common Stock are exchanged for or converted into cash or other securities, then Parent shall elect that the Deferred Amount, if any, be made in the form of such cash or other securities based on the Parent Stock Price that would have applied had the Deferred Amount, if any and as applicable under Section 2.6(b), been issued or paid to the Deferred Amount Holders as of immediately prior to such Change of Control Event.
- (g) The approval and adoption of this Agreement and approval of the Combination Transactions by the Company Stockholders, and the delivery of the Joinder Agreements by the Company Stockholders and Company Warrantholder, constitutes approval by such Company Stockholders and Company Warrantholder as to the specific terms of the Combination Transactions, and the irrevocable agreement of the Company Stockholders and Company Warrantholder to be bound by, and comply with, this Agreement, including the matters set forth in this <u>Section 2.6</u> and Parent's retention of the Deferred Amount.
- **2.7. No Fractional Shares.** Notwithstanding anything to the contrary set forth herein, the number of shares of Parent Common Stock to be issued to each Shareablee Holdco Stockholder and the Shareablee Holdco Warrantholder pursuant to Section 2.1(d) or Section 2.2(b), as applicable, and in connection with the issuance or payment of any Deferred Amount, if any, as applicable, shall be rounded up to the nearest whole number of shares of Parent Common Stock. Notwithstanding anything to the contrary contained in this Agreement, no fractional shares of Parent Common Stock shall be issued to any Shareablee Holdco Stockholder or the Shareablee Holdco Warrantholder in connection with the Combination Transactions and the Parent Stock Issuance and any such fractional share interests shall not entitle the owner thereof to vote or to any other rights of a holder of shares of Parent Common Stock. Each share of Parent Common Stock issuable in the Combination Transactions, or any other securities issued in respect of such shares upon any stock split, stock dividend, merger, consolidation or similar event, shall be book-entry security entitlements.
- **2.8. Deferred Amount Grants**. Notwithstanding anything to the contrary herein, Parent shall be permitted, in its sole discretion, to satisfy its obligations with respect to the applicable portion of the Deferred Amount to any Equity Award Holders with shares issued under the Company Equity Plan, which shares shall have the same terms and conditions as to the Deferred Amount set forth in Section 2.6 of this Agreement.
- **2.9. Withholding**. Notwithstanding any other provision of this Agreement, Parent, the Final Surviving Company, the Exchange Agent and any Affiliate of the foregoing shall be entitled to deduct or withhold from any consideration issuable, payable or otherwise deliverable pursuant to this Agreement such amounts as may be required to be deducted or withheld therefrom under any provision of U.S. federal, state, local or non-U.S. Tax Law or under any applicable Law or Order, and to request and be provided any necessary Tax Forms or any similar information; *provided*, that except in the case of backup withholding, any payment of wages for applicable Tax purposes with respect to Transaction Expenses, the failure of the Company to deliver the FIRPTA Compliance Certificate or the failure of an applicable payee to deliver an applicable Tax Form, the Person intending to deduct or withhold under this Section 2.9 shall provide prompt notice to the applicable payee upon becoming aware that such deduction or withholding is required. Parent shall (and shall cause its Affiliates to) cooperate with the applicable payee and take such commercially reasonable steps as such payee may reasonably request to reduce or eliminate amounts that would otherwise be deducted or withheld under this Section 2.9. To the extent any such amounts are so deducted or withheld and paid to the proper Governmental

Entity, such amounts shall be treated for all purposes under this Agreement as having been paid to the Persons to whom such amounts would otherwise have been paid. Notwithstanding any other provision of this Agreement, the employer portion of any income, payroll or other employment-related taxes withheld by Parent, the Final Surviving Company, the Exchange Agent or any affiliate in connection with the payment to any Deferred Amount Holder that was an Equity Award Holder as of immediately prior to the Closing Date shall be deemed to be paid to such Deferred Amount Holder for all purposes of Section 2.6 of this Agreement. Notwithstanding any other provision of Section 2.6 of this Agreement to the contrary, payment to any Deferred Amount Holder subject to payroll tax withholding (which shall include any Deferred Amount Holder that was an Equity Award Holder as of immediately prior to the Closing Date) may be delayed until no later than March 14 of the year following the year in which such portion of the Deferred Amount becomes payable to such Deferred Amount Holder to facilitate a "sell to cover" transaction to permit such Deferred Amount Holder to satisfy his or her income and/or payroll tax obligation that becomes due upon receipt of such Deferred Amount.

2.10. Post-Closing Reconciliation.

- (a) On the Closing Date, the Company shall deliver to Parent its good faith calculation of (i) the estimated Closing Net Working Capital Amount, (ii) the estimated Closing Indebtedness Amount (excluding the PPP Loan), (iii) the estimated Closing Cash Amount, and (iv) the Closing Transaction Expense Amount (collectively, the "Adjustment Amounts" (which may be positive or negative numbers), and such estimates, the "Estimated Adjustment Amounts"), in each case, accompanied by reasonably detailed back-up documentation for such calculations (the "Estimated Closing Statement"). The Company shall prepare the Estimated Closing Statement in accordance with the applicable definitions in this Agreement and with GAAP. The Company shall make available to Parent and its Representatives the books and records used in preparing the Estimated Closing Statement and reasonable access (on prior notice and during business hours) to employees of the Company as Parent may reasonably request in connection with its review of such statements, and will otherwise cooperate in good faith with Parent's and its Representatives review of such statements and shall take into consideration in good faith any comments of Parent on the Estimated Closing Statement, as applicable. Notwithstanding the foregoing, in no event will any of Parent's rights be considered waived, impaired or otherwise limited as a result of Parent not making an objection prior to the Closing or its making an objection that is not fully implemented in the Estimated Closing Statement, as applicable.
- As soon as reasonably practicable after the Closing Date, and in any event within ninety (90) days after the Closing Date, Parent shall prepare and deliver to the Stockholder Representative a statement that shall set forth a calculation of each Adjustment Amount (the "Final Adjustment Amounts"), accompanied by reasonably detailed back-up documentation for such calculations (the "Post-Closing Statement"). Parent shall prepare such Post-Closing Statement in accordance with the applicable definitions in this Agreement and with GAAP. After the Stockholder Representative's receipt of the Post-Closing Statement, Parent shall make available to the Stockholder Representative and its Representatives the books and records used in preparing the Post-Closing Statement (subject to the execution of customary work paper access letters, if requested) and provide reasonable access (on prior written notice and during business hours) to employees of Parent and the Company as the Stockholder Representative may reasonably request in connection with its review of the Post-Closing Statement, and shall otherwise reasonably cooperate in good faith with Stockholder Representative and its Representatives in their review of the Post-Closing Statement. The Stockholder Representative shall notify Parent in writing (the "Notice of Adjustment Disagreement") within thirty (30) days of the Stockholder Representative's receipt of the Post-Closing Statement (the "Adjustment Review **Period**") if the Stockholder Representative disagrees with any portion of the Post-Closing Statement. The Notice of Adjustment Disagreement shall set forth in reasonable detail the basis for such disagreement, the amounts involved and the Stockholder Representative's proposed adjustments to the Post-Closing Statement with reasonably detailed supporting documentation. If no Notice of Adjustment Disagreement is received by Parent on or prior to the expiration date of the Adjustment Review Period, then the Post-Closing Statement and all amounts set forth therein shall be deemed to have been accepted by the Stockholder Representative and shall become final and binding upon the Parties, the Company Stockholders and the Company Warrantholder. During the thirty (30) days immediately following the delivery of a Notice of Adjustment Disagreement (the "Adjustment **Resolution Period**"), if any, the Stockholder Representative and Parent shall seek in good faith to resolve any disagreement that they may have with respect to the matters specified in the Notice of Adjustment Disagreement. Any items agreed to by the Stockholder Representative and Parent in a written agreement executed and delivered by each of the Stockholder Representative and Parent, together with any items not disputed or objected to by the Stockholder Representative in the Notice of Adjustment Disagreement, are collectively referred to herein as the "Resolved Matters." If at the end of the Adjustment Resolution Period, the Parties have been unable to resolve any differences they may have with respect to the matters specified in the Notice of Adjustment Disagreement,

the Stockholder Representative and Parent, or either of them, shall refer all matters in the Notice of Adjustment Disagreement other than the Resolved Matters (the "*Unresolved Matters*") to Grant Thornton LLP (the "*Independent Accountant*"). In the event that Grant Thornton LLP refuses or is otherwise unable to act as the Independent Accountant, the Stockholder Representative and Parent shall cooperate in good faith to appoint an independent certified public accounting firm in the United States of national recognition mutually agreeable to the Stockholder Representative and Parent, in which event "Independent Accountant" shall mean such firm. Within thirty (30) days after the submission of such matters to the Independent Accountant, the Independent Accountant, acting as an expert and not as an arbitrator, will make a final determination, binding on the Parties, of the appropriate amount of each of the Unresolved Matters. With respect to each Unresolved Matter, such determination, if not in accordance with the position of either the Stockholder Representative or Parent, shall not be in excess of the higher, nor less than the lower, of the amounts advocated by the Stockholder Representative in the Notice of Adjustment Disagreement or Parent in the Post-Closing Statement with respect to such Unresolved Matter. For the avoidance of doubt, the Independent Accountant shall not review any line items in the Post-Closing Statement or make any determination with respect to any matter other than the Unresolved Matters. During the review by the Independent Accountant, Parent and the Stockholder Representative shall each make available to the Independent Accountant such individuals and such information, books, records and work papers, as may be reasonably required by the Independent Accountant to fulfill its obligations under this Section 2.10(b); provided, that the independent accountants of Parent or the Company shall not be obligated to make any working papers available to the Independent Accountant unless and until the Independent Accountant has signed a customary confidentiality and hold harmless agreement relating to such access to working papers in form and substance reasonably acceptable to such independent accountants. The fees and expenses of the Independent Accountant shall be borne by Parent and the Stockholder Representative (on behalf of the As-Converted Stockholders) based on the inverse of the percentage that the Independent Accountant's resolution of the disputed items covered by such Notice of Adjustment Disagreement (before such allocation) bears to the total amount of such disputed items as originally submitted to the Independent Accountant (for example, if the total amount of such disputed items as originally submitted to the Independent Accountant equals \$1,000 and the Independent Accountant awards \$600 in favor of the Stockholder Representative's position, sixty percent (60%) of the fees and expenses of the Independent Accountant would be borne by Parent and forty percent (40%) of the fees and expenses of the Independent Accountant would be borne by the Stockholder Representative (on behalf of the As-Converted Stockholders)). Parent shall pay the expenses of the Independent Accountant, provided that the As-Converted Stockholders shall bear their portion of the expenses as set forth in Section 2.10(d) and Section 2.10(e).

- (c) The "Final Closing Statement" and "Final Adjustment Amounts" shall be (i) in the event that no Notice of Adjustment Disagreement is delivered by the Stockholder Representative to Parent prior to the expiration of the Adjustment Review Period, the Post-Closing Statement delivered by Parent to the Stockholder Representative pursuant to Section 2.10(b), (ii) in the event that a Notice of Adjustment Disagreement is delivered by the Stockholder Representative to Parent prior to the expiration of the Adjustment Review Period and Parent and the Stockholder Representative are able to agree on all matters set forth in such Notice of Adjustment Disagreement, the Post-Closing Statement delivered by Parent to the Stockholder Representative pursuant to Section 2.10(b) as adjusted pursuant to the written agreement executed and delivered by Parent and the Stockholder Representative or (iii) in the event that a Notice of Adjustment Disagreement is delivered by the Stockholder Representative to Parent prior to the expiration of the Adjustment Review Period and Parent and the Stockholder Representative are unable to agree on all matters set forth in such Notice of Adjustment Disagreement, the Post-Closing Statement delivered by Parent to the Stockholder Representative pursuant to Section 2.10(b) as adjusted by the Independent Accountant to be consistent with the Resolved Matters and the final determination of the Independent Accountant of the Unresolved Matters in accordance with Section 2.10(b). The date on which the Final Closing Statement is finally determined in accordance with this Section 2.10(c) is hereinafter referred to as the "Determination Date."
- (d) Upon determination of the Final Closing Statement, if the sum of the Final Adjustment Amounts set forth in the Final Closing Statement is greater than the sum of the Estimated Adjustment Amounts (the amount of such excess, *minus* any costs allocated to the Stockholder Representative (on behalf of the As-Converted Stockholders) pursuant to Section 2.10(b), the "Upward Adjustment Amount"), then, within three (3) Business Days after the Determination Date, Parent shall deposit with the Exchange Agent the number of shares of Parent Common Stock with an aggregate value equal to the Upward Adjustment Amount based on the Parent Stock Price as of the Closing Date (the "Upward Adjustment Consideration") for further distribution to the As-Converted Stockholders in accordance with the additional amounts that each As-Converted Stockholder would have received had the Final Adjustment Amount been included in the Maximum Available Fully Diluted Shares Closing Consideration instead of the Estimated Adjustment Amount at the time of Closing; provided,

that as a condition to Parent's and the Exchange Agent's obligation to make such issuances, the Stockholder Representative shall first deliver to Parent an updated Allocation Schedule setting forth the portion of the Upward Adjustment Consideration payable to each As-Converted Stockholder; *provided*, *further*, that in no event shall the Upward Adjustment Consideration exceed the Maximum Adjustment Consideration. If the absolute value of the Upward Adjustment Amount is less than \$50,000, the "Upward Adjustment Amount" shall be deemed to be \$0 and there shall be no Upward Adjustment Consideration.

- (e) Upon determination of the Final Closing Statement, if the sum of the Final Adjustment Amounts set forth in the Final Closing Statement is less than the sum of the Estimated Adjustment Amounts (the amount of such deficiency, together with any costs allocated to the Stockholder Representative (on behalf of the As-Converted Stockholders) pursuant to Section 2.10(b), the "Downward Adjustment Amount"), then, within three (3) Business Days after the Determination Date, Parent shall deduct from the Purchase Price Holdback Amount the number of shares of Parent Common Stock with an aggregate value equal to the Downward Adjustment Amount based on the Parent Stock Price as of the Closing Date (the "Downward Adjustment Consideration"); provided, that in the event that the Downward Adjustment Consideration exceeds the Maximum Adjustment Consideration, Parent shall be entitled to reduce the Deferred Amount payable to the As-Converted Stockholders by such excess. If the absolute value of the Downward Adjustment Amount is less than \$50,000, the "Downward Adjustment Amount" shall be deemed to be \$0 and there shall be no Downward Adjustment Consideration.
- (f) Parent shall promptly cause the Exchange Agent to issue to each As-Converted Stockholder all or the applicable portion (which may be zero), as applicable, of the Per Share Purchase Price Holdback Amount (in excess of the Downward Adjustment Consideration, if applicable).
- (g) Parent and the Final Surviving Company shall be entitled to conclusively rely upon the updated Allocation Schedule delivered by the Stockholder Representative, including with respect to whether any individual As-Converted Stockholder received the appropriate portion of any such distribution, and in no event will Parent, the Final Surviving Company or any of their Affiliates have any liability to any person on account of payments or distributions made in accordance with the updated Allocation Schedule delivered by the Stockholder Representative; *provided*, that the Stockholder Representative shall be entitled to conclusively rely upon the initial Allocation Schedule prepared by the Company in connection with the Stockholder Representative's preparation and delivery of any updated Allocation Schedule.
- (h) Except with respect to claims of Fraud and subject to <u>Section 7.2</u>, the process set forth in this <u>Section 2.10</u> shall be the exclusive remedy for disputes related to the Adjustment Amounts.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY PARTIES

Subject to the disclosures set forth in the disclosure schedule of the Company, Shareablee Holdco and Shareablee Merger Sub (the "*Company Parties*") dated as of the Closing Date and delivered to Parent concurrently with the execution of this Agreement as <u>Schedule A</u> (the "*Disclosure Schedule*") (each of which disclosures shall indicate the Section and, if applicable, the Subsection of this <u>Article III</u> to which it relates (unless and only to the extent the relevance to other representations and warranties is readily apparent from the actual text of the disclosures,), the Company Parties represent and warrant as of the Closing Date (except to the extent any such representation or warranty expressly relates to a different date (in which case as of such date)) to the Parent Parties as follows:

3.1. Organization.

- (a) Each of the Company Parties is duly incorporated or formed, validly existing and in good standing under the laws of the State of Delaware. Each of the Company Parties has the requisite corporate or other legal power and authority, as applicable, to own, operate, distribute and lease its assets and properties to carry on its businesses as it is now being conducted and as currently proposed to be conducted and is duly qualified to do business and is in good standing in the State of Delaware where the failure to be so qualified and in good standing, individually or in the aggregate with any such other failures, would reasonably be expected to have a material and adverse effect on the Company Parties.
- (b) Each of the Company Parties has made available correct and complete copies of its governing documents, as amended to date, in full force and effect on the Closing Date (collectively, the

"Charter Documents"). The Company Board has not approved or proposed any amendment to any of the current Charter Documents, other than in connection with the transactions contemplated by this Agreement.

3.2. Company Capital Structure.

- Immediately prior to the Closing, the authorized capital stock of the Company consists of (i) 26,000,000 shares of Company Common Stock, \$0.001 par value, of which 9,725,666 shares are issued and outstanding, and (ii) 11,748,879 shares of Company Preferred Stock, \$0.001 par value, (A) 8,662,019 shares of which are designated Series A Preferred Stock and 8,662,019 of which are issued and outstanding, and (B) 3,086,860 shares of which are designated Series A-2 Preferred Stock and 2,287,593 of which are issued and outstanding. In connection with the transactions contemplated hereby as a Liquidation Event (as defined in the Certificate of Incorporation), (x) each share of Company Series A-2 Preferred Stock is entitled to an amount per share equal to the greater of (I) two (2) times \$1.6692, plus any dividends declared but unpaid thereon or (2) such amount per share as would have been payable had all shares of Company Series A-2 Preferred Stock been converted into Company Common Stock immediately prior to such Liquidation Event and (y) each share of Company Series A Preferred Stock is entitled to an amount per share equal to the greater of (i) \$0.9086, plus any dividends declared but unpaid thereon or (ii) such amount per share as would have been payable had all shares of Company Series A Preferred Stock been converted into Company Common Stock immediately prior to such Liquidation Event. There are no other issued and outstanding shares of Company Capital Stock and no commitments or Contracts to issue any shares of Company Capital Stock other than pursuant to the exercise of Company Options and the settlement of Company RSU Awards under the Company Equity Plan and the Company Warrant that is outstanding as of immediately prior to the Closing and set forth on Section 3.2(d)-1 of the Disclosure Schedule. Immediately prior to the Closing, there are 21,657,776 Fully Diluted Shares. The only Equity Interests of the Company held by Persons who are not Accredited Investors are 2,000 shares of Company Common Stock. Section 3.2(a)-1 of the Disclosure Schedule sets forth, as of immediately prior to the Closing, a correct and complete list of the holders of issued and outstanding shares of Company Capital Stock and each other Equity Interest of the Company (other than Company Options and Company RSU Awards set forth on Section 3.2(d)-1 of the Disclosure Schedule) with the number and type of such shares or such other Equity Interests so owned by each such holder, and any beneficial holders thereof. All issued and outstanding shares of Company Capital Stock or other Equity Interests of the Company are duly authorized, validly issued, fully paid and non-assessable and are free of any Liens, outstanding subscriptions, preemptive rights or "put" or "call" rights created by statute, the Charter Documents or any Contract to which the Company is a party or by which the Company is bound. The Company has never declared or paid any dividends on any shares of Company Capital Stock or other Equity Interests. There is no Liability for dividends accrued and unpaid by the Company. The Company is not under any obligation to register under the Securities Act or any other applicable Law any shares of Company Capital Stock, any Company Securities or any other securities or Equity Interests. All issued and outstanding shares of Company Capital Stock and all Company Options and Company RSU Awards were issued in compliance with applicable Law and all requirements set forth in the Charter Documents and any applicable Contracts to which the Company is a party or by which the Company or any of its assets is bound. No shares of Company Capital Stock are subject to vesting, reverse vesting, forfeiture, a right of repurchase or to a "substantial risk of forfeiture" within the meaning of Section 83 of the Code. The consideration received pursuant to <u>Section 2.1(d)</u>, including after giving effect to any indemnification obligations of such Persons under <u>Article VII</u> of this Agreement and any forfeiture of the Deferred Amount, are made in accordance with the Liquidation Event provisions under the Certificate of Incorporation, and satisfy in full the Series A-2 Liquidation Amount and Series A Liquidation Amount (as defined in the Certificate of Incorporation) to which the Company Series A-2 Preferred Stockholders and Company Series A Preferred Stockholders, respectively, are entitled under the Certificate of Incorporation. Immediately prior to the Closing, the authorized capital stock of the Shareablee Holdco consists of (iii) 26,000,000 shares of Shareablee Holdco Common Stock, and (iv) 11,748,879 shares of Shareablee Holdco Preferred Stock, \$0.001 par value. Immediately prior to the Closing, there are no issued and outstanding Equity Interests of Shareablee Holdco Common Stock, other than the shares of Shareablee Holdco Common Stock owned by the Company.
- (b) Section 3.2(b) of the Disclosure Schedule sets forth, as of immediately prior to the Closing, the holder of outstanding Company Warrant, including the number of shares of Company Common Stock subject to such Company Warrant, the grant date, the exercise price for such Company Warrant, the extent to which such Company Warrant is vested and exercisable and the date on which such Company Warrant expires.
- (c) Section 3.2(c) of the Disclosure Schedule identifies, as of immediately prior to the Closing, each Person with an offer letter or other Contract that contemplates a grant of Company Options, grant

of Company RSU Awards, or grant or issuance of other securities of the Company (including the number, series and class of shares, intended exercise price (had such Company Option been granted in the Ordinary Course of Business based on the offer letter or other Contract contemplating such grant, as applicable, that was executed by the Company or otherwise in the Ordinary Course of Business in the absence of this Agreement and the transactions contemplated hereby) and intended vesting schedule (including, any accelerated vesting, if applicable), or who has otherwise been promised Company Options, Company RSU Awards or other securities of the Company, which Company Options or Company RSU Awards have not been granted, or other securities have not been granted or issued, as of immediately prior to the Closing.

- As of immediately prior to the Closing, the Company has reserved 4,500,000 shares of Company Common Stock for issuance to employees, non-employee directors and consultants pursuant to the Company Equity Plan, of which 3,045,527 shares are subject to outstanding and unexercised Company Options, 168,564 shares are subject to outstanding Company RSU Awards, and 507,659 shares remain available for issuance thereunder. The aggregate exercise price applicable to all Company Options (the "*Aggregate Option Exercise Price*") is \$1,224,332.29. Section 3.2(d)-1 of the Disclosure Schedule sets forth, as of immediately prior to the Closing, a correct and complete list of all Equity Award Holders, and each Company Option and Company RSU Award, whether or not granted under the Company Equity Plan, including the number and type of shares of Company Capital Stock subject to each Company Option and Company RSU Award, the number of such shares that are vested or unvested, the "date of grant" of such Company Option or Company RSU Award (in the case of any Company Option, as defined under U.S. Treasury Regulation 1.409A-1(b)(5)(vi)(B)), the vesting commencement date, the vesting schedule (and the terms of any acceleration thereof), the exercise price per share of any Company Option, the Tax status of such Company Option under Section 422 of the Code (or any applicable foreign Tax Law) or Company RSU Award, the term of each Company Option, the plan from which such Company Option or Company RSU Award was granted (if any), whether such Company Option or Company RSU Award was granted with an "early exercise" right in favor of the holder, and the country and state of residence of such Equity Award Holder. All Company Options listed on Section 3.2(d)-1 of the Disclosure Schedule that are denoted as incentive stock options under Section 422 of the Code so qualify. Section 3.2(d)-2 of the Disclosure Schedule indicates, as of immediately prior to the Closing, which Equity Award Holders are Persons that are not employees of the Company (including non-employee directors, consultants, advisory board members, vendors, service providers or other similar Persons), including a description of the relationship between each such Person and the Company. In connection with the Combination Transactions, the treatment of Shareablee Holdco Options and Shareablee Holdco RSU Awards under Article II will not violate the Company Equity Plan, applicable Laws, and the underlying individual agreements for such equity awards.
- (e) With respect to the Company Options and Company RSU Awards, (i) each grant of an equity award was duly authorized no later than the date on which the grant of such equity award was by its terms to be effective by all necessary corporate action, (ii) each award of Company Options or Company RSU Award has been made using the applicable standard form award agreement under the Company Equity Plan, correct and complete copies of which has been made available to Parent, (iii) no Company Options or Company RSU Awards differ in any material respect from such applicable form agreement (other than any vesting acceleration provisions contained therein as indicated in Section 3.2(e) of the Disclosure Schedule), and (iv) there is no agreement, arrangement or understanding (written or oral) to amend, modify or supplement any such award agreements in any case from the forms made available to Parent. Except as set forth on Section 3.2(d)-1 of the Disclosure Schedule, no Company Option that was granted so as to qualify as an incentive stock option as defined in Section 422 of the Code was early exercised by the holder of such Company Option.
- (f) As of immediately prior to the Closing, there are no authorized, issued or outstanding Company Securities other than shares of Company Capital Stock, Company Options, Company RSU Awards, and the Company Warrant. Other than as set forth on Section 3.2(a), Section 3.2(b), Section 3.2(d)1 and Section 3.2(e) of the Disclosure Schedule, as of immediately prior to the Closing, no Person holds any Company Securities or other Equity Interests, stock appreciation rights, stock units, share schemes, calls or rights, or is party to any Contract of any character to which the Company or a Company Stockholder is a party or by which it or its assets is bound, (i) obligating the Company or such Company Stockholder to issue, deliver, sell, repurchase or redeem, or cause to be issued, delivered, sold, repurchased or redeemed, any Company Securities or other rights to purchase or otherwise acquire any Company Securities, whether vested or unvested, or (ii) obligating the Company to grant, extend, accelerate the vesting or repurchase rights of, change the price of, or otherwise amend or enter into any such Company Option, Company RSU Award, call, right or Contract.

- (g) There is no Indebtedness of the Company (i) granting a holder the right to vote on any matters on which any Company Stockholder may vote (or that is convertible into, or exchangeable for, securities having such right), or (ii) the value of which is in any way based upon or derived from capital or voting stock of the Company, issued or outstanding as of immediately prior to the Closing.
- (h) Except as set forth on Section 3.2(h) of the Disclosure Schedule, there are no Contracts relating to voting, purchase, sale or transfer of any Company Capital Stock or Equity Interests of the Company (i) between or among the Company, on one hand, and any equityholder of the Company, on the other hand, other than written Contracts granting the Company the right to purchase unvested shares upon termination of employment or service, and (ii) to the Knowledge of the Company, between or among any of the equityholders of the Company. Except as set forth on Section 3.2(d)-1 of the Disclosure Schedule, neither the Company Equity Plan nor any Contract of any character to which the Company is a party or by which the Company or any of its assets is bound relating to any Company Options or Company RSU Awards requires or otherwise provides for any accelerated vesting of any Company Options or Company RSU Awards or the acceleration of any other benefits thereunder, in each case in connection with the transactions contemplated by this Agreement or upon termination of employment or service with the Company or Parent, or any other event, whether before, upon or following the Closing or otherwise.
- (i) As of the Closing, (i) the number of shares of Company Capital Stock set forth in the Allocation Schedule as being owned by a Person, or subject to Company Options, Company RSU Awards, or the Company Warrant owned by such Person, shall constitute the entire interest of such Person in the issued and outstanding Company Capital Stock or any other Company Securities, (ii) no Person not disclosed in the Allocation Schedule shall have a right to acquire from the Company any shares of Company Capital Stock, Company Options, Company RSU Awards, Company Warrant, Equity Interests or any other Company Securities, and (iii) the shares of Company Capital Stock, Company Options, Company RSU Awards, and the Company Warrant disclosed in the Allocation Schedule shall be free and clear of any Liens.
- (j) (i) The original application for the PPP Loan submitted by the Company was correct and complete at the time of submission in all respects, and the Company was qualified to receive the original principal amount of the PPP Loan under Law and (ii) to the Knowledge of the Company, the entire outstanding balance of the PPP Loan is forgivable under Law by the PPP Lender. The Company has heretofore made available to Parent a correct and complete copy of the PPP Loan Escrow Agreement together with any and all amendments and supplements thereto and "side letters" and similar documentation relating thereto.
- (k) Section 3.2(k) of the Disclosure Schedule sets forth a correct and complete list, as of immediately prior to the Closing, of all Indebtedness of the Company (other than Unpaid Pre-Closing Taxes), including, for each such item of Indebtedness, the amount owed thereunder, the agreement governing the Indebtedness and a general description of assets securing such Indebtedness. Other than the Indebtedness set forth on Section 3.2(k) of the Disclosure Schedule and Unpaid Pre-Closing Taxes, the Company has no Indebtedness.

3.3. No Other Subsidiaries; Ownership Interests.

(a) As of immediately prior to the Closing, the Company does not, directly or indirectly, own any Equity Interests in or any interest convertible or exchangeable or exercisable for, any Equity Interests, in, any Person other than Shareablee Holdco and Shareablee Merger Sub.

3.4. Authority and Enforceability.

(a) Each of the Company Parties has all requisite power and authority to enter into this Agreement and any Related Agreement to which it is a party and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and any Related Agreement to which any Company Party is a party and the consummation of the transactions contemplated hereby and thereby have been duly authorized by all necessary corporate action on the part of such Company Party and no further corporate action is required on the part of the Company Parties to authorize this Agreement and any Related Agreements to which any Company Party is a party and the transactions contemplated hereby and thereby. The Company Written Consent is the only vote, approval or consent of the holders of any class or series of Company Capital Stock or any other securities of the Company that is necessary under the DGCL, any of the Charter Documents or any Contracts to which any Company Party is a party to adopt this Agreement and each of the Related Agreements and approve the transactions contemplated hereby and thereby and to approve the Combination

Transactions. This Agreement and each of the Related Agreements to which any Company Party is a party have been, or, as of the Closing shall be, duly executed and delivered by such Company Party and assuming the due authorization, execution and delivery by the other parties hereto and thereto, constitute, or shall constitute when executed and delivered, the valid and binding obligations of such Company Party enforceable against it in accordance with their respective terms, subject to (i) laws of general application relating to bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally, and (ii) general principles of equity. The Company Board, by resolutions duly adopted (and not thereafter modified or rescinded), has (x) declared that this Agreement, the Related Agreements and the transactions contemplated hereby and thereby, including the Combination Transactions, upon the terms and subject to the conditions set forth herein, are advisable, fair to and in the best interests of the Company and the Company Stockholders, (y) approved this Agreement in accordance with the provisions of the DGCL, and (z) directed that the adoption of this Agreement and approval of the Combination Transactions be submitted to the Company Stockholders for consideration and recommended that all of the Company Stockholders adopt this Agreement and approve the Combination Transactions (collectively, the "Company Board Resolutions").

- (b) No consent, approval, Order or authorization of, or registration, declaration or filing with, or notice to, any Governmental Entity or any other Person is required by or with respect to the Company Parties in connection with the execution and delivery of this Agreement or the consummation of the transactions contemplated hereby, except for (i) the filing of the Shareablee Merger Certificate, the Shareablee Conversion Certificate, the Initial Merger Certificate and the Final Merger Certificate, (ii) in the case of consummation of the transactions contemplated hereby, obtaining the Company Written Consent, and (iii) such other consents, approvals, Orders, authorizations, registrations, declarations, filings and notices that, if not obtained or made, would not materially and adversely affect, and would not reasonably be expected to materially and adversely affect, the Company Parties' ability to perform or comply with the covenants, agreements or obligations of the Company Parties herein or to consummate the transactions contemplated by this Agreement in accordance with this Agreement and applicable Law.
- (c) There are no "fair price," "moratorium," "control share acquisition," "business combination," "interested shareholder" or other similar anti-takeover statutes, regulations or provisions in the Charter Documents applicable to the Company Parties or to the execution, delivery or performance of the transactions contemplated by this Agreement, including the consummation of the Combination Transactions or any of the other transactions contemplated by this Agreement.
- (d) Each Company Stockholder and Company Warrantholder has irrevocably and unconditionally waived, and has agreed not to exercise any, rights of appraisal or any dissenters' rights that each such Company Stockholder or Company Warrantholder may have (whether under Section 262 of the DGCL or other applicable Law) or could potentially have or acquire in connection with the execution and delivery of this Agreement or the consummation of the Combination Transactions.
- **No Conflict.** The execution and delivery by the Company Parties of this Agreement and any Related Agreement to which any Company Party is a party, and the consummation of the transactions contemplated hereby and thereby, shall not (a) conflict with or result in any violation of or default under (with or without notice or lapse of time, or both) or give rise to, any payment obligation, or a right of termination, cancellation, modification or acceleration of any obligation or loss of any benefit under, or require any consent, approval or waiver from any Person pursuant to, (i) any provision of the Charter Documents, (ii) any Material Contract, or (iii) any Law or Order applicable to the Company or any of its respective properties or assets (whether tangible or intangible), except, in the case of clause (ii) or (iii), as would not reasonably be expected to have a material and adverse effect on the Company, or (b) result in the creation of any Lien (other than a Permitted Lien) on any of the material assets of the Company or any of the shares of Company Capital Stock or Equity Interests of the Company. Section 3.5 of the Disclosure Schedule sets forth all notices, consents, waivers and approvals of parties to any Material Contracts with the Company that are required thereunder in connection with the Combination Transactions, or for any such Material Contract to remain in full force and effect without limitation, modification or alteration immediately after the Closing so as to preserve all rights of, and benefits to, the Company under such Material Contracts from and after the Closing, Immediately following the Closing, the Company shall continue to be permitted to exercise all of its rights under the Material Contracts without the payment of any additional amounts or consideration other than ongoing fees, royalties or payments that the Company would otherwise be required to pay pursuant to the terms of such Material Contracts had the transactions contemplated by this Agreement not occurred.

3.6. No Prior Activities of Shareablee Holdco and Shareablee Merger Sub. Except for obligations incurred in connection with its incorporation or organization or the negotiation and consummation of this Agreement and the transactions contemplated hereby, neither Shareablee Holdco nor Shareablee Merger Sub has incurred any obligation or liability nor engaged in any business or activity of any type or kind whatsoever or entered into any agreement or arrangement with any Person. Each of Shareablee Holdco and Shareablee Merger Sub were formed solely for the purpose of entering into the transactions contemplated by this Agreement.

3.7. Company Financial Statements; No Undisclosed Liabilities.

- (a) The Company has delivered to Parent correct and complete copies of (i) the audited financial statements of the Company as of December 31, 2020 and for the twelve (12) month period then ended (including balance sheets, statements of income and statements of cash flows), (ii) the audited financial statements of the Company as of December 31, 2019 and 2018 and for the years then ended (including balance sheets, statements of income and statements of cash flows), and (iii) the unaudited financial statements of the Company as of September 30, 2021 and for the nine (9) month period then ended (including balance sheets, statements of income and statements of cash flows) ((i), (ii), and (iii) collectively, the "*Financials*"), which are included as Section 3.7(a) of the Disclosure Schedule. The Financials (A) are derived from and in accordance with the books and records of the Company, (B) complied as to form with applicable accounting requirements with respect thereto as of their respective dates, (C) fairly and accurately present in all material respects the financial condition of the Company at the dates therein indicated and the results of operations and cash flows of the Company for the periods therein specified, and (D) were prepared in accordance with GAAP, applied on a consistent basis throughout the periods indicated and consistent with each other.
- (b) The Company does not have any Liabilities of any nature other than (i) those set forth on or adequately provided for on the balance sheet (the "Current Balance Sheet") included in the Financials as of September 30, 2021 (the "Balance Sheet Date"), (ii) those incurred by the Company in connection with the execution of this Agreement (including Transaction Expenses), (iii) those arising under the terms of Material Contracts (other than the payment of liquidated damages or arising as a result of a default or breach thereof), and (iv) those which have arisen after the Balance Sheet Date in the Ordinary Course of Business, none of which is material to the Company. As of the date hereof, the Company has no outstanding guarantees of any debt or other obligation of any other Person. All reserves that are set forth in or reflected in the Current Balance Sheet have been established in accordance with GAAP consistently applied. Except for Liabilities reflected in the Financials, the Company does not have any off-balance sheet Liability of any nature to, or any financial interest in, any third parties or entities, the purpose or effect of which is to defer, postpone, reduce or otherwise avoid or adjust the recording of expenses incurred by the Company.
- The Company has established and maintains a system of internal accounting controls reasonably designed to provide reasonable assurances (i) that transactions, receipts and expenditures of the Company are being executed and made only in accordance with appropriate authorizations of management and the Company Board, (ii) that transactions are recorded as necessary (A) to permit preparation of financial statements in conformity with GAAP, and (B) to maintain accountability for assets, and (iii) that the amount recorded for assets on the books and records of the Company is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences. During the three (3) year period prior to the date hereof, (i) neither the Company nor, to the Knowledge of the Company, any Employee, has identified or been made aware of any fraud or suspected fraud, whether or not material, that involves the Company's management or Employees who have a role in the preparation of financial statements or the internal accounting controls utilized by the Company, or any claim or allegation regarding any of the foregoing; (ii) neither the Company nor, to the Knowledge of the Company, any Representative of the Company has received any material complaint, allegation, assertion or claim regarding deficient accounting or auditing practices, procedures, methodologies or methods of the Company or its internal accounting controls or any material inaccuracy in the Company's financial statements; and (iii) no attorney representing the Company has reported to the Company Board or any committee thereof or to any director or officer of the Company evidence of a material violation of securities laws, breach of fiduciary duty or similar violation by the Company or its Representatives. To the Knowledge of the Company, there are no significant deficiencies or material weaknesses in the design or operation of the Company's internal controls that could materially affect the Company's ability to record, process, summarize and report financial data. There has been no material change in the Company's accounting policies the past three (3) years, except as described in the Financials. The Company does not hold any cash outside of the United States.

3.8. Tax Matters.

(a) General Tax Matters.

(i) The Company has (A) prepared and filed all Tax Returns that it is required to file, and such Tax Returns are true, correct and complete in all material respects, and (B) timely paid all Taxes required to be paid by it (whether or not shown on a Tax Return). The Company is not a beneficiary of any extension of time within which to file any Tax Return other than pursuant to customary extensions of the due date for filing a Tax Return.

(ii) Intentionally Omitted.

- (iii) The Company has (A) withheld or collected with respect to its Employees, stockholders, customers and any other Persons all Taxes (including sales, use, value-added, excise and similar Taxes), social security charges, Federal Insurance Contribution Act amounts, Federal Unemployment Tax Act amounts and other similar charges and fees required to be withheld or collected, and has timely paid over such Taxes, charges, amounts and fees to the appropriate Governmental Entities in accordance with applicable Law, and (B) otherwise complied in all material respects with all applicable Laws relating to the payment, withholding, deduction, collection and remittance of such Taxes, charges, amounts and fees.
- (iv) There is no Tax deficiency outstanding, assessed or to the Knowledge of the Company, proposed against the Company. The Company has not executed any waiver or extension of any statute of limitations for the period of assessment or collection of any Tax, and no request for any such waiver or extension has been filed or is currently pending.
- (v) Except as disclosed in Section 3.8(a)(v) of the Disclosure Schedule, no audit or other examination of any Tax Return of the Company by a Governmental Entity is in progress, nor has the Company been notified in writing of any request for such an audit or other examination. The Company has not received written notice of a claim by any Governmental Entity in a jurisdiction where the Company does not file Tax Returns that the Company is or may be subject to Tax, or required to file Tax Returns, in that jurisdiction. The Company has not received written notice from a Governmental Entity of any proposed adjustments relating to any Tax Return filed by the Company. The Company is not a party to nor bound by any closing or other agreement or ruling with any Governmental Entity with respect to Taxes.

(vi) Intentionally Omitted.

- (vii) The Company made available all income, sales and use, value-added and other Tax Returns filed by the Company for all taxable periods ending after December 31, 2018 and all examination reports and statements of deficiencies, adjustments and proposed deficiencies and adjustments in respect of Taxes of the Company. Section 3.8(a)(vii) of the Disclosure Schedule sets forth each jurisdiction where the Company will be required to file a Tax Return following the Closing with respect to any Pre-Closing Tax Period and the type of each such Tax Return. No power of attorney with respect to Taxes has been granted with respect to the Company that will remain in force after the Closing.
- (viii) There are no Liens on the assets of the Company relating or attributable to Taxes, other than Permitted Liens.
- (ix) The Company is not, and has never been, a "United States real property holding corporation" within the meaning of Section 897(c)(2) of the Code.
- (x) During the past two years, the Company has not constituted either a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of stock intended to qualify for tax-free treatment under Section 355 of the Code.
- (xi) The Company has not (a) participated in any "reportable transaction" or "listed transaction" as defined in Section 6707A(c) of the Code or the U.S. Treasury Regulations promulgated thereunder, (b) engaged in any transaction that would reasonably be likely to require the filing of an IRS Schedule UTP (determined without regard to any asset threshold that may avoid the requirement of filing such schedule), or (c) consummated or participated in any transaction that was or is a "tax shelter" transaction as defined in Section 6662(d)(2)(C)(ii) of the Code or the U.S. Treasury Regulations promulgated thereunder. The

Company has not taken a position on any Tax Return that could reasonably be expected to give rise to a substantial understatement of Tax within the meaning of Section 6662(d) of the Code (or any similar provision of state, local or non-U.S. Tax Law).

- (xii) The Company has not, nor has any predecessor of the Company, (A) been a member of an affiliated group (within the meaning of Section 1504(a) of the Code) filing a consolidated U.S. federal income Tax Return, (B) been a party to, or bound by, any Tax sharing, indemnification, reimbursement or allocation agreement, (C) incurred any potential liability for the Taxes of any Person under U.S. Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or non-U.S. Tax Law, including any arrangement for group or consortium relief or any similar arrangement), as a transferee or successor, by operation of Law, by Contract, or otherwise (other than any Contract or arrangement entered into in the Ordinary Course of Business the primary purpose of which is not Taxes), or (D) been a party to any joint venture, partnership or other agreement or arrangement that is treated as a partnership for U.S. federal income Tax purposes.
- (xiii) Neither the Company, nor Parent as a result of its acquisition of the Company, will be required to include any income or gain or exclude any deduction or loss from taxable income for any Tax period or portion thereof ending after the Closing as a result of (A) any adjustment under Section 481 of the Code (or any corresponding provision of state, local or non-U.S. Tax Law) by reason of a change in a method of accounting, or use of an improper method of accounting or otherwise (including as a result of the transactions contemplated by this Agreement), for a taxable period that ends on or prior to the Closing Date, (B) any "closing agreement" within the meaning of Section 7121 of the Code (or any similar provision of applicable state, local or non-U.S. Law) entered into on or prior to the Closing Date, (C) any installment sale or open transaction disposition made on or prior to the Closing Date, or (D) any deferred revenue or prepaid amount received on or prior to the Closing Date outside of the Ordinary Course of Business.
 - (xiv) The Company uses the accrual method of accounting for income Tax purposes.
- (xv) The Company has not, nor has ever had, a permanent establishment (as defined in any applicable Tax treaty or convention) or an office or fixed place of business, in any country other than the United States.
- (xvi) No Tax ruling, clearance or consent from a Governmental Entity has been issued to the Company, and the Company has not applied for any such Tax ruling, clearance or consent. The Company has not entered into any Contract or arrangement with any Governmental Entity that requires it to take any action or to refrain from taking any action relating to Taxes.
- (xvii) Except as disclosed in Section 3.8(a)(xvii) of the Disclosure Schedule, the Company is and always has been a domestic corporation taxable under subchapter C of the Code for U.S. federal income Tax purposes.
- (xviii) The Company has not (A) deferred the employer's share of any "applicable employment taxes" under Section 2302 of the CARES Act, (B) otherwise deferred any Taxes (including the employee portion of any payroll Taxes) or changed any material Tax election or filed an amended Tax Return under, or in response to, any legislation or executive order enacted or issued in response to COVID-19, (C) claimed any Tax credits under Sections 7001 through 7005 of the Families First Coronavirus Response Act (Public Law 116-127) or Section 2301 of the CARES Act, or (D) sought (nor has any Affiliate that would be aggregated with the Company and treated as one employer for purposes of Section 2301 of the CARES Act sought) a covered loan under paragraph (36) of Section 7(a) of the Small Business Act (15 U.S.C. 636(a)), as added by Section 1102 of the CARES Act.
- (xix) The Company has not taken any action and does not know of any fact, agreement, plan or other circumstance that would reasonably be expected to prevent the Combination Transactions from qualifying for the Intended Tax Treatment.
- (xx) The Company does not have accumulated earnings and profits within the meaning of Section 356(a)(2) of the Code as of the Final Merger Effective Time.
- (b) **Tax Matters related to Compensation**. Neither the execution of this Agreement nor the consummation of the transactions contemplated by this Agreement will (either alone or together with any other

event) result in any payment or benefit that would be, individually or in combination with any other payment or benefit, characterized or reasonably expected to be characterized as an "excess parachute payment" within the meaning of Section 280G of the Code (or any corresponding provision of Tax law). There is no contract, agreement, plan or arrangement to which the Company is a party or by which it is bound to compensate any Person for Taxes paid pursuant to Section 4999 of the Code and the regulations thereunder or any similar state Law. None of the shares of outstanding capital stock of the Company is subject to a "substantial risk of forfeiture" within the meaning of Section 83 of the Code.

- (c) **409A Compliance**. Except as set forth on Section 3.8(c) of the Disclosure Schedule, any nonqualified deferred compensation plan to which the Company is a party is and has been in material documentary and operational compliance with Section 409A of the Code and all applicable IRS guidance issued with respect thereto. The per share exercise price for each Company Option is no less than the fair market value of a share of Company Common Stock on the date of grant (and as of any later modification thereof) within the meaning of Section 409A of the Code and as determined in a manner consistent with Code Treas. Reg. 1.409A-1(b)(5)(iv)(B(2)(i). No compensation is includable in the gross income of any Employee as a result of Section 409A of the Code with respect to any arrangements or agreements covering any Employee. There is no contract, agreement, plan or arrangement to which the Company is a party or by which it is bound to compensate any Person for Taxes paid pursuant to Section 409A of the Code and the regulations thereunder or any similar state Law.
- **3.9. Compliance with Securities Act**. The Company, Company Stockholders and Company Warrantholder are aware that the Parent Common Stock to be issued to the Company Stockholder and the Company Warrantholder pursuant to the transactions contemplated by this Agreement shall constitute "restricted securities" within the meaning of the Securities Act. The Company, Company Stockholders and Company Warrantholder further acknowledge that the Parent Common Stock issued hereunder is acquired for investment only and not with a view to any public distribution thereof, and shall not be sold or otherwise disposed of in violation of any of the registration requirements of the Securities Act.

3.10. Title to Properties; Absence of Liens; Condition and Sufficiency of Assets.

- (a) The Company owns no, and has never owned any, real property.
- (b) Section 3.10(b) of the Disclosure Schedule sets forth a list, as of immediately prior to the Closing, of all leases, lease guaranties, subleases, agreements for the leasing, use or occupancy of, or otherwise granting a right in or relating to all real property currently leased, subleased or licensed by or from the Company or otherwise used or occupied by the Company for the operation of its business (the "Leased Real Property"), including all amendments, terminations and modifications thereof ("Lease Agreements"), and there are no other Lease Agreements for real property affecting the Leased Real Property or to which the Company is bound. There is not, under any of such Lease Agreements, any existing default (or to the Company's Knowledge, any event which with notice or lapse of time, or both, would be reasonably likely to constitute a default) which is or would be material to the Company, and no rent is past due. The Lease Agreements are valid and effective in accordance with their respective terms, subject to (i) laws of general application relating to bankruptcy, insolvency and the relief of debtors, and (ii) rules of law governing specific performance, injunctive relief and other equitable remedies. The Company has not received any written notice of a default, alleged failure to perform, or any offset or counterclaim with respect to any such Lease Agreement, which has not been fully remedied and withdrawn.
- (c) The Company has valid title to, or valid leasehold interest in all of its material properties, and interests in properties and assets, real and personal, reflected on the Current Balance Sheet or acquired after the Balance Sheet Date (except properties and assets, or interests in properties and assets, sold or otherwise disposed of since the Balance Sheet Date in the Ordinary Course of Business), or, with respect to leased properties and assets, valid leasehold interests in such properties and assets that afford the Company valid leasehold possession of the properties and assets that are the subject of such leases, in each case, free and clear of all Liens, except Permitted Liens.
- (d) All material machinery, equipment, and other tangible assets of the Company (other than real property) currently being used in the conduct of the business of the Company have been maintained in all material respects in accordance with generally accepted industry practice (giving due account to the age and length of use and ordinary wear and tear), are in all material respects in good operating condition and repair,

ordinary wear and tear excepted, and are adequate and suitable for the uses to which they are being put. No maintenance of such assets has been materially deferred by the Company.

3.11. Intellectual Property.

- (a) **Definitions**. For all purposes of this Agreement, the following terms shall have the following respective meanings:
- (i) "Company IP" shall mean all Intellectual Property Rights in and to the Company Products, and any other Intellectual Property Rights used in the conduct of the business as currently conducted.
- (ii) "*Company Owned IP*" shall mean any and all Company IP, including Registered IP and Company Software, that are owned or purported to be owned by the Company.
- (iii) "Company Products" shall mean all products and services that have been or are as of immediately prior to the Closing commercially provided, made available, marketed, distributed, offered online, offered for sale, sold, leased, loaned, or licensed by or on behalf of the Company (including through resellers and other channel partners), and any product or service currently under development by or on behalf of the Company that is scheduled for commercial release within one hundred and twenty (120) days following the Closing Date.
- (iv) "*Company Software*" means the software, in source code, object code or bytecode format, and Company Source Code that embodies any Company Owned IP, excluding Open Source Material.
- (v) "Company Source Code" shall mean any software source code owned or purported to be owned by the Company. For the sake of clarity, Company Source Code shall not include Open Source Material.
- (vi) "Contaminant" means any "back door," "drop dead device," "time bomb," "Trojan horse," "virus," "malware," or "worm" (as such terms are commonly understood in the software industry) or any other code, software routines, or hardware components designed or intended to have any of the following functions: (A) disrupting, disabling, harming or otherwise impeding in any manner the operation of, or permitting or causing unauthorized access to, a system, network, or other device or any information contained thereon or processed thereby; or (B) damaging, accessing or destroying any data or file without the user's consent.
- (vii) "Intellectual Property Rights" means (A) all inventions, all improvements thereto, and all patents, patent applications and invention disclosures, worldwide, together with all reissuances, continuations, continuations-in-part, divisions, supplementary protection certificates, extensions and re-examinations thereof; (B) all registered and unregistered trademarks, service marks, trade names, logos, trade dress, slogans and other indicia of source, worldwide, and registrations and applications for registration thereof and any and all goodwill associated therewith; (C) all copyrights in published and unpublished copyrightable works, and all other rights of authorship recognized by statute or otherwise, including without limitation audiovisual works, collective works, computer programs or software (whether in source code or executable form), documentation, compilations, derivative works, literary works, websites, and sound recordings, and all applications, registrations and renewals in connection therewith; (D) all mask works and semiconductor chip rights, and all applications, registrations and renewals in connection therewith; (E) all trade secrets and other confidential information, whether tangible or intangible, including inventions, technical information, processes, drawings, designs, technology, research studies, computer programs, marketing studies, algorithms, customer lists, ideas, formulas, know-how, methods, prototypes, systems, and techniques; (F) databases, data and data compilations and collections and technical data; (G) domain names and uniform resource locators, and all contractual rights to the foregoing; (H) the right to bring any cause of action related to past, present, or future infringement, misappropriation, or violation of the foregoing; and (I) all other proprietary and other rights relating to any or all of the foregoing.
- (viii) "IT Systems" means (A) all computing and/or communications systems and equipment, including any internet, intranet, extranet, e-mail, or voice mail systems and the hardware associated with such systems; and (B) all software, the tangible media on which it is recorded (in any form) and all supporting documentation, data and databases and the hardware associated with such systems.

- (ix) "*Open Source Material*" shall mean any software or other materials that are distributed as "free software" or "open source software" (as such terms are commonly understood in the software industry), including software code or other materials that are licensed under a Creative Commons License, open database license, the Mozilla Public License, the GNU General Public License, GNU Lesser General Public License, Common Public License, Apache License, BSD License, or MIT License and all other licenses identified by the Open Source Initiative as "open source licenses" (such licenses or agreements are collectively, "*Open Source Licenses*").
- (x) "Personal Information" means (A) data in the possession, care, custody or control of the Company (or a third party on behalf of the Company), which could be used to identify or is otherwise associated with an individual person, household or device, including name, address, telephone number, electronic mail address, medical or health information, identification number, health insurance information, location data, an online identifier, bank account number, credit card number, one or more factors specific to the physical, physiological, genetic, mental, economic, cultural or social identity or any other data which may be used to identify an individual, household, or device, and (B) any other information which is classified as "personal data," "personal information," "personally identifiable information" (or similar term) under the Security Policies, Privacy Statements, or applicable Privacy and Security Requirements.
- (xi) "Privacy and Security Requirements" means (A) all Laws governing the Processing of Personal Information or the security of the Company's IT Systems or data, including the following Laws and their implementing regulations: the Health Insurance Portability and Accountability Act, the Federal Trade Commission Act, the CAN-SPAM Act, Canada's Anti-Spam Legislation, the Computer Fraud and Abuse Act, the California Consumer Privacy Act, the Telephone Consumer Protection Act, the Telemarketing and Consumer Fraud and Abuse Prevention Act, and any ancillary rules, binding guidelines, or other binding instruments made or issued by a Governmental Entity under the foregoing instruments, state data security Laws, state data breach notification Laws, the General Data Protection Regulation (EU) 2016/679, any applicable Laws concerning requirements for website and mobile application privacy policies and practices, and Laws governing online marketing or telemarketing; (B) the Company's Privacy Statements; (C) the requirements of any agreements or codes of conduct to which the Company is bound related to the Processing of Personal Information, Third Party Data or the security of the Company's IT Systems or data; and (D) the Payment Card Industry Data Security Standards and, to the extent the Company purports to adhere to any other industry standards, such other industry standards.
- (xii) "*Privacy Statements*" means the Company's internal privacy policies, privacy statements made to customers or data providers (including in Contracts with customers or data providers), publicly posted privacy policies (including if posted on or in connection with the Company's products and services) and any other representation or statement by the Company regarding the confidentiality, Processing, or security of Personal Information.
- (xiii) "*Process*," "*Processing*" or "*Processed*" shall mean any receipt, collection, use, disclosure, retention, storage, disposal, transfer, security, protection, analysis, or other processing of data.
- (xiv) "*Registered IP*" means all U.S. and foreign (A) utility models, supplementary protection certificates, patents and applications for any of the foregoing, (B) registered trademarks, service marks, trade names, logos, trade dress, and slogans and applications to register any of the foregoing, (C) registered copyrights and applications for copyright registrations, and (D) domain name registrations.
 - (xv) "Security Policies" has the meaning set forth in Section 3.11(q)(iii).
- (xvi) "*Shrink-Wrap Software*" shall mean any generally commercially available software in executable code or hosted form that is available for an annual cost of not more than \$15,000 in the aggregate for all users and work stations and not distributed by the Company or incorporated into or used in the Company Products.
- (xvii) "*Third Party Data*" means any and all information, including Personal Information, that is (A) collected by the Company which either: (i) identifies a Company customer or its Employees, visitors or users, (ii) is unique to such customer or its Employees, visitors or users, or (iii) could provide insight into such Employees', users' or visitors' behavior if analyzed, aggregated or otherwise examined; (B) Processed by the Company for or on behalf of a customer, Employee or user; or (C) licensed or received from any other Person.

(b) Registered IP.

- (i) Section 3.11(b)(i) of the Disclosure Schedule lists, as of immediately prior to the Closing: (A) each item of Registered IP which is owned or purported to be owned by the Company or subject to a valid obligation of assignment to the Company (whether owned exclusively, jointly with another Person, or otherwise) ("Company Registered IP"); (B) any other Person that has an ownership interest in such item of Company Registered IP and the nature of such ownership interest; (C) the jurisdiction in which such item of Company Registered IP has been registered or filed and the applicable registration and application numbers; (D) the filing date, and issuance/registration/grant date; (E) the status thereof; and (F) any formal actions that must be taken by the Company within ninety (90) days of the Closing Date with respect to any Company Registered IP, including the payment of any registration, maintenance or renewal fees or the filing of any documents, applications or certificates.
- (ii) All necessary registration, maintenance and renewal fees in connection with the Company Registered IP that are or shall be due for payment on or before the Closing Date have been or shall be timely paid and all necessary documents and certificates in connection with such Company Registered IP that are or shall be due for filing on or before the Closing Date have been or shall be timely filed with the relevant Governmental Entity or third party for the purposes of maintaining such Company Registered IP.
- (c) **Title, Validity and Enforceability**. The Company exclusively owns all right, title, and interest to and in the Company Owned IP free and clear of all Liens (other than Permitted Liens). All Company Registered IP is subsisting and enforceable and, to the Knowledge of the Company, valid.

(d) **Contracts**.

- (i) Section 3.11(d)(i) of the Disclosure Schedule lists (and the Company has made available to Parent a correct and complete copy of), as of immediately prior to the Closing, all licenses or Contracts to which the Company is a party or by which the Company is bound pursuant to which any Intellectual Property Right is licensed to the Company (other than (A) non-exclusive software licenses or software-as-a-service agreements with respect to Shrink-Wrap Software and Open Source Materials, (B) nondisclosure agreements entered in the Ordinary Course of Business, and (C) licenses of Intellectual Property Rights pursuant to $\underline{Section \ 3.11(\underline{d})(\underline{iv})(\underline{D})}$, (collectively "Inbound Licenses").
- (ii) Section 3.11(d)(ii) of the Disclosure Schedule lists (and the Company has made available to Parent a correct and complete copy of), as of immediately prior to the Closing, each license or Contract to which the Company is a party or by which the Company is bound pursuant to which the Company has granted to any Person any license under, agreed not to assert or enforce, or in which any Person has otherwise received or acquired any right (whether or not currently exercisable) or interest in or to, any Company Owned IP (other than (A) nondisclosure agreements entered in the Ordinary Course of Business; (B) nonexclusive licenses to provide the Company Products to the Company' customers entered in the Ordinary Course of Business; and (C) access or licenses to Company IP granted to Employees pursuant to Personnel Agreements (collectively, "Outbound Licenses").
- (iii) Prior to the date hereof, the Company has delivered to Parent evidence of Approved Data Consents for the Data Partners listed on Schedule C-1(a). With respect to each of its existing agreements with the Data Partners, the Company (x) does not currently and has not previously infringed, misappropriated, or violated any Intellectual Property Rights of any of the Data Partners, (y) is compliant in all material respects with all material obligations thereunder, including any limits or restrictions on data Processing and compliance with all relevant data privacy laws and regulations, and (z) has not received any written or, to the Knowledge of the Company, oral communications from any of the Data Partners questioning or raising concerns about the Company's use of such data partner's data.
- (iv) The Company has made available to Parent a correct and complete copy of each standard form of Contract prepared and provided by the Company to a Person at any time in connection with the Company's business, including each of the Company's unmodified standard forms of: (A) director or employee agreement containing any assignment or license of Intellectual Property Rights or any confidentiality provision; (B) consulting or contractor agreement containing any assignment or license of Intellectual Property Rights or any confidentiality provision; (C) confidentiality or nondisclosure agreement; and (D) customer contract providing for non-exclusive use of or access to the Company Products (collectively, the "Standard Form

- **Agreements**"). For clarity, the Standard Form Agreements do not include agreements under which the Company has licensed or received any rights in or to data from another Person.
- (e) **Disclosure of Company Owned IP**. The Company has the exclusive right to bring infringement actions with respect to the Company Owned IP. The Company has not transferred full or partial ownership of, or granted any exclusive license with respect to, any Intellectual Property Rights that are or, as of the time of such transfer or exclusive license were, owned or purported to be owned by the Company.
- (f) **Development of Company Owned IP**. The Company has not jointly developed any Company Owned IP with any other Person. Without limiting the generality of the foregoing:
- (i) Each Employee that developed any Intellectual Property Rights for or on behalf of the Company has signed a valid, enforceable agreement (A) containing an assignment to the Company of all of such Person's rights, title and interests in and to any Intellectual Property Rights and a waiver of such Person's non-assignable rights (including moral rights), and (B) containing customary confidentiality provisions protecting the rights of the Company in Company Owned IP (such agreements, the "*Personnel Agreements*"). The Company and, to the Knowledge of the Company, all other parties thereto are in compliance in all material respects with the provisions of the Personnel Agreements.
- (ii) No Employee (A) has made any written claim of ownership, or, to the Knowledge of the Company, any oral claim of ownership with respect to any Company Owned IP, or (B) has any reasonable basis for a claim, right (whether or not currently exercisable) or interest to or in any Company Owned IP.
- (iii) To the Knowledge of the Company, no Employee is: (A) bound by or otherwise subject to any Contract with a Person restricting such Employee from performing (or in the case of former Employees of the Company, having performed) such Employee's duties for the Company, or (B) in breach of any Contract with any former employer or other Person concerning Intellectual Property Rights or confidentiality due to his/her activities as an Employee.
- (iv) No funding, facilities or personnel of any Governmental Entity or any public or private university, college or other educational or research institution were used to develop or create, in whole or in part, any Intellectual Property Right for or on behalf of the Company.
- (g) **Standards Bodies**. The Company is not, nor has ever been, a member or promoter of, or a contributor to, any industry standards body or any similar organization that requires or obligates the Company to grant or offer to any other Person any license or right to any Company Owned IP. The Company is not a party to any Contract with a standards body or any similar organization involving the license to the Company of "standards essential" Intellectual Property Rights, and the Company has no reasonable basis to believe that in the absence of such Contract, it has any Liability therefor.
- (h) **Protection of Trade Secrets**. The Company has taken commercially reasonable steps and precautions necessary consistent with general industry practice to maintain the confidentiality of, and otherwise protect and enforce its rights in, all proprietary and confidential information that the Company holds, or purports to hold, as a trade secret or maintains, or purports to maintain, as confidential.
- (i) **Enforcement**. To the Company's Knowledge no Person has infringed, misappropriated or otherwise violated, and no Person is currently infringing, misappropriating or otherwise violating, any Company Owned IP. Neither the Company nor its Representatives have sent or otherwise delivered to any Person any letter or other written or electronic communication or correspondence or claim regarding any actual, alleged or suspected infringement or misappropriation in any material respect of any Company Owned IP.
- (j) **Non-Infringement**. The operation of the business of the Company as currently conducted and as contemplated to be conducted has not and does not infringe, violate or misappropriate, any Intellectual Property Rights of any Person, or constitute unfair competition or trade practices under applicable Law. Without limiting the generality of the foregoing:
- (i) No infringement, misappropriation, or similar claim or Legal Proceeding involving or relating to any Intellectual Property Rights of another Person (a "*Claimant*") is pending or, to the Knowledge of the Company, currently threatened against the Company or, to the Knowledge of the Company,

against any other Person, in each case, alleging that the Claimant is entitled to be indemnified, defended, held harmless or reimbursed by the Company.

- (ii) The Company has not received any notice or other communications (in each case, in writing or, to the Knowledge of the Company, otherwise) relating to any actual, alleged or suspected infringement, misappropriation or violation by the Company of any Intellectual Property Rights of another Person, including any letter or other communication suggesting or offering that the Company obtain a license to any Intellectual Property Rights of another Person.
- (k) Contaminants. The Company Products, Company Software and IT Systems do not contain any Contaminant (including relating to or resulting from the display, manipulation, or Processing of data) or defect that materially and adversely affects the use, availability, integrity, functionality or performance of such Company Product, Company Software, IT Systems or any product or system containing or authorized or intended to be used in conjunction with such Company Software, Company Product or IT Systems or causes or would cause the Company to fail to comply, in any material respect, with any applicable warranty or other contractual commitment covering the use, functionality or performance of such Company Product, Company Software, IT Systems or any product or system containing or authorized or intended to be used in conjunction with such Company Product, Company Software, or IT Systems. The Company has implemented and maintains reasonable measures used in its industry designed to prevent the introduction of Contaminants into the Company Products, Company Software and IT Systems, including firewall protections and regular virus scans and for taking and storing on-site and off-site back-up copies of Company Software and critical data. The Company has implemented a vulnerability management program consistent with general industry practice. In the past three (3) years, the Company has documented and remediated all known material Contaminants and defects in all the Company Products, Company Software and IT Systems, and such documentation is retained and is available internally at the Company.
- (l) **Company Source Code**. No Company Source Code has been delivered, licensed or made available to any escrow agent or other Person who is not, or was not, as of the date thereof, an employee, consultant or contractor of the Company subject to a Personnel Agreement. The Company has no duty or obligation (whether present or contingent) to deliver, license or make available any Company Source Code to any escrow agent or other Person. No event has occurred, and no circumstance or condition exists (including the consummation of the transactions contemplated hereby), that shall or could reasonably be expected to result in the delivery, license, or disclosure of any Company Source Code to any other Person.

(m) Open Source.

- (i) Section 3.11(m)(i) of the Disclosure Schedule lists, as of immediately prior to the Closing: (A) each item of Open Source Materials that is contained in, distributed with, or linked with the current version of any Company Software or Company Product or from which any part of the current version of any Company Software or Company Product is derived; (B) the name of or a link to (or other locator for) the applicable Open Source License for each such item of Open Source Material; (C) the Company Product to which each such item of Open Source Material relates; and (D) whether such Open Source Material has been modified by or on behalf of the Company.
- (ii) No Company Software or Company Product contains, is linked with, derived from, or is distributed with, any Open Source Materials in a manner: (A) that requires that the Company grant a license under, or refrain from asserting any one or more of its Intellectual Property Rights or that any Company Software or part thereof (except for the applicable, unmodified third party Open Source Material itself): (1) be disclosed or distributed in source code form; (2) be licensed for the purpose of making modifications or derivative works; or (3) be redistributable at no charge; or (B) that otherwise imposes any limitation, restriction or condition on the Company's commercial exploitation of any Company Software or part thereof (in each case (A) and (B), except for the applicable, unmodified third party Open Source Material itself).
 - (iii) The Company has not distributed any Company Source Code pursuant to an Open Source License.

(n) Privacy.

(i) <u>Compliance with Privacy Statements</u>. The Company has provided notice of its privacy practices in its Privacy Statements in accordance with applicable Privacy and Security Requirements.

No Privacy Statement, or disclosure made or contained in any of the Privacy Statements is, or has been, inaccurate, misleading or deceptive in violation of the Privacy and Security Requirements. The Company's privacy and security practices conform, and at all times have conformed, in all material respects to the then-current Privacy Statements.

- (ii) <u>Disclosure of Privacy Statement</u>. Section 3.11(n)(ii) of the Disclosure Schedule lists each current Privacy Statement of the Company. The Company has, at all times, posted Privacy Statements on the Company's websites, mobile applications, or where otherwise required under applicable Privacy and Security Requirements.
- (iii) <u>Contractual Obligations</u>. The Company has contractually obligated all Persons that have access to Personal Information to comply with the Security Policies and Process Personal Information in a manner sufficient to meet the Company's obligations under the Privacy and Security Requirements and the Company's other contractual obligations applicable to Personal Information, including any confidentiality obligations. To the Knowledge of the Company, no third party to whom the Company has provided access to Personal Information has failed to comply with such confidentiality obligations, Security Policies, Privacy Statements, or applicable Law, including the applicable Privacy and Security Requirements. The Company has not been and is not in breach, nor has it received notice of a breach, of any contractual obligation relating to the privacy, security, or Processing of Personal Information.
- (iv) <u>Compliance with Privacy and Security Requirements</u>. The Company currently and has at all times: (A) complied in all material respects with (1) all applicable Privacy and Security Requirements governing the Processing of Personal Information and Third Party Data, and (2) requirements of self-regulatory organizations that apply to the Company and that the Company has agreed or represented to comply with in writing (including, to the extent applicable, the Payment Card Industry Data Security Standards), (B) implemented and maintained commercially reasonable measures that provide reasonable assurance that the Company complies with applicable Privacy and Security Requirements and that the Company will not acquire, fail to secure, or Process such Personal Information or Third Party Data in a manner that (1) violates applicable Privacy and Security Requirements, (2) is inconsistent with any notice to or consent from the provider of Personal Information or Third Party Data, (3) violates any policy adopted by the Company, (4) breaches any requirement of or contractual commitment made by the Company that is applicable to such Personal Information or Third Party Data, or (5) violates any Privacy Statement.
- (v) <u>Effects of Execution of Agreement</u>. Neither the execution, delivery nor performance of this Agreement nor the consummation of transactions contemplated hereby shall result in any violation of any Privacy Statement of the Company, any contractual obligation of the Company governing the privacy or security of Personal Information, or any applicable Privacy and Security Requirements, which violation would have a material and adverse effect on the Company.
- (vi) Third Party Data and Personal Information. During the past six (6) years, the Company has: (A) provided adequate notice and obtained any necessary consents required for the Processing of Personal Information and Third Party Data under applicable Privacy and Security Requirements, the Security Policies, and Privacy Statements; (B) complied with any contractual obligation, agreement, permit, license, government filing or other obligation regarding the Processing of Personal Information or Third Party Data; (C) complied with all verifiable data subject requests made by data subjects concerning their Personal Information; and (D) abided by any verifiable and applicable opt-outs related to Personal Information. The Company has not, and currently does not send any communications for marketing purposes, or that include marketing content, to individuals via text.
- (vii) <u>Audits, Claims, Complaints</u>. The Company has not received any written claims, notices or complaints, and there are no Legal Proceedings, regarding the Company's privacy, data protection, or information security practices or the Processing of any Personal Information. The Company has not received notice of, and the Company has no Knowledge of, any circumstance (including any circumstance arising as the result of a Legal Proceeding or an audit or inspection carried out by any other Person) that has given rise to or would reasonably be expected to give rise to, any notice, court order, warrant, regulatory opinion, audit result, or allegation, from a Governmental Entity, tribunal or court in any jurisdiction or any other Person: (A) alleging or confirming non-compliance with any applicable Privacy and Security Requirements or Privacy Statements; (B) requiring or requesting the Company to amend, rectify, cease Processing, de-combine, permanently anonymize, block, or delete any Personal Information, or to decommission or alter the exploitation or operation of the Company's business, in whole or in part; (C) prohibiting or threatening to prohibit the transfer of

Personal Information to any place; or (D) permitting or mandating any Governmental Entity to investigate, request information from, or enter the premises of, the Company. The Company is not under investigation by any Governmental Entity for a violation of any applicable Privacy and Security Requirements. The Company has not received any notice of or request for any Legal Proceeding pertaining to its policies or practices in any way relating to the Processing of Personal Information. The Company has not acted or failed to act in a manner that would trigger a notification or reporting requirement to any Person or Governmental Entity under any Contract or applicable Privacy and Security Requirements.

- (o) **Disputes**. The Company has not been involved in a dispute or Legal Proceeding in respect of any actual or alleged violation of applicable Privacy and Security Requirements or Privacy Statements. The Company has not received, and the Company has no Knowledge of any circumstance that would reasonably be expected to give rise to, any notice from a Person claiming a right to compensation, indemnification or other relief against the Company under an applicable Privacy and Security Requirement or Privacy Statement.
- (p) **Training.** All of the Employees who have access to Personal Information, including Personal Information that is subject to applicable Privacy and Security Requirements, have received professional and appropriate training with respect to compliance with such applicable Privacy and Security Requirements and obligations under Privacy Statements or the Company's contractual obligations relating to privacy, cybersecurity, data protection, Third Party Data, or Personal Information.

(q) **Security Measures**.

- (i) <u>Implementation of Security Measures and Written Program</u>. The Company has implemented and maintained reasonable security practices, policies, controls and other measures reasonably designed to ensure the security, confidentiality, integrity and availability of IT Systems, Personal Information and Third Party Data, and to protect IT Systems, Personal Information and Third Party Data from any Security Incident, Contaminant, loss, theft and interruption, and any unauthorized Processing. Such security practices, policies, controls and other measures are, without limitation, consistent in all material respects with applicable Privacy and Security Requirements, customary industry practices, and any written contractual obligations to third parties.
- (ii) <u>Security Incidents</u>. The Company has not experienced any actual or reasonably suspected breach of security, unlawful, unauthorized or accidental loss, destruction, modification, acquisition, unavailability, or Processing of, or access to, IT Systems or Personal Information or Third Party Data (a "*Security Incident*"). To the Knowledge of the Company, no vulnerabilities exist in the Company Products which present a risk of a Security Incident. No circumstance has arisen: (A) in which applicable Privacy and Security Requirements would require the Company to notify a Governmental Entity, the media, any of its Employees, customers or any other Person of a Security Incident, or (B) in which the Company has notified any of the foregoing of a Security Incident.
- (iii) <u>Compliance with Security Policies</u>. The Company has implemented and maintained reasonable written information security policies and programs that govern the Processing, security, confidentiality, integrity and availability of IT Systems, Personal Information and Third Party Data (collectively, "Security Policies"). The Company has at all times complied with such Security Policies. The Security Policies and the implementations of such policies: (A) identify internal and external risks to the security of the IT Systems, Personal Information and Third Party Data; (B) implement customary industry practices regarding administrative, electronic and physical safeguards to control those risks and safeguard the security, confidentiality, integrity, and availability of the IT Systems, Personal Information and Third Party Data (including on the systems of Persons with access to such IT Systems, Personal Information and Third Party Data (including on the systems of Persons with applicable Privacy and Security Requirements in the case of any Security Incident; and (E) comply with all applicable Privacy and Security Requirements, the Privacy Statements, and the Company's contractual obligations relating to privacy, cybersecurity, data protection, Personal Information and Third Party Data.
- (iv) <u>IT Systems</u>. The IT Systems are reasonably sufficient for the operation of the business of the Company as currently conducted. The IT Systems are in sufficient working condition to effectively perform all computing, information technology and data Processing operations necessary for the operation of the Company as currently conducted. The Company has not experienced any material disruption

to, or material interruption in, the conduct of its business attributable to a defect, error, or other failure or deficiency of any IT System. The Company has implemented and routinely tested commercially standard business continuity and disaster recovery plans relating to its IT Systems.

(r) **Effect of Transaction**. All Company Owned IP immediately prior to the Closing is (and immediately following Closing shall be) fully transferable, alienable and licensable by the Company without restriction and without payment of any kind to any Person in each case, in the same manner as such Company Owned IP was transferable, alienable and licensable immediately prior to the Closing. Neither the execution, delivery or performance of this Agreement or any other agreements referred to in this Agreement nor the consummation of any of the transactions contemplated hereby will, with or without notice or lapse of time, result in, or give any other Person the right or option to cause or declare: a loss of, or Lien on, any Company Owned IP or any other Intellectual Property Rights incorporated into or used in the operations of the Company's business as currently conducted; an obligation for Parent to offer any discount or be bound by any "most favored pricing" or "most favored access" terms under any Contract to which the Company is a party or bound; or the release, disclosure or delivery of any Company IP by or to any escrow agent or other Person; or the grant, assignment or transfer to any other Person of any license or other right or interest in, under, or with respect to, any of the Company IP or any other Intellectual Property Rights of Parent.

3.12. Agreements, Contracts and Commitments.

- (a) Schedule 3.12(a) of the Disclosure Schedule sets forth a list of the Contracts the Company is a party to as of immediately prior to the Closing, except for this Agreement, the Related Agreements and the Contracts specifically identified on Section 3.11(d) of the Disclosure Schedule:
- (i) (A) any type of employment, contractor or consulting Contract used by the Company with respect to any current Employee (other than with respect to administrative staff), (B) any Contract to grant any severance, change of control payments, retention bonus, or termination pay (in cash or otherwise) to any current Employee, (C) any other form of Employee Agreement with a current Employee (other than with respect to administrative staff, any Personnel Agreements or any individual agreements pursuant to the forms of employment, contractor or consulting agreements provided under Section 3.12(a) (i)(A)), (D) each employment agreement or offer letter that is not immediately terminable at-will by the Company without advance notice, severance, or other cost or liability, (E) any form bonus or commission plan, or any bonus agreements or commission agreements or commitments with any current Employee, (F) any separation agreement or settlement agreement with any current or former employee pursuant to which the Company has any current actual or potential Liability or ongoing obligation, and (G) any settlement agreement, consent decree, or other similar agreement with any Governmental Entity, (1) pursuant to which claims for harassment or discrimination were specifically released; or (2) under which the Company has any current actual or potential Liability;
- (ii) any Contract or plan, including any stock option or incentive plan, stock appreciation rights plan or stock purchase plan, any of the benefits of which shall be increased, or the vesting of benefits of which shall be accelerated or may be accelerated, solely based on the occurrence of any of the transactions contemplated by this Agreement or the value of any of the benefits of which shall be calculated solely on the basis of any of the transactions contemplated by this Agreement;
 - (iii) any lease of any real property or material personal property;
- (iv) any Contract relating to capital expenditures and involving future payments in excess of \$40,000 in any fiscal year;
- (v) any mortgages, indentures, guarantees, loans or credit agreements, security agreements or other agreements or instruments relating to the borrowing of money or extension of credit or other Indebtedness;
- (vi) any purchase order or Contract for the purchase of tangible items of equipment or related services in any amount in excess of \$40,000 in any fiscal year;
 - (vii) any Inbound License;
 - (viii) any Outbound License;

- (ix) any confidentiality and non-disclosure agreements (whether the Company is the beneficiary or the obligated party thereunder), other than those related to commercial transactions in the Ordinary Course of Business;
- (x) any Contract providing for "offshoring" or outsourcing development of any material items of Company Software by, for or on behalf of the Company;
 - (xi) any Contract with federal, state, city, county, parish, municipal or other Governmental Entities;
- (xii) (A) any management service, legal partnership or joint venture Contract, and (B) any Contract that involves a sharing of revenues, profits or cash flows with other Persons;
- (xiii) any agency, dealer, distribution, sales representative, remarketer, reseller, or other Contract for the distribution of Company Products;
- (xiv) any Contract pursuant to which the Company is bound to or has committed to provide any product or service to any third party on a "most favored nation," "most favored pricing," or "most favored access" basis, or similar terms;
- (xv) any Contract granting any license or other rights in, to or from the Company with respect to Personal Information or Third Party Data;
- (xvi) any Contract pursuant to which the Company has acquired, or has agreed to acquire, a business or entity, or a material portion of the assets of a business or entity, whether by way of merger, consolidation, purchase of stock, purchase of assets, exclusive license or otherwise, or any Contract pursuant to which it has any material ownership interest in any other Person:
 - (xvii) any agreement of indemnification with Employees of the Company;
- (xviii) any Contract with any investment banker, broker, advisor or similar party, or any accountant, legal counsel or other Person retained by the Company, in connection with this Agreement and the transactions contemplated hereby;
- (xix) any Contract or other arrangement to settle any Legal Proceeding or to settle any threatened or reasonably anticipated Legal Proceeding; and
- (xx) any other Contract, without duplication of the foregoing, that involves the payment to or from the Company in excess of \$40,000 in any fiscal year.
- (b) The Company has made available correct and complete copies of each Contract required to be disclosed pursuant to <u>Sections 3.2</u>, <u>3.10</u>, <u>3.11</u> (except for each Contract entered into on a Standard Form Agreement), <u>3.12</u> and <u>3.19(a)</u>. For the purposes of this Agreement, each of the foregoing Contracts referenced in this <u>subsection 3.12(b)</u> shall each be a "*Material Contract*" and collectively are the "*Material Contracts*."
- (c) The Company has performed in all material respects all of the obligations required to be performed by it and is entitled to all benefits under, and has not received written or, to the Knowledge of the Company, oral notice of being in default in respect of, any Material Contract. Each of the Material Contracts is valid, binding and enforceable against the Company and the other parties thereto, and is in full force and effect, subject only to the effect, if any, of applicable bankruptcy and other similar Laws affecting the rights of creditors generally and rules of Law governing specific performance, injunctive relief and other equitable remedies. There exists no material default or event of default or event, occurrence, condition or act, with respect to the Company, that, with the giving of notice, the lapse of time or the happening of any other event or condition, would reasonably be expected to (i) become a material default or material event of default under any Material Contract, or (ii) give any third party (A) the right to declare a material default or exercise any material remedy under any Material Contract, (B) the right to a material rebate, chargeback, refund, credit, penalty or change in delivery schedule under any Material Contract, (C) the right to cancel, terminate or modify any Material Contract. The Company has not received any written notice or any other written communication regarding, and to the Knowledge of the Company, is not aware of, any actual or possible

violation or breach of, default under, or intention of any party thereto to cancel or modify any Material Contract. The Company has heretofore made available to Parent (1) correct and complete copies of each written Material Contract together with any and all material amendments and supplements thereto and "side letters" and similar documentation relating thereto, and (2) summaries of each oral Material Contract, if any.

3.13. Interested Party Transactions.

- (a) Except as disclosed on Section 3.13(a) of the Disclosure Schedule, no Employee, or, to the Knowledge of the Company, any Company Stockholder (or, to the Knowledge of the Company, any of their members, directors, officers, employees, Affiliates, "associates," or members of any of their "immediate family" (as such terms are respectively defined in Rule 12b-2 and Rule 16a-1 of the 1934 Act)) (each, an "*Interested Party*"), has or has had, directly or indirectly, (i) any interest in any entity which furnished or sold, or furnishes or sells, services, products, or technology that the Company furnishes or sells, or proposes to furnish or sell, (ii) any interest in any entity that purchases from or sells or furnishes to the Company, any goods or services, (iii) any interest in, or is a party to, any Contract to which the Company is a party (except for any Contract relating to normal compensation or welfare benefits provided for services as an Employee), (iv) is the beneficiary of any management or other similar fees paid by the Company, or (v) is indebted to or has borrowed money from or lent money to, or is a guarantor or indemnitor of the Company (other than any such indebtedness, guarantee or indemnity that will be discharged or extinguished at or prior to Closing); *provided*, that ownership of no more than two percent (2%) of the outstanding voting securities of a publicly traded entity shall not be deemed to be an "interest in any entity" for purposes of this Section 3.13. Section 3.13(a) of the Disclosure Schedule sets forth, as of immediately prior to the Closing, each Interested Party that holds any Company Securities the vesting of which shall accelerate upon the consummation of the transactions contemplated by this Agreement.
- (b) All transactions pursuant to which any Employee, or, to the Knowledge of the Company, any other Interested Party, has purchased any services, products, or technology from, or sold or furnished any services, products or technology to, the Company that were entered into on or after the inception of the Company, have been on an arms'-length basis on terms no less favorable to the Company than would be available from an unaffiliated party.
- **3.14. Company Authorizations.** Section 3.14 of the Disclosure Schedule sets forth, as of immediately prior to the Closing, each consent, license, permit, grant or other authorization (a) pursuant to which the Company currently operates, provides any services or holds any interest in any of its material properties, or (b) which is required for the operation of the business of the Company as currently conducted or the holding of any such interest, except for any consent, license, permit, grant or other authorization the lack of which would not reasonably be expected to have a material and adverse effect on the Company, taken as a whole (collectively, including any such items that are required to be disclosed in Section 3.14 of the Disclosure Schedule, "**Company Authorizations**"). All of the Company Authorizations have been issued or granted to the Company, are in full force and effect and constitute all Company Authorizations required to permit the Company to operate or conduct its business as currently conducted or operated or hold any interest in its properties or assets, except for any immaterial Company Authorization. The Company is in compliance in all material respects with all such Company Authorizations, and, as of immediately prior to the Closing, shall have applied for and not been denied renewals of such Company Authorizations.

3.15. Litigation.

- (a) Except as disclosed on Section 3.15(a) of the Disclosure Schedule, there is no Legal Proceeding of any nature pending, or to the Knowledge of the Company, threatened, against the Company, its properties or assets (tangible or intangible) or any of the Employees (in their capacities as such). To the Knowledge of the Company, no Governmental Entity has at any time challenged or investigated the legal right of the Company to conduct its respective operations as presently or previously conducted. To the Knowledge of the Company, there is no Legal Proceeding of any nature pending or threatened, against any Person who has a contractual right or a right pursuant to laws of the State of Delaware to indemnification from the Company related to facts and events existing prior to the Shareablee Merger Effective Time.
- (b) To the Knowledge of the Company, there is no reasonable basis for any Person to assert a claim against the Company, its properties or assets (tangible or intangible) or any of the Employees (in their capacities as such) based upon: (i) the Company entering into this Agreement, the Related Agreements or any of the transactions contemplated hereby or thereby, including a claim that such Employee breached a fiduciary

duty in connection therewith, (ii) any confidentiality or similar agreement entered into by the Company regarding its assets, or (iii) any claim that the Company has agreed to sell or dispose of any of its assets to any party other than Parent, whether by way of merger, consolidation, sale of assets or otherwise.

- (c) The Company has no Legal Proceeding of any nature pending against any other Person.
- **3.16. Books and Records**. The Company has made available to Parent correct and complete copies of (a) all documents identified on the Disclosure Schedule, (b) the Charter Documents, each as currently in effect, and (c) the minute book containing records of proceedings, consents, actions and meetings of the board directors and the stockholders of the Company. At the Closing, the minute book shall be in the possession of the Company.
- **3.17.** Environmental, Health and Safety Matters. The Company is in compliance in all material respects with all Environmental, Health and Safety Requirements in connection with the ownership, use, maintenance and operation of its business, assets and properties. There are no pending, or to the Knowledge of the Company, threatened allegations by any Person that the properties or assets of the Company are not, or that its business has not been conducted, in compliance in all material respects with all Environmental, Health and Safety Requirements. The Company has not retained or assumed any Liability of any other Person under any Environmental, Health and Safety Requirements. There are no past or present facts, circumstances or conditions that would reasonably be expected to give rise to any material Liability of the Company with respect to Environmental, Health and Safety Requirements.
- **3.18. Brokers' and Finders' Fees**. The Company has not incurred, and shall not incur, directly or indirectly, any Liability for brokerage or finders' fees or agents' commissions, fees related to investment banking or similar advisory services or any similar charges in connection with this Agreement or any transaction contemplated hereby, nor shall Parent or the Company incur, directly or indirectly, any such Liability based on arrangements made by or on behalf of the Company.

3.19. Employee Benefit Plans and Compensation.

- (a) **Schedule**. Section 3.19(a) of the Disclosure Schedule contains a correct and complete list, as of immediately prior to the Closing, of each Company Employee Plan and each Employee Agreement; *provided*, that the Company may list only the applicable form(s) of offer letter(s) or employment agreement(s) for current Employees, and need not list each individual offer letter on such form(s) *provided* that individual agreements that materially deviate from such form(s) shall be listed separately. The Company has not made any commitment or communicated in writing or, to the Knowledge of the Company, orally any intent to establish or enter into any new Company Employee Plan or Employee Agreement, to modify any Company Employee Plan or Employee Agreement (except to the extent required by Law or to conform any such Company Employee Plan or Employee Agreement to the requirements of any applicable Law, in each case as previously disclosed to Parent in writing, or as required by this Agreement) or the terms of the applicable Company Employee Plan or Employee Agreement. No Company Employee Plan is self-funded.
- (b) **Documents**. The Company has made available to Parent (i) correct and complete copies of all documents embodying each Company Employee Plan and each Employee Agreement, in each case, including all amendments thereto and all related trust documents and plan summaries, (ii) the most recent annual report (Form 5500 and all schedules and financial statements attached thereto), if any, required under ERISA or the Code or by any other applicable Law in connection with each Company Employee Plan, (iii) the most recent summary plan description together with the summary(ies) of material modifications thereto, if any, required under ERISA or by any other applicable Law with respect to each Company Employee Plan, (iv) all current material written agreements and contracts relating to each Company Employee Plan, (v) all material correspondence to or from any Governmental Entity relating to any Company Employee Plan other than routine correspondence in the normal course of operations of such Company Employee Plan, (vii) current policies pertaining to fiduciary liability insurance covering fiduciaries for each Company Employee Plan, if any, and (viii) the most recent IRS (or any other applicable Tax authority) determination or opinion letter issued with respect to each Company Employee Plan for which determination letters are currently available.
- (c) **Employee Plan Compliance**. The Company has performed in all material respects all obligations required to be performed by it under, is not in material default or violation of, and, as of the date hereof, has no Knowledge of any default or violation by any other party to, any Company Employee Plan, and

each Company Employee Plan has been established and maintained in accordance in all material respects with its terms and in material compliance with all applicable Laws, including ERISA and the Code. To the Knowledge of the Company, any Company Employee Plan intended to be qualified under Section 401(a) of the Code is so qualified and has obtained a favorable IRS determination letter (or opinion letter, if applicable) as to its qualified status under the Code, and to the Knowledge of the Company there has been no event or condition that has adversely affected or could reasonably be expected to adversely affect such qualified status. No "prohibited transaction," within the meaning of Section 4975 of the Code or Sections 406 and 407 of ERISA, and not otherwise exempt under Section 408 of ERISA, has occurred with respect to any Company Employee Plan. There are no Legal Proceedings pending or to the Knowledge of Company, threatened (other than routine claims for benefits) against any Company Employee Plan or against the assets of any Company Employee Plan. There are no audits, inquiries or Legal Proceeding pending or to the Knowledge of the Company, threatened by the IRS, DOL, or any other Governmental Entity with respect to any Company Employee Plan. The Company is not subject to any penalty or Tax with respect to any Company Employee Plan under Section 502(i) of ERISA or Sections 4975 through 4980 of the Code. Except as set forth on Section 3.19(c) of the Disclosure Schedule, the Company has made all contributions and other payments required by and due under the terms of each Company Employee Plan. Each Company Employee Plan can be amended, terminated or otherwise discontinued on or after the Shareablee Merger Effective Time in accordance with its terms, without Liability to Parent or the Company (other than ordinary administration expenses).

- (d) **No Pension Plan**. The Company has never maintained, established, sponsored, participated in, or contributed to, any Pension Plan subject to Title IV of ERISA or Section 412 of the Code.
- (e) **No Self-Insured Plan**. The Company has never maintained, established, sponsored, participated in or contributed to any self-insured plan that provides benefits to Employees (including any such plan pursuant to which a stop loss policy or contract applies).
- (f) Collectively Bargained, Multiemployer and Multiple Employer Plan. At no time has the Company of any ERISA Affiliate contributed to or been obligated to contribute to or had any material liability with respect to any multiemployer plan (as defined in Section 3(37) of ERISA), "funded welfare plan" (within the meaning of Section 4119 of the Code), "multiple employer welfare arrangement" (within the meaning of Section 3(40) of ERISA), "voluntary employees' beneficiary association" (within the meaning of Section 501(c)(9) of the Code), or other plan subject to Title IV of ERISA or subject to Section 412 of the Code or Section 302 of ERISA. The Company has never maintained, established, sponsored, participated in or contributed to any multiple employer plan or to any plan described in Section 413 of the Code.
- Ro Post-Employment Obligations. No Company Employee Plan or Employee Agreement provides, or reflects or represents any Liability to provide, post-employment or retiree life insurance, health or other employee welfare benefits to any Person for any reason, except as may be required by COBRA or other applicable Law, and the Company is not party to any written or oral Contract pursuant to which any Employee (either individually or to Employees as a group) or such Employee's dependents would be provided with post-employment or retiree life insurance, health or other employee welfare benefits, except to the extent required by applicable Law. The Company and each ERISA Affiliate have each complied in all material respects with the notice and continuation coverage requirements, and all other requirements, of Section 4980B of the Code and Parts 6 and 7 of Title I of ERISA, and the regulations thereunder, with respect to each Company Employee Plan that is, or was during any Tax year for which the statute of limitations on the assessment of federal income Taxes remains open, by consent or otherwise, a group health plan within the meaning of Section 5000(b)(1) of the Code.
- (h) **Health Plans**. The Company and each Company Employee Plan that is a "group health plan" as defined in Section 733(a)(1) of ERISA (each, a "*Health Plan*") is and has been in compliance, in all material respects, with the Patient Protection and Affordable Care Act of 2010 ("*PPACA*"), and no event has occurred, and no condition or circumstance exists, that could reasonably be expected to subject the Company, any ERISA Affiliate or any Health Plan to any material liability for penalties or excise Taxes under Code Sections 4980D or 4980H or any other provision of the PPACA.
- (i) **Effect of Transaction**. Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby (either alone or upon the occurrence of any additional or subsequent events) shall (i) result in any payment or benefit (including severance, bonus or otherwise) becoming granted or due to any Employee of the Company, (ii) result in any forgiveness of Indebtedness with respect to any Employee, (iii) increase any benefits otherwise payable by the Company, or (iv) result in the

obligation to fund benefits or result in the acceleration of the time of payment or vesting of any such benefits, except as required under Section 411(d)(3) of the Code.

- (j) **COVID-19 Measures**. The Company is, and at all times has been, in compliance with all mandatory quarantine, isolation, "shelter in place", "stay at home", social distancing, shut down, workplace safety, closure, state or county health officer order, published occupational health standard, and any other Law in connection with or in response to COVID-19 ("COVID-19 Measures") applicable to any location in which the Company operates, except where the failure to be in compliance has not been and would not reasonably be expected to have a material and adverse effect on the Company, taken as a whole. The Company has not received any written or, or to the Knowledge of the Company, oral complaint from any Employee alleging that the Company is not in material compliance with workplace COVID-19 Measures or failed to provide a safe working environment, appropriate equipment or accommodation in relation to COVID-19.
- Employment Matters. The Company is, and at all times has been, in material compliance with all applicable Laws regarding labor and employment, including all such Laws regarding collective bargaining agreements and arrangements and collective bargaining obligations, employment practices, terms and conditions of employment, worker classification (including the proper classification of workers as independent contractors and consultants and employees as exempt or non-exempt), prohibited discrimination, harassment, and retaliation, equal employment opportunity, fair employment practices, leaves of absence, meal and rest periods, vacation, sick and other paid time off, immigration status, recordkeeping, employee notices, employee safety and health, wages and hours (including overtime wages), compensation and hours of work, pay equity, background investigations, and in each case, with respect to Employees: (i) is not liable for any arrears of wages, severance pay or any Taxes or any penalty for failure to comply with any of the foregoing, and (ii) is not liable for any payment to any trust or other fund governed by or maintained by or on behalf of any Governmental Entity, with respect to unemployment compensation benefits, social security, health and welfare, or other benefits or obligations for Employees (other than routine payments to be made in the normal course of business and consistent with past practice). The Company has paid, and as of the Closing will have paid, in full to all Employees all wages, salaries, commissions, bonuses, benefits, and other compensation that are due and owing to such Persons. There are no Legal Proceedings, administrative matters, complaints, grievances, or investigations pending, or to the Knowledge of the Company, threatened against the Company or any of the Employees relating to any Employee, Employee Agreement or Company Employee Plan or other matter relating to labor, employment or benefits, including (without limitation), any claim relating to unfair labor practices, employment discrimination, harassment, retaliation, compensation, misclassification of workers, failure to pay wages or provide required breaks, or any other labor-, employment- or benefits-related matter arising under applicable Law. There are no pending or, to the Knowledge of the Company, threatened actions against the Company or any trustee under any worker's compensation policy or long term disability policy. The Company is not a party to any conciliation agreement, consent decree or other agreement or Order with any Governmental Entity with respect to labor or employment practices. The employment or engagement of each current Employee is terminable at the will of the Company, and the Company has no obligation to provide any particular form or period of notice prior to terminating the employment or engagement of any current Employee, and any such termination would result in no Liability to the Company. No former Employee (or spouse or other dependent of any former Employee) is receiving, scheduled to receive, owed, entitled to, or eligible for any benefits (whether from the Company or otherwise) relating to such former Employee's service with the Company or termination thereof.
- (l) **Labor**. There is no labor union, trade union, works council, employees' association, collective bargaining representative, affiliated bargaining agent or similar organization or representative (each a "*Labor Organization*") representing, purporting to represent or, to the Knowledge of the Company, seeking to represent any current Employees of the Company. The Company has not experienced any labor dispute, strike, slowdown, work stoppage, boycott, picketing, lockout, job action, or threat of any of the foregoing, or union organizing activity or question concerning representation, by or with respect to any of its Employees. To the Knowledge of the Company, there is not currently pending or threatened against the Company any: (i) labor dispute, strike, slowdown, work stoppage, boycott, picketing, lockout, job action or labor dispute; (ii) unfair labor practice charge or complaint before the National Labor Relations Board or any other Governmental Entity; (iii) request, application or demand by any Labor Organization for recognition as the collective bargaining representative of any Employees or with respect to union organizing activity; (iv) representation petition concerning representation regarding any Employees; or (v) Legal Proceeding, or labor disputes or grievances relating to any labor matters involving any Employee, including charges of unfair labor practices. The Company is not presently, nor has been in the past, a party to, or bound by, or obligated to negotiate, any collective bargaining agreement or other Contract with a Labor Organization representing any Employees

(collectively, "Labor Agreements"), nor is any such Labor Agreement presently being negotiated, nor is there any duty on the part of the Company to bargain with any Labor Organization or representative for or with respect to any Labor Agreement. The Company has not received any notices from the National Labor Relations Board, any Governmental Entity or any state labor relations agency or any Labor Organization alleging unfair labor practices or other non-compliance with labor Law, and there are no arbitration opinions interpreting and enforcing any Labor Agreement to which the Company is a party, or by which the Company is bound. To the Knowledge of the Company, no person has applied to have the Company declared a common or related employer pursuant to applicable Law. The Company has not taken any action that constituted a "plant closing" or "mass layoff" within the meaning of the WARN Act, issued any notification of a plant closing or mass layoff required by the WARN Act, or incurred any liability or obligation under the WARN Act that remains unsatisfied, including as a result of the COVID-19 pandemic or any Law, Order, directive, guidelines or recommendations by any Governmental Entity. No terminations prior to the Closing would trigger any notice requirement or other obligations under the WARN Act. No reduction in salary or wages or employee layoff, facility closure or shutdown, reduction-in-force, furlough, temporary layoff, or material work schedule change or reduction in hours that could reasonably be expected to trigger any notice requirement under the WARN Act, has occurred within the past two (2) years or is currently contemplated, planned or announced, including as a result of the COVID-19 pandemic or any Law, Order, directive, guidelines or recommendations by any Governmental Entity.

- (m) **No Interference or Conflict.** To the Knowledge of the Company, no Employee is obligated under any Contract, or subject to any Order of any Governmental Entity, that would conflict with or that conflicted with such Person's employment or engagement with the Company, interfere with such Person's efforts to promote the interests of the Company, or interfere with the business of the Company. Neither the execution nor delivery of this Agreement, nor the carrying on of the business of the Company as presently conducted or proposed to be conducted, nor any activity of such Employees in connection with the carrying on of the business of the Company as presently conducted or proposed to be conducted, shall, to the Knowledge of the Company, conflict with or result in a breach of the terms, conditions, or provisions of, or constitute a default under, any contract or agreement under which any of such Employees is now bound.
- (n) **Certain Employee Matters**. Section 3.19(n) of the Disclosure Schedule contains a correct and complete list of the current employees of the Company as of immediately prior to the Closing and shows with respect to each such employee: (i) the employee's name (or employee identification number if the employee's name must be redacted as required by applicable Laws), position held, department, base salary or hourly wage rate, as applicable, including each employee's status as either exempt or non-exempt, any bonus, commission, and all other compensation and profit sharing entitlements, (ii) date of hire, (iii) vacation or paid time off eligibility for the current calendar year (including accrued paid time off or sick leave from prior years), (iv) leave status (if applicable), (v) details of any visa or work permit, (vi) employing entity and location of employment, and (vii) any prior notice period or severance or termination payment (in cash or otherwise) to which he or she would be entitled in connection with termination of employment. No current employee has given notice to the Company to terminate his or her employment with the Company, and to the Knowledge of the Company, no employee listed on Section 3.19(n) of the Disclosure Schedule intends to terminate his or her employment for any reason.
- (o) **Contractor List**. Section 3.19(o) of the Disclosure Schedule lists, as of immediately prior to the Closing, (i) all current contractors, consultants and advisors to the Company; (ii) the location at which such contractors, consultants and advisors are providing services; (iii) the dates of engagement and the notice or termination provisions applicable to each individual (if any); (iv) the current terms of each individual's compensation; and (v) a description of each individual's services. Except as set forth on Section 3.19(o) of the Disclosure Schedule, all contractors, consultants and advisors to the Company can be terminated immediately and without advance notice or Liability on the part of the Company.
- (p) **Misconduct**. Except as disclosed on Section 3.19(p) of the Disclosure Schedule, during the three year period prior to the date hereof, no allegations of sexual harassment, sexual misconduct or unlawful discrimination or harassment while employed by, or providing services to, the Company have been made, or to the Company's Knowledge, threatened, (i) against any Employee, or (ii) by any Employee. The Company has not entered into any settlement agreement or conducted any investigation of allegations of sexual harassment, sexual misconduct or unlawful discrimination or harassment by or regarding any Employee or other representative of the Company. To the extent required by applicable Law, the Company has established and distributed to its Employees a policy against harassment, discrimination and retaliation and a complaint

procedure, and it has required all Employees to undergo anti-harassment, anti-discrimination and anti-retaliation training.

- (q) **Work Authorization**. All of the employees of the Company are authorized and have appropriate documentation to work in the jurisdiction in which they are working. Further, to the extent verification of eligibility to work is required by applicable Law, the Company has confirmed that all contractors or consultants engaged by the Company are authorized and have appropriate documentation to work in the jurisdiction in which they are working.
- **3.20. Insurance.** Section 3.20 of the Disclosure Schedule lists, as of immediately prior to the Closing, all insurance policies and fidelity bonds covering the assets, business, equipment, properties, operations, employees, officers and directors (in such capacities as employees, officers and directors) of the Company, including the type of coverage, the carrier, the policy limits of coverage, the term and the annual premiums of such policies. There are and have been no claims in the last three (3) years for which an insurance carrier has denied or threatened to deny coverage. All premiums due and payable under all such policies and bonds have been paid (or if installment payments are due, shall be paid if incurred prior to the Closing Date), and the Company is otherwise in material compliance with the terms of such policies and bonds.
- **3.21. Compliance with Laws**. In the past five (5) years, except as would not have a material and adverse effect on the Company, individually or in the aggregate, the Company has complied with, and is not in violation of, any Law. In the past five (5) years, the Company has not received any written notice, nor to the Knowledge of the Company, oral communication from a Governmental Entity of suspected, potential or actual material violation with respect to any Law, or provided any notice to any Governmental Entity regarding any material violation by the Company of any Law or Order.
- **3.22. Export Control and Sanctions Laws**. The Company has conducted any export transactions in accordance in all material respects with applicable provisions of United States export and re-export controls and sanctions laws and regulations, including the Export Administration Act and Regulations, the Foreign Assets Control Regulations, the International Traffic in Arms Regulations, other controls administered by the United States Department of Commerce or the United States Department of State, the regulations administered by the Office of Foreign Assets Control of the U.S. Treasury Department ("OFAC"), and all other applicable import/export controls and sanctions laws and regulations in other countries in which the Company conducts business. The Company has not engaged in any transactions or dealings with, or exported any products, technology, or services to, (a) any country or territory that is subject to a U.S. Government embargo (currently, Cuba, Iran, North Korea, Syria, and the Crimea Region) (collectively, the "Embargoed Countries"); (b) any instrumentality, agent, entity, or individual that is located in, or acting on behalf of, or directly or indirectly owned or controlled by any governmental entity of, any Embargoed Country; (c) any individual or entity identified on any list of designated and prohibited parties maintained by the U.S. Government, the United Kingdom, or the European Union, including, but not limited to, the List of Specially Designated Nationals and Blocked Persons, the Foreign Sanctions Evaders List, or the Sectoral Sanctions Identifications List, which are maintained by OFAC, or the Entity List, Denied Persons List, or Unverified List, which are maintained by the Bureau of Industry and Security of the U.S. Commerce Department (collectively, the "Prohibited Party Lists"). Neither the Company nor any of its respective actual or beneficial owners appears on a Prohibited Party List. Without limiting the foregoing: (i) the Company has obtained all export and import licenses, license exceptions and other consents, notices, waivers, approvals, Orders, authorizations, registrations, declarations and filings with any Governmental Entity required for (ii) the export, import and re-export of products, services, software and technologies, and (iii) releases of technologies and software to foreign nationals located in the United States and abroad (collectively, "Export Approvals"), (A) the Company is in compliance in all material respects with the terms of all applicable Export Approvals, (B) there are no pending or, to the Knowledge of the Company, threatened claims against the Company with respect to such Export Approvals, (C) to the Knowledge of the Company, there are no actions, conditions or circumstances pertaining to the Company's export transactions that would reasonably be expected to give rise to any future claims, and (D) no Export Approvals for the transfer of export licenses to Parent or the Final Surviving Company are required, except for such Export Approvals that can be obtained expeditiously and without material cost. Neither the Company nor any of its Employees (in their capacities as such) are or have been the subject of any allegation, voluntary disclosure, investigation, prosecution or other enforcement action related to violations of export control and sanctions laws and regulations.
- **3.23. Anti-Corruption Laws**. Neither the Company nor any Affiliate, Employee or agent of the Company (in their capacities as such or relating to their employment, services or relationship with the

Company), has (a) directly or indirectly made, offered, promised, authorized, solicited, or accepted any contribution, gift, bribe, rebate, payoff, influence payment, kickback, or other payment to or from any Person, including a "foreign official" (as such term is defined in the U.S. Foreign Corrupt Practices Act of 1977, as amended), foreign political party or official thereof, or candidate for foreign political office, regardless of what form, whether in money, property, or services (i) to obtain favorable treatment for business or Contracts secured (ii) to pay for favorable treatment for business or Contracts secured, (iii) to obtain special concessions or for special concessions already obtained; (iv) to improperly influence or induce any act or decision, (v) to secure any improper advantage, or (vi) in violation of applicable Law (including Anti-Corruption Laws), or (b) established or maintained any fund or asset that has not been recorded in the books and records of the Company. The Company has established and maintains internal controls (including, but not limited to, accounting systems, purchasing systems and billing systems) and written policies and procedures reasonably designed to promote compliance with Anti-Corruption Laws and to ensure that the books and records of the Company accurately and fairly reflect, in reasonable detail, all transactions and dispositions of funds and assets. Neither the Company nor any Employees (in their capacity as such) are or have been the subject of any voluntary disclosure, investigation, prosecution, or other enforcement action related to the Anti-Corruption Laws.

- **3.24. Due Diligence**. No written financial information or certificate furnished by or on behalf of the Company Parties to the Parent Parties in connection with this Agreement or the Related Agreements or the transactions contemplated hereby or thereby (in each case as modified or supplemented by other information so furnished) when taken as a whole contains any material misstatement of fact or omits to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not materially misleading; *provided that*, with respect to projected financial information, the Company Parties represent only that such information was prepared in good faith based upon assumptions believed to be reasonable at the time.
- **3.25. No Other Representations or Warranties.** Except for the representations and warranties contained in <u>Article III</u>, any Related Agreement or any certificate delivered hereunder, no Company Party nor any other Person makes any other representation or warranty, express or implied, at law or in equity, on behalf of the Company Parties, or the business, operations or assets of the Company Parties. Except for the representations and warranties contained in <u>Article IV</u> or the representations or warranties expressly made by the Parent Parties in any Related Agreement, the Company Parties acknowledge and agree that no representation or warranty of any kind whatsoever, express or implied, is made or shall be deemed to have been made by or on behalf of the Parent Parties in connection with this Agreement, and each of the Company Parties and the Stockholder Representative (on behalf of the Company Stockholders and Company Warrantholder) hereby expressly disclaims reliance upon any such representation or warranty.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE PARENT PARTIES

Each of Parent, Parent Holdco LLC and Parent Merger Sub (collectively, the "*Parent Parties*") represents and warrants as of the Closing Date (except to the extent any such representation or warranty expressly relates to a different date (in which case as of such date)) to the Company as follows:

- **4.1. Organization**. Each of the Parent Parties is duly incorporated or formed, validly existing and in good standing under the laws of the State of Delaware. Each of the Parent Parties has the requisite corporate or other legal power and authority, as applicable, to own, lease and operate its assets and properties to carry on its business as it is now being conducted.
- **4.2. Authority and Enforceability**. Each of the Parent Parties has all requisite corporate or other legal power and authority, as applicable, to enter into this Agreement and any Related Agreements to which it is a party and to consummate the transactions contemplated hereby and thereby. The execution and delivery by each of the Parent Parties of this Agreement and any Related Agreements to which it is a party and the consummation of the transactions contemplated hereby and thereby have been duly authorized by all necessary corporate action, as applicable, on the part of each of the Parent Parties. This Agreement and any Related Agreements to which any of the Parent Parties is a party have been duly executed and delivered by the Parent Parties, as the case may be, and, assuming the due authorization, execution and delivery by the other parties hereto and thereto, constitute the valid and binding obligations of the Parent Parties, as the case may be, in accordance with their terms, subject to (a) Laws of general application relating to bankruptcy, insolvency, fraudulent conveyance, reorganization,

moratorium and other similar laws relating to or affecting creditors' rights generally and (b) general principles of equity.

4.3. Capitalization. Immediately prior to the Closing, the shares of Parent Common Stock to be issued to the Company Stockholders and Company Warrantholder in connection with the Parent Stock Issuance will be duly and validly reserved for issuance. Upon consummation of the Combination Transactions and the other transactions contemplated by this Agreement in accordance with the terms hereof, the shares of Parent Common Stock to be issued to the Company Stockholders and Company Warrantholder in connection with the Parent Stock Issuance shall be duly authorized, validly issued, fully paid and nonassessable, and will be free of any Liens, other than Liens created by the Company Stockholders and Liens imposed by the securities Laws, if any. The Parent Stock Issuance will not, at the time of issuance in accordance with the terms of this Agreement, violate any pre-emptive rights, rights of first offer, rights of first refusal or similar rights of any Person.

4.4. No Conflict.

- (a) The execution and delivery by each of the Parent Parties of this Agreement and any Related Agreement to which it is a party, and the consummation by each of the Parent Parties of the transactions contemplated hereby and thereby, shall not (i) result in the creation of any Lien (other than a Permitted Lien or as disclosed in the Parent SEC Documents) on any of the material assets of Parent or its subsidiaries or any of the Equity Interests of Parent or its subsidiaries, (ii) conflict with or result in any violation of or default under (with or without notice or lapse of time, or both) (iii) the Parent Organizational Documents or (iv) any Law or Order applicable to the Parent Parties, other than, in the case of this clause (a), such conflicts, violations or defaults as would not, individually or in the aggregate, reasonably be expected to (A) result in a material effect with respect to Parent or (B) prevent or materially delay the consummation of the Combination Transactions and the other transactions contemplated by this Agreement.
- (b) No consent, approval, Order or authorization of, or registration, declaration or filing with, or notice to, any Governmental Entity or any other Person is required by or with respect to the Parent or any of its Subsidiaries in connection with the execution and delivery of this Agreement or the consummation of the transactions contemplated hereby, except for (i) the filing of the Shareablee Merger Certificate, the Shareablee Conversion Certificate, the Initial Merger Certificate and the Final Merger Certificate, (ii) the filing of applicable notices and applications with NASDAQ, (iii) the filing of applicable reports on Form 8-K relating to this Agreement and the transactions contemplated hereby, (iv) the filing of any required registration statements, (v) the filing of any required beneficial ownership reports by the Company Stockholders, (vi) the notification of Bank of America, N.A., to the extent required by the Credit Agreement, dated May 5, 2021, between Parent, certain Subsidiaries of Parent, Bank of America, N.A., and the lenders party thereto, and (vii) such other consents, approvals, Orders, authorizations, registrations, declarations, filings and notices that, if not obtained or made, would not materially and adversely affect, and would not reasonably be expected to materially and adversely affect, Parent or any of its Subsidiaries' ability to perform or comply with the covenants, agreements or obligations of Parent herein or to consummate the transactions contemplated by this Agreement in accordance with this Agreement and applicable Law.
- (c) No vote or other action of the stockholders of Parent is required by Law, NASDAQ rules, the certificate of incorporation or bylaws (or similar charter or organizational documents) of Parent in order for the Parent Parties to enter into this Agreement or the Related Agreements or consummate the transactions contemplated hereby or thereby.
- **4.5. Litigation**. As of immediately prior to the Closing, there is no Legal Proceeding pending (or, to the Knowledge of the Parent Parties, being threatened) against the Parent Parties, that challenges, or that may have the effect of preventing, delaying, making illegal or otherwise interfering with, the Combination Transactions or any of the other transactions contemplated by this Agreement.
- **4.6. No Prior Activities of Parent Holdco LLC and Parent Merger Sub.** Except for obligations incurred in connection with its incorporation or organization or the negotiation and consummation of this Agreement and the transactions contemplated hereby, neither Parent Holdco LLC nor Parent Merger Sub has not incurred any obligation or liability nor engaged in any business or activity of any type or kind whatsoever or entered into any agreement or arrangement with any Person.
- **4.7. Financial Statements**. Except as disclosed in the Parent SEC Documents filed and publicly available prior to the Closing (excluding any disclosures in any risk factors section that do not constitute

statements of fact, disclosures in any forward-looking statements disclaimer and other disclosures that are generally cautionary, predictive or forward-looking in nature):

- (a) Parent has timely filed or furnished all forms, documents and reports required to be filed or furnished by it with the SEC since January 1, 2019 (all such documents and reports filed or furnished by Parent, the "Parent SEC Documents"). As of their respective dates or, if amended, as of the date of the last such amendment (and, in the case of registration statements, on the dates of effectiveness), the Parent SEC Documents complied in all material respects with the applicable requirements of the Securities Act, the Exchange Act and the Sarbanes-Oxley Act of 2002, as amended, as the case may be, and the applicable rules and regulations promulgated thereunder, and none of the Parent SEC Documents when filed contained any untrue statement of a material fact or omitted to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. To the Knowledge of Parent, none of the Parent SEC Documents is the subject of ongoing SEC review or outstanding SEC comment. To the extent that complete and correct copies are not available on the SEC's website, Parent has made available copies of all comment letters received from the SEC since January 1, 2019 and relating to the Parent SEC Documents, together with all written responses of Parent thereto. There are no internal investigations, SEC inquiries or investigations or other governmental inquiries or investigations pending or, to the Knowledge of Parent, threatened, in each case regarding any accounting practices of Parent.
- (b) The financial statements (including the related notes) of the Parent included in the Parent SEC Documents complied, at the time the respective statements were filed, as to form in all material respects with the applicable accounting requirements and the published rules and regulations of the SEC with respect thereto, were prepared in accordance with GAAP (except, in the case of unaudited quarterly financial statements, as permitted by Form 10-Q and related rules and regulations of the SEC) applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto) and fairly present in all material respects the consolidated financial position of the Parent and its consolidated Subsidiaries as of the dates thereof and their consolidated results of operations and cash flows for the periods then ended (subject, in the case of unaudited quarterly financial statements, to normal year-end audit adjustments, the absence of complete footnotes and to any other adjustments described therein, including any notes thereto, or with respect to pro-forma financial information, subject to the qualifications stated therein).
- (c) Parent has established and maintains disclosure controls and procedures and internal controls over financial reporting (as such terms are defined in Rules 13a-15 and 15d-15 under the Exchange Act). Such disclosure controls and procedures are effective to provide reasonable assurance that information required to be disclosed by Parent in the reports that Parent files under the Exchange Act is recorded and reported on a timely basis to the individuals responsible for the preparation of Parent's filings with the SEC and other public disclosure documents. Since January 1, 2019, neither Parent nor, to Parent's Knowledge, Parent's independent accountant, has identified or been made aware of any material weaknesses in the design or operation of internal control over financial reporting that would be reasonably likely to adversely affect in any material way Parent's ability to record, process, summarize and report financial information. Parent maintains "disclosure controls and procedures" (as defined in the Rule 13a-15(e) of the Exchange Act) in compliance with the Exchange Act.
- (d) Parent is not a party to, and has not entered into any Contract to become a party to, any joint venture, off-balance sheet partnership or any similar Contract, including any structured finance, special purpose or limited purpose entity or Person, or any "off-balance sheet arrangements" (as defined in Item 303(a) of Regulation S-K of the SEC), where the result, purpose or intended effect of such Contract is to avoid disclosure of any material transaction involving, or material liabilities of, Parent in Parent's audited financial statements or other Parent SEC Documents.
- (e) Since January 1, 2019, there have not been any disagreements between Parent and the independent accountants engaged as the principal accountants to audit Parent's consolidated financial statements, or an independent accountant who was previously engaged to audit a significant subsidiary of Parent and on whom the principal accountants expressed reliance in their report, on any matter of accounting principles or practices, financial statement disclosure, or auditing scope or procedure, which disagreement(s), if required to be disclosed in the Parent SEC Documents pursuant to the published rules and regulations of the SEC applicable thereto, were not so disclosed in a timely manner.
- **4.8. NASDAQ Compliance**. (a) Parent is in compliance with the applicable listing and corporate governance rules and regulations of NASDAQ and (b) Parent has caused all shares of Parent Common Stock to

be issued pursuant to this Agreement including, without limitation, shares of Parent Common Stock issuable upon exercise of Converted Options and settlement of Converted RSU Awards, to be approved for listing on the NASDAQ.

- **4.9. Brokers' and Finders' Fees.** Parent has not incurred, and shall not incur, directly or indirectly, any Liability for brokerage or finders' fees or agents' commissions, fees related to investment banking or similar advisory services or any similar charges in connection with this Agreement or any transaction contemplated hereby.
- **4.10. Form S-8 Registration Statement**. Parent is, and at all relevant times on and after the Closing Date shall be, eligible to register all shares of Parent Common Stock issuable upon the exercise of the Converted Options and the settlement of the Converted RSU Awards pursuant to Form S-8.
- **4.11. Due Diligence**. No written financial information or certificate furnished by or on behalf of the Parent Parties to the Company Parties in connection with this Agreement or the Related Agreements or the transactions contemplated hereby or thereby (in each case as modified or supplemented by other information so furnished) when taken as a whole contains any material misstatement of fact or omits to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not materially misleading; *provided that*, with respect to projected financial information, the Parent Parties represent only that such information was prepared in good faith based upon assumptions believed to be reasonable at the time.
- **4.12. No Other Representations or Warranties.** Except for the representations and warranties contained in <u>Article IV</u> or any Related Agreement, no Parent Party nor any other Person makes any other representation or warranty, express or implied, at law or in equity, on behalf of the Parent Parties, or the business, operations or assets of the Parent Parties. Except for the representations and warranties contained in <u>Article III</u>, any Related Agreement or any certificate delivered hereunder, the Parent Parties acknowledge and agree that no representation or warranty of any kind whatsoever, express or implied, is made or shall be deemed to have been made by or on behalf of the Company Parties in connection with this Agreement, and each of the Parent Parties hereby expressly disclaims reliance upon any such representation or warranty.

ARTICLE V STOCKHOLDER AND TAX MATTERS

5.1. Company Board Recommendation. Immediately after the execution and delivery of this Agreement, the Company shall solicit, obtain and deliver to Parent the Company Written Consent.

5.2. Tax Matters.

- (a) **Preparation of Tax Returns**. Parent shall prepare or cause to be prepared and file or cause to be filed, at the expense of the Shareablee Holdco Stockholders and the Company Stockholders, all Tax Returns for Shareablee Holdco or the Company for a Pre-Closing Tax Period or Straddle Period that are required to be filed after the Closing Date (the "**Parent Prepared Returns**"). In the event that any Parent Prepared Return that is an income Tax Return or other material Tax Return that shows any Unpaid Pre-Closing Taxes that are subject to indemnification by the Company Indemnitors pursuant to this Agreement, Parent shall submit such Parent Prepared Return to the Stockholder Representative for review and comment at least twenty (20) days (or, if such Parent Prepared Return is required to be filed within twenty (20) days after the Closing Date, as soon as practicable after the Closing Date) prior to the filing of such Parent Prepared Return (taking into account any validly obtained extensions of time to file); provided that any failure or delay in providing any Parent Prepared Return to the Stockholder Representative shall not relieve the Company Indemnitors of any indemnification obligations with respect to such Tax Return except to the extent the Company Indemnitors are actually prejudiced as a result thereof. Parent shall consider in good faith all reasonable comments received from the Stockholder Representative within fifteen (15) days after Parent has provided the applicable Parent Prepared Return.
- (b) **Cooperation**. The Stockholder Representative shall cooperate fully, as and to the extent reasonably requested by Parent, in connection with the filing of any Tax Returns with respect to Shareablee Holdco's or the Company's operations and any pending or threatened audits, Legal Proceedings or assessments with respect thereto or to Taxes owed by Shareablee Holdco or the Company. Such cooperation shall include the retention and (upon the relevant Party's request) the provision of records and information in such Party's

possession which are reasonably relevant to any such matters and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Notwithstanding the foregoing or any other provision herein to the contrary, in no event shall the Stockholder Representative or any Company Indemnitor be entitled to review or otherwise have access to any Tax Return, or information related thereto, of Parent or any of its Affiliates (other than Tax Returns, and information related thereto, of Shareablee Holdco or the Company for Pre-Closing Tax Periods).

- (c) **Unpaid Pre-Closing Taxes**. Each Company Indemnitor shall bear and, promptly upon request by Parent, pay (or cause to be paid) to Parent (or at Parent's election, the appropriate Governmental Entity) the amount of such Company Indemnitor's Pro Rata Share of any Unpaid Pre-Closing Taxes to the extent not taken into account in the Final Closing Statement; *provided*, *however*, that in the case of any withholding Taxes described in clause (d) of the definition of "Unpaid Pre-Closing Taxes" and withheld in respect of a Company Indemnitor, such Company Indemnitor shall bear and, promptly upon request by Parent, pay (or cause to be paid) to Parent (or at Parent's election, the appropriate Governmental Entity) the amount of such Taxes. The Person(s) required to do so under applicable law shall, at its own expense, file (or cause to be filed) all Tax Returns and other documentation with respect to Transfer Taxes that are required to be filed by applicable Law. If required by applicable Law, the Parties shall, and shall cause their respective Affiliates to, join in the execution of any such Tax Returns and other documentation.
- (d) Tax Treatment. The Combination Transactions are intended to qualify for the Intended Tax Treatment and, subject to Section 356 of the Code, the transactions described in Section 2.1(a), Section 2.1(d), Section 2.2(a), Section 2.2(b), Section 2.3(a)(ii) and Section 2.3(b)(i) are intended to qualify as tax-free exchanges pursuant to Section 354(a)(1) of the Code. The Parties shall report all such transactions in accordance therewith on their respective U.S. federal income Tax Returns for the taxable year including the Closing Date and shall file all applicable U.S. state and local income Tax Returns in a manner consistent therewith, unless otherwise required by a Taxing Authority following an audit or other examination in which the Intended Tax Treatment was defended reasonably and in good faith. Notwithstanding the foregoing, Parent makes no representations or warranties to Shareablee Holdco, the Company, any Shareablee Holdco Stockholder, any Company Stockholder, the Company Warrantholder or the Shareablee Holdco Warrantholder and such Persons make no representations to Parent Holdco LLC regarding the Tax treatment or any of the Tax consequences of the transactions contemplated by this Agreement. Each Party and the Shareablee Holdco Stockholders, the Company Stockholders, the Company Warrantholder and the Shareablee Holdco Warrantholder acknowledge that they are relying solely on their own Tax advisors in connection with the transactions contemplated by this Agreement.
- (e) **Tax Audits**. Parent shall, within fifteen (15) Business Days of the receipt of written notice from a Governmental Entity regarding the commencement of any audit, examination or other administrative or judicial proceeding with respect to any Taxes of Shareablee Holdco or the Company for any Pre-Closing Tax Period (each a "*Tax Contest*"), provide the Stockholder Representative with written notice thereof; *provided* that, the failure or delay in providing such notice to the Stockholder Representative will not affect the obligations of the Company Indemnitors hereunder unless and to the extent the Company Indemnitors are actually prejudiced thereby. Parent shall keep the Stockholder Representative reasonably informed concerning the progress of any Tax Contest and, except with the consent of the Stockholder Representative (such consent not to be unreasonably withheld, conditioned or delayed), no settlement of any such Tax Contest shall be determinative of the amount of Losses relating to such matter. To the extent this <u>Section 5.2(e)</u> conflicts with <u>Section 7.5</u>, this <u>Section 5.2(e)</u> shall control.

5.3. Indemnification of Directors and Officers of the Company; Insurance.

(a) During the period ending six (6) years after the Closing Date, the Final Surviving Company or its respective successors shall, or Parent shall cause the Final Surviving Company or their respective successors to, fulfill their obligations to the present and former members of the Company Board, present and former managers and present and former officers of the Company (such directors, managers and officers being herein called the "Company Indemnitees") pursuant to the terms of the Charter Documents after giving effect to the transactions contemplated by Article I (such obligations, collectively, the "D&O Indemnifiable Matters"). Notwithstanding the foregoing, the obligations of Parent and the Final Surviving Company or their respective successors in respect of the D&O Indemnifiable Matters (i) shall be subject to any limitation imposed by applicable Law, the terms of the Charter Documents or the terms of the applicable indemnification agreement, and (ii) shall not be deemed to release any Company Indemnitee who is also an

officer, manager or director of the Company from his or her obligations pursuant to this Agreement or any Related Agreement.

- (b) Prior to the Closing Date, the Company shall purchase and fully pay (and such purchase price shall be included in the Closing Transaction Expense Amount) for an extended reporting period endorsement under the Company's existing directors' and officers' liability insurance coverage in a form reasonably acceptable to Parent that shall provide the Company Indemnitees with coverage for six (6) years following the Closing Date of not less than the existing coverage and have other terms not materially less favorable to the insured persons than the Company's directors' and officers' liability insurance coverage presently maintained by the Company (the "**D&O Tail Policy**"). Parent shall not, and shall cause the Final Surviving Company to not, take any action to eliminate such D&O Tail Policy. The cost of the D&O Tail Policy shall be considered a Transaction Expense for purposes of this Agreement. At or prior to the Closing, the Company shall provide a copy of the D&O Tail Policy to Parent, along with written confirmation from the insurance provider that such D&O Tail Policy will be bound at Closing.
- (c) If Parent, the Final Surviving Company, or any of their successors or assigns (i) consolidates with or merges into any other Person and shall not be the continuing or surviving company or entity of such consolidation or merger, (ii) transfers or conveys all or substantially all of their assets to any Person, or (iii) ceases to exist for any reason, then, in each such case, to the extent necessary, Parent shall use its commercially reasonable efforts to cause the proper provision to be made so that the successors and assigns of Parent or the Final Surviving Company, as the case may be, shall assume the obligations set forth in this Section 5.3.
- (d) Notwithstanding anything in this Agreement to the contrary, the obligations under this <u>Section 5.3</u> shall not be terminated or modified in such a manner as to adversely affect any Company Indemnitee to whom this <u>Section 5.3</u> applies without the consent of such affected Company Indemnitee, it being understood and agreed that the Company Indemnitees are intended to be express third party beneficiaries of this Agreement.

5.4. Securities Law Exemption; Transfer Restrictions; Stop-Transfer Instructions, Securities Law Compliance.

- (a) Each of the Parties shall use commercially reasonable efforts to cause the issuance of all shares of Parent Common Stock contemplated by this Agreement in connection with the Combination Transactions (excluding, for the avoidance of doubt, shares of Parent Common Stock issuable upon the exercise of the Converted Options and the settlement of the Converted RSU Awards) to validly qualify for an exemption from the registration and prospectus delivery requirements of the Securities Act and any applicable state "blue sky" securities laws.
- (b) The shares of Parent Common Stock issued to the Shareablee Holdco Stockholders and Shareablee Holdco Warrantholder hereunder constitute "restricted securities" under the Securities Act, and may not be transferred absent valid and effective registration under the Securities Act or an exemption therefrom, and any such transfer shall be subject to compliance with applicable state securities Laws. All shares of Parent Common Stock issued to the Shareablee Holdco Stockholders and Shareablee Holdco Warrantholder hereunder shall bear a legend or legends referencing restrictions applicable to such shares under applicable securities Laws and under this Agreement (the "Restrictive Legends"), which Restrictive Legend shall state in substance:

"The securities evidenced by this certificate have been issued and sold without registration under the United States Securities Act of 1933, as amended (the "Securities Act"), or the securities laws of any state of the United States (a "State Act") in reliance upon certain exemptions from registration under said acts. The securities evidenced by this certificate cannot be sold, assigned or otherwise transferred within the United States unless such sale, assignment or other transfer is (i) made pursuant to an effective registration statement under the Securities Act and in accordance with each applicable State Act or (ii) exempt from, or not subject to, the Securities Act and each applicable State Act."

(c) Following the Closing until the date the Restrictive Legends are removed from all shares of Parent Common Stock issued in connection with this Agreement, Parent shall use commercially reasonable efforts to comply with all applicable provisions of Rule 144 under the Securities Act, including without

limitation SEC reporting requirements and information requirements in Rule 144(c) under the Securities Act, to enable each Shareablee Holdco Stockholder and the Shareablee Holdco Warrantholder to transfer the shares of Parent Common Stock they receive from time to time pursuant to (and to the extent permitted by) the exemptions provided by Rule 144 under the Securities Act. So long as a Shareablee Holdco Stockholder or the Shareablee Holdco Warrantholder owns any Parent Common Stock issued pursuant to this Agreement that is characterized as "restricted securities" under federal securities Laws, Parent shall promptly furnish upon the written request of the Stockholder Representative, to the extent accurate, a written statement of Parent that it has complied with the reporting requirements of Rule 144 under the Securities Act and all other documents reasonably required to transfer the Parent Common Stock without restriction of any kind.

Except as expressly provided in this Section 5.4(d), or in accordance with a written consent granted by Parent, each of the Shareablee Holdco Stockholders and Shareablee Holdco Warrantholder shall not transfer, directly or indirectly, any of the Parent Common Stock (other than shares issued with respect to the Deferred Amount) owned by he, she or it, other than, (i) with respect to each Shareablee Holdco Stockholder except for Tania Yuki Mustapic, (A) thirty five percent (35%) of the aggregate Parent Common Stock acquired by him, her or it pursuant to this Agreement following the six (6) month anniversary of the Closing Date, (B) an additional fifty percent (50%) of the aggregate Parent Common Stock acquired by him, her or it pursuant to this Agreement following the twelve (12) month anniversary of the Closing Date, and (C) the remaining fifteen percent (15%) of the aggregate Parent Common Stock acquired by him, her or it pursuant to this Agreement following the eighteen (18) month anniversary of the Closing Date, (ii) with respect to Tania Yuki Mustapic, (A) twenty five percent (25%) of the aggregate Parent Common Stock acquired by her pursuant to this Agreement following the six (6) month anniversary of the Closing Date, (B) an additional twenty five percent (25%) of the aggregate Parent Common Stock acquired by her pursuant to this Agreement following the one (1) year anniversary of the Closing Date, (C) an additional twenty five percent (25%) of the aggregate Parent Common Stock acquired by her pursuant to this Agreement following the two (2) year anniversary of the Closing Date, and (D) the remaining twenty five percent (25%) of the aggregate Parent Common Stock acquired by her pursuant to this Agreement following the (3) year anniversary of the Closing Date, and (iii) with respect to the Shareablee Holdco Warrantholder, one hundred percent (100%) of the aggregate Parent Common Stock acquired by him, her or it pursuant to this Agreement following the six (6) month anniversary of the Closing Date (each, as applicable, the "Transfer Prohibition Period"). Except as expressly provided in this Section 5.4(d), or in accordance with a written consent granted by Parent, each of the Deferred Amount Holders shall not transfer, directly or indirectly, any of the Parent Common Stock that he, she or it received as part of the Deferred Amount until the six (6) month anniversary of receipt of any such shares (as applicable, the "Deferred **Amount Prohibition Period**"). Notwithstanding the foregoing, the restrictions and limitations on transfers contained in this Section 5.4(d) shall not apply to any direct or indirect transfer of Parent Common Stock (x) by any Shareablee Holdco Stockholder or the Shareablee Holdco Warrantholder who is a natural person for bona fide estate planning purposes to a trust, limited liability company or other estate planning tool formed under the laws of the United States or any political subdivision thereof solely for the benefit of such Shareablee Holdco Stockholder or Shareablee Holdco Warrantholder and such Shareablee Holdco Stockholder's or Shareablee Holdco Warrantholder's spouse, parents, siblings and descendants (whether by birth or adoption) (a "Family Group") (or a re-transfer of such Parent Common Stock by such trust, limited liability company or other estate planning tool back to such Shareablee Holdco Stockholder or Shareablee Holdco Warrantholder upon the revocation of any such trust) or pursuant to the applicable Laws of descent or distribution among such Shareablee Holdco Stockholder's or Shareablee Holdco Warrantholder's Family Group, (y) by any Shareablee Holdco Stockholder or the Shareablee Holdco Warrantholder to its Affiliates (excluding, for the avoidance of doubt, any portfolio companies and limited partners of such Shareablee Holdco Stockholder or Shareablee Holdco Warrantholder), but only if such transfer to an Affiliate is part of a corporate reorganization or restructuring for nominal consideration or for no consideration, and (z) by any Shareablee Holdco Stockholder or the Shareablee Holdco Warrantholder to cover applicable Taxes with respect to his, her or its ownership of such Parent Common Stock pursuant to the transactions contemplated hereby; provided, further, that (I) no such transfer shall be permitted if such transfer will result in an "owner shift" under Section 382(g) of the Code with respect to Parent and (II) other than with respect to a transfer contemplated by clause (z), the transferee agrees in writing with Parent to be bound by this <u>Section 5.4(d)</u> to the same extent as its transferor. Parent agrees that it shall not waive compliance with this Section 5.4(d) with respect to any Shareablee Holdco Stockholder unless it makes a similar waiver with respect to all other Shareablee Holdco Stockholders. For the avoidance of doubt, as of the date hereof, other than this Section 5.4, there are no other contractual transfer restrictions between Parent and the Shareablee Holdco Stockholders and Shareablee Holdco Warrantholder with respect to Parent Common Stock.

(e) To ensure compliance with the restrictions and limitations on transfers imposed by this <u>Section 5.4</u>, Parent may issue appropriate "stop-transfer" instructions to its transfer agent and include a restrictive legend on any shares of Parent Common Stock that are subject to this <u>Section 5.4</u>. Parent shall not be required (x) to transfer on its books any shares of Parent Common Stock that have been sold or otherwise transferred in violation of any of the provisions of this <u>Section 5.4</u> or the requirements of the Securities Act or any applicable state act, or (y) to treat as owner of such shares of Parent Common Stock, or to accord the right to vote or pay dividends, to any purchaser or other transferee to whom such shares of Parent Common Stock have been purportedly so transferred. Upon expiration of the applicable Transfer Prohibition Period or applicable Deferred Amount Prohibition Period, or earlier with respect to the end of the respective transfer restriction periods of the Shareablee Holdco Stockholders and Shareablee Holdco Warrantholder as set forth in <u>Section 5.4(d)</u> above, Parent shall upon request, cause its transfer agent to remove all restrictive legends effected under this <u>Section 5.4</u> from the Parent Common Stock issuable pursuant to this Agreement, and Parent shall not thereafter issue any stop-transfer instructions to its transfer agent with respect to the restrictions imposed by this <u>Section 5.4</u>.

ARTICLE VI ADDITIONAL AGREEMENTS

- **6.1. Conduct of the Business of the Company.** During the period from the execution and delivery of this Agreement and continuing until the earlier of the termination of this Agreement and the Closing, the Company shall, except to the extent expressly required by applicable Law or as consented to by Parent in writing, conduct its business in the Ordinary Course of Business and use its commercially reasonable efforts to (a) preserve intact its present business organization and keep available the services of its Employees, (b) maintain in effect all Company Authorizations, (c) maintain its assets and properties, (d) maintain its relationships with its customers, lenders, data providers, suppliers, lessors and others having material business relationships with it, and (e) manage its working capital (including the timing of collection of accounts receivable and of the payment of accounts payable) and pay and perform all of its undisputed debts and other obligations (including Taxes) when due, and, in each case of the foregoing, in the Ordinary Course of Business consistent with past practice.
- **6.2. Public Disclosure**. So long as this Agreement is in effect, neither the Company nor its Representatives (including the Company Stockholders) shall issue or cause the publication of any press release or other public announcement or disclosure with respect to this Agreement, the Related Agreements or the transactions contemplated hereby or thereby, including the Combination Transactions, without the prior written consent of Parent; *provided*, the Company and its Representatives may, without further consent of Parent, make statements with respect to this Agreement, the Related Agreements or the transactions contemplated hereby or thereby, including the Combination Transactions, that are consistent with (and do not otherwise include or refer to any terms or conditions that are not otherwise contained in) public statements with respect to this Agreement, the Related Agreements or the transactions contemplated hereby or thereby, including the Combination Transactions, that have previously been made by Parent. Parent shall provide the Company reasonable prior written notice of any press releases or other public announcements made prior to the Closing with respect to this Agreement, the Related Agreements or the transactions contemplated hereby or thereby.
- **6.3. Commercially Reasonable Efforts.** Each of the Parties agrees to use its commercially reasonable efforts, and to cooperate with each other Party, to take, or cause to be taken, all actions, and to do, or cause to be done, all things necessary, appropriate or desirable to consummate and make effective, in the most expeditious manner practicable, the Combination Transactions and the other transactions contemplated by this Agreement, and including to execute and deliver such other instruments and do and perform such other acts and things as may be necessary or reasonably desirable for effecting completely the consummation of the Combination Transactions and the other transactions contemplated hereby.
- **6.4. Exchange Documentation**. The Company shall use commercially reasonable efforts to cause the Exchange Documentation to be executed and delivered to the Company on or prior to the Closing by each of the Company Stockholders and the Company Warrantholder.
- **6.5. Confidentiality**. The Parties acknowledge that Parent and the Company have previously executed that certain Mutual Non-Disclosure Agreement dated March 23, 2021, by and between the Company and Parent (the "Confidentiality Agreement"), which shall terminate as of the Closing. After the Closing, each of the Company Stockholder, the Company Warrantholder and the Stockholder Representative (each, an "Obligated Party") shall maintain in confidence and shall cause its Affiliates to maintain in confidence, any

written, oral or other information relating to (i) Parent and its Affiliates (including Shareablee LLC), and (ii) the Company and its business whether before or after the Closing Date (collectively, "Confidential Information"), in each case, until the two (2) year anniversary of the Closing Date, except that the requirements of this Section 6.5 shall not apply to the extent that (i) any such information is or becomes generally available to the public other than as a result of disclosure in a manner not permitted hereby, (ii) any such information is required by applicable Law or a Governmental Entity to be disclosed (including any report, statement, testimony or other submission to such Governmental Entity) or (iii) any such information is reasonably necessary to be disclosed in connection with any Legal Proceeding or any dispute with respect to this Agreement (including any response to any summons, subpoena or other legal process or formal or informal investigative demand issued to the disclosing Party in the course of any Legal Proceeding). In the event that an Obligated Party is required by Law or any Legal Proceeding to disclose any of the Confidential Information, to the extent permitted by Law and to the extent reasonably practicable, the Obligated Party shall provide Parent with notice as promptly as practicable of any such requirement so that Parent may, at its sole cost and expense, seek a protective order or other appropriate remedy and/or waive compliance with the provisions of this Agreement. If, in the absence of a protective order or other remedy or the receipt of a waiver by Parent, such Obligated Party is nonetheless required to disclose any of the Confidential Information, such Obligated Party may disclose only that portion of the Confidential Information which is required to be disclosed; provided, however, that such Obligated Party shall use commercially reasonable efforts to cooperate with Parent, at Parent's sole cost and expense, in its efforts to obtain an appropriate protective order or other reliable assurance that confidential treatment will be accorded to the Confidential Information. The Obligated Parties shall not oppose any action by Parent to obtain an appropriate protective order or other reliable assurance that confidential treatment will be accorded the Confidential Information. Notwithstanding anything herein to the contrary, following Closing, the Stockholder Representative shall be permitted to disclose information to Representatives of the Stockholder Representative and to the Company Stockholders, in each case who have a need to know such information; *provided* that such persons are subject to confidentiality obligations with respect thereto.

6.6. Employee Benefits Covenant.

- (a) Subject to the terms of the Parent Plans, the provisions of the Parent Plans' administrators and applicable Law (i) Parent shall cause each Continuing Employee to receive credit for purposes of eligibility and vesting under employee benefits plans maintained by Parent and the Final Surviving Company (the "*Parent Plans*") in which such Continuing Employee participates immediately following Closing for years of service with the Company prior to the Closing Date, and (ii) Parent shall cause any and all pre-existing condition limitations, eligibility waiting periods and evidence of insurability requirements under any Parent Plans that are group health or welfare plans in which such Continuing Employee will participate to be waived and will, for the plan year in which the Closing Date occurs, provide credit for any co-payments and deductibles incurred prior to the Closing Date for purposes of satisfying any applicable deductible, out-of-pocket or similar requirements under any such plans that may apply after the Closing Date.
- (b) The provisions of this <u>Section 6.6</u> are for the sole benefit of the Parties hereto and nothing herein, express or implied, is intended or shall be construed to confer upon or give to any Person (including any Continuing Employee) other than the Parties to this Agreement and their respective successors and permitted assigns, any legal or equitable or other rights (including any third party beneficiary rights) or remedies under or by reason of any provision of this <u>Section 6.6</u>. Nothing contained herein, express or implied (i) shall be construed to establish, amend or modify any benefit plan, program, agreement or arrangement (including any benefit plan of Parent), (ii) shall alter or limit Parent's ability to amend, modify or terminate any benefit plan, program, agreement or arrangement (including any benefit plan of Parent) subject to compliance with this <u>Section 6.6</u>, or (iii) is intended to confer upon any Continuing Employee any right to employment or continued employment for any period of time by reason of this Agreement, or any right to a particular term or condition of employment.

ARTICLE VII SURVIVAL OF REPRESENTATIONS AND WARRANTIES; INDEMNIFICATION

7.1. Survival. The representations and warranties of the Company Parties contained in this Agreement and in the Certificates shall survive the Closing Date until 11:59 p.m. (New York time) on the date that is twelve (12) months following the Closing Date; *provided*, that in the event of a claim for Fraud, such representation or warranty shall survive indefinitely following the Closing Date; *provided*, *further*, that the representations and warranties contained in <u>Section 3.8</u> (Tax Matters) (the "*Tax Representations*") and the Fundamental Representations of the Company Parties shall remain in full force and effect until 11:59 p.m. (New

York time) on the date that is thirty (30) days following the expiration of the statute of limitations, including tolling, applicable to such representation or warranty; provided, further, that the Critical Representations shall remain in full force and effect until 11:59 p.m. (New York time) on the date that is twenty four (24) months following the Closing Date; provided further, if an indemnification claim or notice is timely given under, and in accordance with, this Article VII with respect to any representation or warranty prior to the applicable expiration date, such claim shall be preserved until such claim is finally resolved and satisfied. The representations and warranties of the Parent Parties contained in this Agreement shall survive the Closing Date until 11:59 p.m. (New York time) on the date that is twelve (12) months following the Closing Date; *provided*, that in the event of a claim for Fraud, such representation or warranty shall survive indefinitely following the Closing Date; provided, further, that the Fundamental Representations of the Parent Parties shall remain in full force and effect until 11:59 p.m. (New York time) on the date that is thirty (30) days following the expiration of the statute of limitations, including tolling, applicable to such representation or warranty; provided, further, if an indemnification claim or notice is timely given under, and in accordance with, this Article VII with respect to any representation or warranty prior to the applicable expiration date, such claim shall be preserved until such claim is finally resolved and satisfied. For the avoidance of doubt, it is the intention of the Parties that the foregoing respective survival periods and termination dates supersede any applicable statutes of limitations that would otherwise apply to such representations and warranties and the right to make indemnification claims in respect thereof under this Agreement, and the Company Indemnitors, on the one hand, and Parent and Parent Holdco LLC, on the other hand, respectively, hereby waive any defenses based on the statute of limitations to the extent inconsistent with the survival periods set forth herein.

7.2. Indemnification.

- (a) From and after the consummation of the Combination Transactions, the Company Indemnitors shall (x) jointly and severally, to the extent of the Deferred Amount, and (y) as to any amounts exceeding the Deferred Amount, severally, and not jointly, in accordance with each Company Indemnitor's Pro Rata Share, indemnify and hold harmless Parent and its directors, officers, employees, affiliates (including the Final Surviving Company), agents and other representatives (the "*Parent Indemnified Parties*"), from and against all Losses paid, incurred, suffered or sustained by the Parent Indemnified Parties, or any of them, relating to, resulting from or arising out of any of the following:
- (i) (A) the failure of any Fundamental Representation of the Company Parties or Tax Representation to be true and correct, in each case as of the Closing Date (or, if given as of a specific date, at and as of such date), (B) the failure of any Critical Representation to be true and correct, in each case as of the Closing Date (or, if given as of a specific date, at and as of such date), and (C) the failure of any other representation or warranty of the Company, Shareablee Holdco or Shareablee Merger Sub in this Agreement or the Certificates to be true and correct as of the Closing Date (or, if given as of a specific date, at and as of such date), in each case including any third party allegation that, if true, would result in or constitute such a failure or breach of any representation or warranty of the Company, Shareablee Holdco or Shareablee Merger Sub; *provided*, that all materiality qualifications (such as "material") in such representations and warranties shall be disregarded for the purposes of this <u>Article VII</u>, including in connection with determining whether a breach has occurred and the amount of Losses incurred; *provided*, *further*, any disclosure set forth in the Disclosure Schedule that is for informational purposes will be disregarded for purposes of this <u>Article VII</u> (other than in connection with any claim of Fraud), including in connection with determining whether a breach has occurred;
- (ii) any failure by the Company to perform or comply with any covenant or agreement applicable to it contained in this Agreement or the Related Agreements;
- (iii) regardless of the disclosure of any matter set forth in the Disclosure Schedule, any inaccuracy in any information set forth in the Allocation Schedule (including any inaccuracy in any information set forth in an updated Allocation Schedule required to be delivered under this Agreement) (including any claim by or on behalf of any actual or purported present or former equityholder of the Company, in his, her or its capacity as such, with respect to the Allocation Schedule);
- (iv) regardless of the disclosure of any matter set forth in the Disclosure Schedule, (I) any Unpaid Pre-Closing Taxes, Closing Transaction Expense Amount or Closing Indebtedness Amount, in each case, to the extent not taken into account in the Final Closing Statement; and (II) any and all obligations of the Company under that certain PPP Loan Escrow Agreement, including any payments required to be made or indemnity obligations by the Company thereunder.

- (b) From and after the consummation of the Combination Transactions, Parent and Parent Holdco LLC shall (x) jointly and severally indemnify and hold harmless each Company Stockholder and its directors, officers, employees, affiliates, heirs, beneficiaries, agents and other representatives (the "Company Stockholder Indemnified Parties" and together with the Parent Indemnified Parties, each an "Indemnified Party" and collectively, the "Indemnified Parties"), from and against all Losses paid, incurred, suffered or sustained by the Company Stockholder Indemnified Parties, or any of them, relating to, resulting from or arising out of any of the following:
- (i) (A) the failure of any Fundamental Representation of the Parent or Parent Holdco LLC to be true and correct, in each case as of the Closing Date (or, if given as of a specific date, at and as of such date), and (B) the failure of any other representation or warranty of the Parent or Parent Holdco LLC in this Agreement to be true and correct as of the Closing Date (or, if given as of a specific date, at and as of such date), in each case including any third party allegation that, if true, would result in or constitute such a failure or breach of any representation or warranty of the Parent or Parent Holdco LLC; provided, that all materiality qualifications (such as "material") in such representations and warranties shall be disregarded for the purposes of this <u>Article VII</u>, including in connection with determining whether a breach has occurred and the amount of Losses incurred;
- (ii) any failure by the Parent or Parent Holdco LLC to perform or comply with any covenant or agreement applicable to it contained in this Agreement or the Related Agreements;
- (c) The rights of the Indemnified Parties to indemnification, compensation or reimbursement, payment of Losses or any other remedy under this Agreement shall not be affected by any investigation conducted with respect to, or any knowledge acquired (or capable of being acquired) at any time, whether before or after the execution and delivery of this Agreement or the Closing Date (other than as expressly set forth in the Disclosure Schedule), with respect to the accuracy or inaccuracy of or compliance with, any representation, warranty, covenant or agreement made by a Party or any other matter. The waiver of any condition based on the accuracy of any such representation or warranty, or on the performance of or compliance with any such covenant or agreement, will not affect the right to indemnification, compensation or reimbursement, payment of Losses, or any other remedy based on any such representation, warranty, covenant or agreement. Other than with respect to any claim for Fraud, no Indemnified Party shall be required to show reliance on any representation, warranty, certificate or other agreement in order for such Indemnified Party to be entitled to indemnification, compensation or reimbursement hereunder. Nothing in this Agreement shall limit the right of any Indemnified Party to pursue remedies under any Related Agreement against the parties thereto.

7.3. Certain Limitations.

- (a) Except in the case of Fraud, (i) the Parent Indemnified Parties, as a group, may not recover any Losses pursuant to an indemnification claim under Section 7.2(a)(i)(C) (Other Representations) unless and until the Parent Indemnified Parties, as a group, shall have paid, incurred, suffered or sustained at least \$450,000 in Losses in the aggregate (the "Company Indemnitors Deductible"), at which time the Parent Indemnified Parties shall be entitled to recover solely such amounts in excess of the Company Indemnitors Deductible, and (ii) the Company Stockholder Indemnified Parties may not recover any Losses pursuant to an indemnification claim under Section 7.2(b)(i)(B) (Other Representations) unless and until the Company Stockholder Indemnified Parties shall have paid, incurred, suffered or sustained at least \$450,000 in Losses in the aggregate (the "Parent Parties Deductible"), at which time the Company Stockholder Indemnified Parties shall be entitled to recover solely such amounts in excess of the Parent Parties Deductible.
- (b) Subject to the limitations set forth in <u>Section 2.6</u> and this <u>Section 7.3</u>, (i) during the twelve (12) month survival period set forth in <u>Section 7.1</u>, the Parent Indemnified Parties shall be entitled to recover any Losses by way of setoff from Parent's obligation to issue Parent Common Stock (or to pay cash) in respect of the Deferred Amount for an indemnification claim made pursuant to <u>Section 7.2(a)</u>, and (ii) during the twenty four (24) month survival period set forth in <u>Section 7.1</u>, the Parent Indemnified Parties shall be entitled to recover any Losses by way of setoff from Parent's obligation to issue Parent Common Stock (or to pay cash) in respect of the remaining Deferred Amount, if any, to the extent such indemnification claim is made pursuant to <u>Section 7.2(a)</u>.
- (c) Any indemnity amounts payable to a Parent Indemnified Party in accordance with this $\frac{\text{Article VII}}{\text{Section 7.2(a)}(x)}$, and (ii) second, from the Company Indemnitors in

accordance with <u>Section 7.2(a)(y)</u>. Any indemnity amounts payable to a Company Stockholder Indemnified Party in accordance with this <u>Article VII</u> shall be satisfied either in cash or Parent Common Stock, on a Pro Rata Share basis, in the sole discretion of Parent and subject to <u>Section 2.3(c)</u>.

- (d) In no event shall the aggregate liability of the Company Indemnitors under this Article VII (in accordance with each Company Indemnitor's Pro Rata Share) (i) for all indemnification claims under Section 7.2(a)(i)(C) (Other Representations) of this Agreement exceed \$4,500,000, and (ii) for all other indemnification claims under this Agreement exceed \$40,624,210 (unless such indemnity claim is being made in respect of Fraud committed by such Company Indemnitor, Section 7.2(a)(i)(I)(B) (Critical Representations), Section 7.2(a)(iii) (Allocation Schedule) or Section 7.2(a)(iv) (Unpaid Pre-Closing Taxes, Closing Transaction Expense Amount or Closing Indebtedness Amount; PPP Loan Escrow), in which event there shall be no limitation on the liability of such Company Indemnitor hereunder or under applicable Law). In no event shall the aggregate liability of the Parent and Parent Holdco LLC under this Article VII for all indemnification claims under Section 7.2(b)(i)(B) (Other Representations) of this Agreement exceed \$4,500,000, and for all other indemnification claims under this Agreement exceed \$40,624,210 (unless such indemnity claim is being made in respect of Fraud committed by Parent or Parent Holdco LLC, in which event there shall be no limitation on the liability of Parent and Parent Holdco LLC hereunder or under applicable Law).
- (e) The amount of any Losses that are subject to indemnification under this <u>Article VII</u> shall be calculated net of the amount of any insurance proceeds, indemnification payments or reimbursements actually received by the Indemnified Parties from third parties (other than in respect of the Deferred Amount) in respect of such Losses (net of any reasonable and actual out-of-pocket costs or expenses incurred in obtaining such insurance, indemnification or reimbursement, including any increases in insurance premiums attributable to such recovery); *provided* that nothing in this <u>Section 7.3(e)</u> shall be construed as or give rise to an obligation to seek any such insurance, indemnification or reimbursement.
- (f) For purposes of this <u>Article VII</u>, each share of Parent Common Stock (including the Deferred Amount, as applicable) shall be deemed to have a value equal to the Parent Stock Price as of the date of the final determination of the relevant claim.
- (g) For the avoidance of doubt, any Losses for indemnification under this Agreement shall be determined without duplication of recovery due to the facts giving rise to such Losses constituting a breach of more than one representation, warranty, covenant or agreement.
- (h) The Parties acknowledge and agree that their sole and exclusive remedy with respect to any and all claims for any breach of any representation, warranty, covenant, agreement or obligation set forth herein or otherwise resulting from or arising out of this Agreement will be pursuant to the indemnification provisions set forth in this <u>Article VII</u>; *provided*, that the foregoing clause of this sentence shall not be deemed a waiver by any party of (i) any right to specific performance, equitable or injunctive relief, (ii) any right or remedy under any Related Agreement to which it is a party, or (iii) any right or remedy with respect to a Party's own Fraud, and all claims related thereto shall survive until thirty (30) days following the expiration of the applicable statute of limitations. Subject to the other limitations contained herein, the obligations of the Company Indemnitors under this <u>Article VII</u> shall not be reduced, offset, eliminated or subject to contribution by reason of any action or inaction by the Company that contributed to any inaccuracy or breach giving rise to such obligation, it being understood that the Company Indemnitors, not the Company, shall have the sole obligation for the indemnification obligations under this <u>Article VII</u>.

7.4. Claims for Indemnification; Resolution of Conflicts.

(a) Subject to the limitations set forth in <u>Section 7.3</u>, if an Indemnified Party wishes to make an indemnification claim under this <u>Article VII</u>, such Indemnified Party ("**Beneficiary**") shall deliver a written notice (an "**Indemnification Claim Notice**") to the party required to provide indemnification under this Agreement (the "**Indemnifying Party**") (i) stating that the Beneficiary has paid, incurred, suffered or sustained, or reasonably anticipates that it may pay, incur, suffer or sustain, Losses for the Beneficiary is entitled to indemnification under this <u>Article VII</u> (a "**Liability Claim**"), and (ii) specifying such Losses in reasonable detail (to the extent available), and, if applicable, the nature of the misrepresentation, breach of warranty or covenant to which such item is related. Any such Indemnification Claim Notice with respect to a Third Party Claim must be provided as soon as reasonably practicable, but no later than twenty (20) Business Days, after any Beneficiary becomes aware of any such Liability Claim; provided, however, that no delay on the part of the

Beneficiary in giving such Indemnification Claim Notice shall relieve the Indemnifying Party of any indemnification obligation hereunder unless (and then solely to the extent that) the Indemnifying Party is materially prejudiced by such delay. A Beneficiary may update an Indemnification Claim Notice from time to time to reflect any change in circumstances following the date thereof.

- (b) If the Indemnifying Party shall not object in writing within the twenty (20) Business Day period after receipt of an Indemnification Claim Notice by delivery of a written notice of objection containing a reasonably detailed description of the facts and circumstances supporting an objection to the applicable indemnification claim (an "Indemnification Claim Objection Notice"), such failure to so object shall be an irrevocable acknowledgment by the Indemnifying Party that the Beneficiary is entitled to the full amount of the claim for Losses set forth in such Indemnification Claim Notice.
- (c) In the event that the Indemnifying Party shall timely deliver an Indemnification Claim Objection Notice in accordance with Section 7.4(b), the Beneficiary and Indemnifying Party shall attempt in good faith to agree upon the rights of the respective Parties with respect to each of such claims. If the Beneficiary and Indemnifying Party should so agree, a memorandum setting forth such agreement shall be prepared and signed by both Parties. The Beneficiary and the Indemnifying Party shall be entitled to conclusively rely on any such memorandum.
- (d) In the event that there is a dispute relating to any Indemnification Claim Notice or Indemnification Claim Objection Notice that cannot be settled in accordance with <u>Section 7.4(c)</u>, either of Parent or the Stockholder Representative may file suit with respect to such dispute in any court having jurisdiction in accordance with <u>Section 9.11</u>.
- (e) Notwithstanding the foregoing, (i) any notice that is required to be delivered to any Company Indemnitors pursuant to this <u>Article VII</u> shall be deemed satisfied by delivery of such notice to the Stockholder Representative, and (ii) any notices required to be delivered by, or any actions that are required to be taken by, any Company Indemnitors pursuant to this <u>Article VII</u> shall be satisfied by delivery by, or action taken by, the Stockholder Representative (other than provisions related to an obligation to make or a right to receive any payments). Any rights of the Company Stockholder Indemnified Parties shall be exercised only by the Stockholder Representative and not by any individual Company Stockholder Indemnified Party. Any obligation by Parent or Parent Holdco LLC to make any payments to any Company Stockholder Indemnified Party under this <u>Article VII</u> shall be deemed to be satisfied by payment (or issuance) of such indemnifying payment to the Company Stockholder Indemnified Parties, based on their Pro Rata Shares implied by the Allocation Schedule.
- Third Party Claims. If any Indemnification Claim Notice identifies a Liability Claim brought by a Person that is not a Party or an Affiliate of a Party (a "Third Party Claim"), then the Indemnifying Party has the right, exercisable by written notice to the Beneficiary within 15 days after receipt of such Indemnification Claim Notice, to assume and conduct the defense of such Third Party Claim in accordance with the limits set forth in this Agreement with counsel selected by the Indemnifying Party and reasonably acceptable to the Beneficiary (such acceptance not to be unreasonably withheld, conditioned or delayed); provided, however, that (a) the Third Party Claim must not seek (and continue not to seek) equitable remedies, and the Third Party Claim, based upon the remedy sought, must not be reasonably likely to result in criminal liability of, or equitable remedies against, the Beneficiary, (b) the Third Party Claim must not relate to the Beneficiary's employees, material customers, material suppliers, or Taxes, and (c) the Indemnifying Party must expressly agree in writing that, as between the Indemnifying Party and the Beneficiary, the Indemnifying Party may only satisfy and discharge the Third Party Claim in accordance with the limits set forth in this Agreement (the conditions set forth in clauses (a) through (c) are, collectively, the "Litigation Conditions"). If the Indemnifying Party does not assume, or is not entitled to assume, the defense of a Third Party Claim in accordance with this Section 7.5, then the Beneficiary may continue to defend the Third Party Claim. If the Indemnifying Party has assumed the defense of a Third Party Claim as provided in this Section 7.5, then the Indemnifying Party will not be liable for any legal expenses incurred by the Beneficiary in connection with the defense of the Third Party Claim; provided, however, that, if any of the Litigation Conditions cease to be met, or the Indemnifying Party fails to take reasonable steps necessary to defend diligently such Third Party Claim, then the Beneficiary may assume its own defense, and the Indemnifying Party shall be liable for all reasonable costs or expenses paid or incurred in connection with such defense. The Indemnifying Party or the Beneficiary, as the case may be, has the right to participate in (but not control), at its own expense, the defense of any Third Party Claim that the other Party is defending as provided in this Agreement. The Indemnifying Party, if it has assumed the defense of any Third Party Claim as provided in this Agreement, may not, without the prior written

consent of the Beneficiary, consent to a settlement of, or the entry of any judgment arising from, any such Third Party Claim that (A) does not include as an unconditional term thereof the giving by the claimant or the plaintiff to the Beneficiary a complete release from all liability in respect of such Third Party Claim, or (B) grants any injunctive or equitable relief. The Beneficiary has the right to settle any Third Party Claim, the defense of which has not been assumed by the Indemnifying Party.

7.6. Stockholder Representative.

- By voting in favor of the adoption of this Agreement, the approval of the principal terms of the Combination Transactions, and the consummation of the Combination Transaction, executing a Joinder Agreement or participating in the Combination Transactions and receiving the benefits thereof, including the right to receive the consideration payable in connection with the Combination Transactions, and without any further action of any of the Company Stockholders, the Company Warrantholder or the Company, each Company Stockholder and the Company Warrantholder shall be deemed to have approved the designation of, and hereby designates, as of the Closing, Shareholder Representative Services LLC as the representative, exclusive agent and attorney-in-fact for and on behalf of the Company Stockholders and Company Warrantholder for all purposes in connection with this Agreement (including Article VII) and the agreements ancillary hereto, including to give and receive notices and communications, to authorize satisfaction of claims by Parent, to object to such payments, to agree to, negotiate, enter into settlements and compromises of, and demand arbitration and comply with orders of courts and awards of arbitrators with respect to such claims, and to take all other actions that are either (i) necessary or appropriate in the judgment of the Stockholder Representative for the accomplishment of the foregoing, or (ii) permitted by the terms of this Agreement. Notwithstanding the foregoing, the Stockholder Representative shall have no obligation to act on behalf of the Company Stockholders or Company Warrantholder, except as expressly provided herein, and for purposes of clarity, there are no obligations of the Stockholder Representative in any ancillary agreement, schedule, exhibit or the Disclosure Schedule. The Stockholder Representative may resign at any time upon at least ten (10) days prior written notice to the Company Stockholders and Company Warrantholder. Such agency may be changed by the Company Stockholders and Company Warrantholder from time to time upon not less than ten (10) days prior written notice to Parent; provided, that the Stockholder Representative may not be removed unless the former holders of a majority of Company Capital Stock (on an as-converted basis) agree to such removal and to the identity of the substituted agent. A vacancy in the position of Stockholder Representative may be filled by the former holders of a majority of Company Capital Stock (on an as-converted basis). No bond shall be required of the Stockholder Representative. The engagement fee of the Stockholder Representative shall be paid by the Company prior to the Closing and considered a Transaction Expense for purposes of this Agreement. After the Closing, notices or communications to or from the Stockholder Representative shall constitute notice to or from the Company Stockholders and Company Warrantholder. Furthermore, each Company Stockholder and the Company Warrantholder agrees to the appointment of Tania Yuki Mustapic as the sole member of the "Advisory Committee" under that certain engagement letter, dated as of the date hereof, with the Stockholder Representative.
- Neither the Stockholder Representative nor its members, managers, directors, officers, contractors, agents and employees (collectively, the "Stockholder Representative Group") shall incur liability of any kind with respect to any action or omission by the Stockholder Representative in connection with their services pursuant to this Agreement and the agreements ancillary hereto, except to the extent resulting from the bad faith, gross negligence or willful misconduct of the Stockholder Representative. The Company Stockholders and Company Warrantholder shall, jointly and severally, indemnify, defend and hold harmless the Stockholder Representative Group from and against any and all losses, liabilities, damages, penalties, fines, forfeitures, actions, fees, claims, judgments, amounts paid in settlement and reasonable out-of-pocket costs and expenses (including the out-of-pocket fees and expenses of counsel and experts and their staffs, in connection with seeking recovery from insurers and all expense of document location, duplication and shipment) (collectively, "Representative Losses") to the extent arising out of or in connection with the Stockholder Representative's execution and performance of this Agreement and the agreements ancillary hereto or thereto, in each case as such Representative Loss is suffered or incurred; provided, that in the event that any such Representative Loss is finally adjudicated to have been caused by the bad faith, gross negligence or willful misconduct of the Stockholder Representative, the Stockholder Representative shall promptly reimburse the Company Stockholders and Company Warrantholder the amount of such indemnified Representative Loss to the extent attributable to such bad faith, gross negligence or willful misconduct. Representative Losses may be recovered by the Stockholder Representative from any funds that become payable to the Company Stockholders and Company Warrantholder under this Agreement at such time as such amounts would otherwise be distributable to the Company Stockholders and Company Warrantholder; provided, that while the Stockholder

Representative may be paid from the aforementioned source of funds, this does not relieve the Company Stockholders and Company Warrantholder from their obligation to promptly pay such Representative Losses as they are suffered or incurred. In no event shall the Stockholder Representative be required to advance its own funds on behalf of the Company Stockholders or Company Warrantholder, or otherwise incur any financial liability in the exercise or performance of any of its powers, rights, duties or privileges or pursuant to this Agreement or the transactions contemplated hereby or thereby. Furthermore, the Stockholder Representative shall not be required to take any action unless the Stockholder Representative has been provided with funds, security or indemnities which, in its determination, are sufficient to protect the Stockholder Representative against the costs, expenses and liabilities which may be incurred by the Stockholder Representative in performing such actions. Notwithstanding anything in this Agreement to the contrary, any restrictions or limitations on liability or indemnification obligations of, or provisions limiting the recourse against non-parties otherwise applicable to, the Company Stockholders and Company Warrantholder set forth elsewhere in this Agreement are not intended to be applicable to the indemnities provided to the Stockholder Representative hereunder. The Company Stockholders and Company Warrantholder acknowledge and agree that the foregoing indemnities and immunities shall survive the resignation or removal of the Stockholder Representative, the termination of this Agreement and the Closing.

- The powers, immunities and rights to indemnification granted to the Stockholder Representative Group hereunder: (i) are coupled with an interest and shall be irrevocable and survive the death, incompetence, bankruptcy or liquidation of any Company Stockholder or the Company Warrantholder, and shall be binding on any successor thereto, and (ii) shall survive the delivery of an assignment by any Company Stockholder or the Company Warrantholder of the whole or any fraction of his, her or its interest in the Deferred Amount. A decision, act, consent or instruction of the Stockholder Representative under this Agreement shall constitute a decision of the Company Stockholders and Company Warrantholder, and shall be final, binding and conclusive upon the Company Stockholders, Company Warrantholder and their respective successors as if expressly confirmed and ratified in writing by the Company Stockholders and Company Warrantholder, and all defenses which may be available to any Company Stockholder or the Company Warrantholder to contest, negate or disaffirm the action of the Stockholder Representative taken in good faith under this Agreement are waived. Parent may rely upon any such decision, act, consent or instruction of the Stockholder Representative as being the decision, act, consent or instruction of the Company Stockholders and Company Warrantholder. Parent is hereby relieved from any Liability to any Person for any acts done by Parent in accordance with such decision, act, consent or instruction of the Stockholder Representative. The Stockholder Representative shall be entitled to: (A) rely upon the Allocation Schedule, (B) rely upon any signature believed by it to be genuine, and (C) reasonably assume that a signatory has proper authorization to sign on behalf of the applicable Company Stockholder, Company Warrantholder or other party.
- **7.7. Treatment of Payments**. Any payment under <u>Article VII</u> of this Agreement shall be treated by the Parties, Shareablee Holdco Stockholders and Shareablee Holdco Warrantholder, including for U.S. federal, state, local and non-U.S. income Tax purposes, as an adjustment to the consideration paid pursuant to <u>Article II</u>, unless otherwise required by applicable Law.

ARTICLE VIII TERMINATION, EFFECT OF TERMINATION

- **8.1. Termination**. This Agreement may be terminated and the Combination Transactions may be abandoned at any time by Parent, by written notice to the Company, if the Company Written Consent is not obtained by the Company and delivered to Parent within one (1) hour following the execution and delivery of this Agreement.
- **8.2. Effect of Termination**. In the event of termination of this Agreement as provided in <u>Section 8.1</u>, this Agreement shall become void and there shall be no Liability or obligation on the part of Parent, the Company, or its officers, directors or stockholders; *provided*, that the Company shall promptly pay to Parent any and all documented legal fees and expenses incurred by Parent in the diligence, preparation and negotiation of this Agreement, the Related Agreements and the transactions contemplated hereby and thereby; *provided*, *further*, nothing herein shall relieve any Party from liability for the willful breach of any of its representations, warranties or covenants contained herein that occurred prior to such termination; and *provided*, *further*, that the provisions of <u>Section 6.5</u> (Confidentiality), <u>Section 6.2</u> (Public Disclosure), <u>Section 7.6</u> (Stockholder Representative), <u>Article IX</u> (General Provisions) and this <u>Section 8.2</u> shall remain in full force and effect and survive any termination of this Agreement. For purposes of this Agreement, "*willful breach*" shall mean an

intentional act or intentional omission taken by a Party where such Party knows that such action or omission would constitute a material breach of this Agreement.

ARTICLE IX GENERAL PROVISIONS

- **9.1. Notices**. (a) Any notice or other communication required or permitted to be delivered to any party under this Agreement shall be in writing and shall be deemed properly delivered, given and received (i) upon receipt when delivered by hand, (ii) upon transmission, if sent by electronic mail transmission (with confirmation of transmission and no "error" message or other notification of non-delivery is generated and received by the sending Party), or (iii) one (1) Business Day after being sent by courier or express delivery service; *provided*, that in each case the notice or other communication is sent to the address or electronic mail address set forth beneath the name of such party below (or to such other address or electronic mail address as such party shall have specified in a written notice given to the other Parties:
 - (b) if to Parent or Parent Holdco LLC, to:

comScore, Inc. 11950 Democracy Drive, Suite 600 Reston, VA 20190 Attention: General Counsel Email:

with a copy to (which shall not constitute notice):

Vinson & Elkins LLP 1001 Fannin Street, Suite 2500 Houston, TX 77002 Attention: Benji Barron Email:

(c) if to the Company (prior to the Closing), to:

Shareablee, Inc. 123 William Street, Level 19 New York, NY 10038 Attention: Tania Yuki Mustapic Email:

with a copy to (which shall not constitute notice):

Cole Schotz, P.C. 1325 Avenue of the America, 19th Floor New York, NY 10019 Attention: Jordan Fisch Email:

- (d) If to a Company Stockholder, to its address as set forth in the Allocation Schedule.
- (e) If to the Stockholder Representative:

Shareholder Representative Services LLC 950 17th Street, Suite 1400 Denver, CO 80202 Attention: Managing Director Email:

9.2. Interpretation. The words "include," "includes" and "including" when used herein shall be deemed in each case to be followed by the words "without limitation." The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or

interpretation of this Agreement. The words "hereof," "herein" and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement as a whole and not to any particular provision of this Agreement. Article, section, paragraph, and schedule references are to the articles, sections, paragraphs, and schedules of this Agreement unless otherwise specified. The meaning assigned to each term defined herein shall be equally applicable to both the singular and the plural forms of such term, and words denoting any gender shall include all genders. The word "extent" and the phrase "to the extent" when used in this Agreement shall mean the degree to which a subject or other thing extends, and such word or phrase shall not merely mean "if." Dollar thresholds shall not be indicative of what is material. The terms "any" or "or" are not exclusive. A reference to any specific legislation or to any provision of any legislation shall include any amendment to, and any modification or re-enactment thereof, any legislative provision substituted therefor and all regulations and statutory instruments issued thereunder or pursuant thereto. When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period will be excluded.

- **9.3. Counterparts.** This Agreement may be executed in any number of counterparts, each of which shall be enforceable against the Parties actually executing such counterparts, and all of which together shall constitute one instrument. The exchange of copies of this Agreement and signature pages by email in .pdf or .tif format (and including, without limitation, any electronic signature complying with the U.S. ESIGN Act of 2000, e.g., www.docusign.com), or by any other electronic means intended to preserve the original graphic and pictorial appearance of a document, or by combination of such means, shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes. Such execution and delivery shall be considered valid, binding and effective for all purposes.
- **9.4. Entire Agreement; Assignment.** This Agreement, the schedules hereto, the Disclosure Schedule, the Confidentiality Agreement, and the Related Agreements: (a) constitute the entire agreement among the Parties with respect to the subject matter hereof and supersede all prior agreements and understandings both written and oral, among the Parties with respect to the subject matter hereof, (b) except as set forth in <u>Section 5.3</u> with respect to the Company Indemnitees and <u>Article VII</u> with respect to the Parent Indemnified Parties, are not intended to confer upon any other Person any rights or remedies hereunder, and (c) shall not be assigned by operation of law or otherwise; *provided*, that Parent may assign its rights and delegate its obligations hereunder to its Affiliates as long as Parent remains ultimately liable for all of Parent's obligations hereunder.
- **9.5. Severability**. In the event that any provision of this Agreement or the application thereof, becomes or is declared by a court of competent jurisdiction to be illegal, void or unenforceable, the remainder of this Agreement shall continue in full force and effect and the application of such provision to other Persons or circumstances shall be interpreted so as reasonably to effect the intent of the Parties. The Parties further agree to replace such illegal, void or unenforceable provision of this Agreement with a valid and enforceable provision that shall achieve, to the extent possible, the economic, business and other purposes of such illegal, void or unenforceable provision.
- **9.6. Waiver**. At any time following the Closing, Parent, on the one hand, and the Stockholder Representative, on the other hand, may, to the extent legally allowed, (a) extend the time for the performance of any of the obligations of the other Party or Parties, (b) waive any inaccuracies in the representations and warranties made to such Party or Parties contained herein or in any document delivered pursuant hereto, and (c) waive compliance with any of the covenants, agreements or conditions for the benefit of such Party or Parties contained herein. Any agreement on the part of a Party or Parties to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such Party or Parties. For purposes of this Section 9.6, the Company Stockholders and Company Warrantholder are deemed to have agreed that any waiver signed by the Stockholder Representative shall be binding upon and effective against all Shareablee Holdco Stockholders and the Shareablee Holdco Warrantholder whether or not they have signed such waiver.
- **9.7. Amendment**. Subject to applicable Law, this Agreement may be amended by the Parties at any time by execution and delivery of an instrument in writing signed on behalf of each of the Parties; *provided* that the execution and delivery of an instrument in writing signed on behalf of the Stockholder Representative shall not be required for any such amendment that occurs prior to the Closing and does not adversely affect the Stockholder Representative. Notwithstanding the foregoing or anything to the contrary herein, the provisions of Section 5.3 and this sentence shall not be amended or waived in a manner adverse to any Company Indemnitee without the prior written consent of the affected Company Indemnitee. For purposes of this Section 9.7, the Company Stockholders and Company Warrantholder agree that any amendment of this Agreement signed by

the Stockholder Representative shall be binding upon and effective against the Shareablee Holdco Stockholders and Shareablee Holdco Warrantholder whether or not they have signed such amendment.

- **9.8. Specific Performance.** The Parties agree that, in the event of any breach or threatened breach by the other Party or Parties or the Stockholder Representative of any covenant, obligation or other agreement set forth in this Agreement, (a) each Party shall be entitled, without any proof of actual damages (and in addition to any other remedy that may be available to it), to specific performance to enforce the observance and performance of such covenant, obligation or other agreement and an injunction preventing or restraining such breach or threatened breach, and (b) no Party shall be required to provide or post any bond or other security or collateral in connection with any such decree, order or injunction or in connection with any related Legal Proceeding.
- **9.9. Other Remedies**. Except as otherwise set forth herein, any and all remedies herein expressly conferred upon a Party shall be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such party, and the exercise by a Party of any one remedy shall not preclude the exercise of any other remedy.
- **9.10. Governing Law**. This Agreement shall be governed by and construed in accordance with the internal laws of the State of Delaware without giving effect to any choice or conflict of Law provision or rule (whether of the State of Delaware or any other jurisdiction).

9.11. Exclusive Jurisdiction; Waiver of Jury Trial.

- (a) ANY LEGAL PROCEEDING ARISING OUT OF OR BASED UPON THIS AGREEMENT, THE RELATED AGREEMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY MAY BE INSTITUTED FIRST, IN THE COURT OF CHANCERY WITHIN NEW CASTLE COUNTY IN THE STATE OF DELAWARE (AND ANY APPELLATE COURT THEREOF LOCATED WITHIN SUCH COUNTY) AND TO THE EXTENT SUCH COURT OF CHANCERY (OR APPELLATE COURT THEREOF LOCATED WITHIN SUCH COUNTY) LACKS JURISDICTION OVER THE MATTER, THE FEDERAL COURTS OF THE UNITED STATES OF AMERICA LOCATED WITHIN NEW CASTLE COUNTY IN THE STATE OF DELAWARE (OR APPELLATE COURT THEREOF LOCATED WITHIN SUCH COUNTY), AND EACH PARTY IRREVOCABLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF SUCH COURTS IN ANY SUCH LEGAL PROCEEDING. SERVICE OF PROCESS, SUMMONS, NOTICE OR OTHER DOCUMENT BY MAIL TO SUCH PARTY'S ADDRESS SET FORTH HEREIN SHALL BE EFFECTIVE SERVICE OF PROCESS FOR ANY LEGAL PROCEEDING BROUGHT IN ANY SUCH COURT. THE PARTIES IRREVOCABLY AND UNCONDITIONALLY WAIVE ANY OBJECTION TO THE LAYING OF VENUE OF ANY LEGAL PROCEEDING IN SUCH COURTS AND IRREVOCABLY WAIVE AND AGREE NOT TO PLEAD OR CLAIM IN ANY SUCH COURT THAT ANY SUCH LEGAL PROCEEDING BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM.
- (b) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT OR THE RELATED AGREEMENTS IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES AND, THEREFORE, EACH SUCH PARTY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LEGAL PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE RELATED AGREEMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY. EACH PARTY TO THIS AGREEMENT CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT SEEK TO ENFORCE THE FOREGOING WAIVER IN THE EVENT OF A LEGAL PROCEEDING, (B) SUCH PARTY HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) SUCH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 9.11(b).
- **9.12. Rules of Construction**. The Parties agree that they have been represented by counsel during the negotiation and execution of this Agreement and, therefore, waive the application of any law, regulation, holding or rule of construction providing that ambiguities in an agreement or other document shall be construed against the party drafting such agreement or document.

9.13. Acknowledgments.

- The Company acknowledges and agrees that it has conducted its own independent review and analysis of the business, assets, condition, operations and prospects of Parent and Parent Holdco LLC and acknowledges that the Company and its Representatives have been provided with access to the properties, premises and records of Parent for this purpose. In entering into this Agreement, the Company has relied solely upon its own investigation and analysis and the representations and warranties of the Parent Parties set forth in Article IV, and the Company acknowledges and agrees that, except for the representations and warranties of the Parent Parties expressly set forth in Article IV, as qualified by the Parent SEC Documents, neither Parent, nor Parent Holdco LLC, nor any of their respective Representatives nor any other Person acting on Parent's or Parent Holdco LLC's behalf makes or has made, and the Company is not relying on and has not relied on, any representation or warranty, either express or implied, as to the accuracy or completeness of any of the information provided or made available to the Company or any of its Representatives, or otherwise with respect to Parent, Parent Holdco LLC, any of their businesses or the Combination Transactions. Without limiting the generality of the foregoing, the Company acknowledges and agrees that neither Parent, Parent Holdco LLC nor any of their respective Representatives or any other Person has made, and the Company is not relying on and has not relied on, any representation or warranty to the Company or any of its Representatives with respect to (i) any projections, estimates or budgets for Parent, or (ii) any materials, documents or information relating to Parent, Parent Holdco LLC or their respective businesses made available to the Company or any of its Representatives in any "data room," online data site, confidential memorandum, other offering materials or otherwise, except, in the case of (i) and (ii), as set forth in the representations and warranties set forth in <u>Article IV</u>.
- Each of Parent and Parent Holdco LLC has conducted its own independent review and analysis of the business, assets, condition, operations and prospects of the Company, Shareablee Holdco and Shareablee Merger Sub and acknowledges that each of Parent and Parent Holdco LLC has been provided with access to the properties, premises and records of the Company, Shareablee Holdco and Shareablee Merger Sub for this purpose. In entering into this Agreement, each of Parent and Parent Holdco LLC has relied solely upon its own investigation and analysis and the representations and warranties of the Company Parties set forth in Article III and in the Certificates, and each of Parent and Parent Holdco LLC acknowledges and agrees that, except for the representations and warranties of the Company Parties expressly set forth in Article III, the Certificates and the Related Agreements, neither the Company, Shareablee Holdco, Shareablee Merger Sub nor any of their respective Representatives nor any other Person has made, and neither Parent nor Parent Holdco LLC is relying on or has relied on, any representation or warranty, either express or implied, as to the accuracy or completeness of any of the information provided or made available to Parent or Parent Holdco LLC or any of their Representatives, or otherwise with respect to the Company, Shareablee Holdco, Shareablee Merger Sub or their respective businesses or the Combination Transactions. Without limiting the generality of the foregoing, Parent and Parent Holdco LLC acknowledge and agree that neither the Company, Shareablee Holdco, Shareablee Merger Sub nor any of their respective Representatives or any other Person has made, and neither Parent nor Parent Holdco LLC is relying on or has relied on, any representation or warranty to Parent or Parent Holdco LLC or any of their Representatives or any other Person with respect to (i) any projections, estimates or budgets for the Company, or (ii) any materials, documents or information relating to the Company made available to Parent, Parent Holdco LLC or any of their Representatives in any "data room," online data site, confidential memorandum, other offering materials or otherwise, except, in the case of (i) and (ii) as set forth in the representations and warranties set forth in Article III, the Certificates or the Related Agreements.
- (c) The rights and remedies of the Parties are cumulative and not alternative. Neither any failure nor any delay by any Party in exercising any right, power, or privilege under this Agreement or any of the documents referred to in this Agreement will operate as a waiver of such right, power, or privilege and no single or partial exercise of any such right, power, or privilege will preclude any other or further exercise of such right, power, or privilege or the exercise of any other right, power, or privilege. To the maximum extent permitted by applicable Law, (i) no waiver that may be given by a Party will be applicable except in the specific instance for which it is given; and (ii) no notice to or demand on one Party will be deemed to be a waiver of any obligation of that Party or of the right of the Party giving such notice or demand to take further action without notice or demand as provided in this Agreement or the documents referred to in this Agreement by such Person.

9.14. Release by Company Stockholders and Company Warrantholder.

(a) Effective as of the Shareablee Merger Effective Time, upon execution of a Letter of Transmittal, each Company Stockholder and the Company Warrantholder, on behalf of himself, herself or itself

and each of his, her or its past, present and future controlled Affiliates, parent(s) and subsidiary companies, representatives, and assigns (each, an "Company Stockholder Releasing Party" and, collectively, the "Company Stockholder Releasing Parties") will absolutely, unconditionally and irrevocably release, acquit and forever discharge the Company and each of its respective past, present and future controlled Affiliates (including Shareablee Holdco), parent(s) and subsidiary companies, joint ventures, predecessors, successors and assigns, and their respective past, present and future representatives, investors, equityholders, insurers and indemnitees, firms, corporations, limited liability companies, partnerships, trusts, associations, organizations, stockholders, members, managers, directors, officers, employees, partners, trustees, principals, consultants, contractors, family members, heirs, executors, administrators, predecessors, successors and assigns (collectively the "Released Parties"), of and from any and all manner of action or inaction, cause or causes of action, encumbrances, contractual obligations, promises, Liabilities or damages (whether for compensatory, special, incidental or punitive damages, equitable relief or otherwise) of any kind or nature whatsoever, past, present or future, at law, in equity or otherwise (including with respect to conduct which is negligent, grossly negligent, willful, intentional, with or without malice, or a breach of any duty, applicable Law or rule), whether known or unknown, whether fixed or contingent, whether concealed or hidden, whether disclosed or undisclosed, whether liquidated or unliquidated, whether foreseeable or unforeseeable, whether anticipated or unanticipated, whether suspected or unsuspected ("Claims"), which such Company Stockholder Releasing Parties, or any of them, ever have had or ever in the future may have against the Released Parties, or any of them, in each case, to the extent arising solely as a result of the ownership or purported ownership of any Company Securities or other security or interest of the Company and any Equity Interests of Shareablee Holdco, and which, in each case, are based on acts, events or omissions occurring prior to or contemporaneously with the Shareablee Merger Effective Time (the "Company Stockholder Released Claims"); provided, however, that the foregoing release shall not release, impair or diminish, and the term "Company Stockholder Released Claims" shall not include, in any respect (i) a Company Stockholder's or the Company Warrantholder's right pursuant to this Agreement and the Related Agreements, including the right to receive the consideration set forth in the Allocation Schedule; (ii) any Claims for indemnification, insurance benefits, reimbursement or advancement of expenses in such Company Stockholder's or the Company Warrantholder's capacity as an Employee under the Company's organizational documents or any indemnification agreement in effect as of the date hereof (or any fiduciary insurance policy maintained by the Company or the Final Surviving Company for the benefit of a Company Stockholder or the Company Warrantholder, or any indemnification agreements with a Company Stockholder, Company Warrantholder or its board designee) with respect to any act, omission, event or transaction occurring prior to or contemporaneously with the Shareablee Merger Effective Time; or (iii) the rights of any Company Stockholder Releasing Party in his or her capacity as an Employee; *provided*, *further*, that the "Company Stockholder Released Claims" shall include any Claims that a Company Stockholder Releasing Party may have in right of consideration to be received in connection with this Agreement other than as set forth in the Allocation Schedule.

- (b) Upon execution of a Joinder Agreement, each Company Stockholder and the Company Warrantholder shall represent and acknowledge that he, she or it has read this release, this Agreement and the Related Agreements, and understands their terms and has been given sufficient opportunity to review this release, this Agreement and the Related Agreements, and to ask questions of the Company's Representatives. Each Company Stockholder and the Company Warrantholder will further represent that, in signing the Joinder Agreement, he, she or it does not rely, and has not relied, on any representation or statement made by any Representative of Parent or any other Person with respect to the subject matter, basis or effect of this release or otherwise, except such express representations and warranties set forth in this Agreement.
- (c) Without limiting the generality of Section 9.14(a), with respect to the Company Stockholder Released Claims, each Company Stockholder and the Company Warrantholder, upon execution of a Joinder Agreement, will acknowledge that he, she or it is familiar with Section 1542 of the Civil Code of the State of California ("Section 1542") and expressly waives all rights under Section 1542 and any similar applicable Law or common law principle in any applicable jurisdiction prohibiting or restricting the waiver of unknown claims. Section 1542 reads as follows:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS THAT THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY."

- (d) Notwithstanding the provisions of Section 1542 or any similar applicable Law or common law principle in any applicable jurisdiction, upon execution of a Joinder Agreement, each Company Stockholder and the Company Warrantholder will expressly acknowledge that the foregoing release is intended to include in its effect all claims within the scope of such release which any Company Stockholder or the Company Warrantholder does not know or suspect to exist in his, her or its favor against any of the Released Parties (including, without limitation, unknown and contingent claims), and that the foregoing release expressly contemplates the extinguishment of all such claims (except to the extent expressly set forth in this Section 9.14).
- **9.15. No Third Party Beneficiaries**. Nothing in this Agreement, except for <u>Section 5.3</u> with respect to the Company Indemnitees and <u>Article VII</u> with respect to the Parent Indemnified Parties, is intended to, or shall be construed to, confer upon any other Person any rights or remedies hereunder.

[Signature pages to follow]

In Witness Whereof, Parent, Parent Holdco LLC, Parent Merger Sub, the Company, Shareablee Holdco, Shareablee Merger Sub and the Stockholder Representative have caused this Agreement to be signed, all as of the date first written above.

COMSCORE, INC.

By: /s/ William P. Livek

Name: William P. Livek

Title: Chief Executive Officer and Executive Vice Chairman

SS MEDIA HOLDCO, LLC

By: /s/ William P. Livek

Name: William P. Livek

Title: President

SS MEDIA MERGER SUB, INC.

By: /s/ William P. Livek

Name: William P. Livek

Title: President

Signature Page to Agreement and Plan of Merger

In Witness Whereof, Parent, Parent Holdco LLC, Parent Merger Sub, the Company, Shareablee Holdco, Shareablee Merger Sub and the Stockholder Representative have caused this Agreement to be signed, all as of the date first written above.

SHAREABLEE, INC.

By: /s/ Tania Yuki Mustapic

Name: Tania Yuki Mustapic Title: Chief Executive Officer

SHAREABLEE HOLDCO, INC.

By: /s/ Tania Yuki Mustapic

Name: Tania Yuki Mustapic
Title: Chief Executive Officer

SHAREABLEE MERGER SUB, INC.

By: /s/ Tania Yuki Mustapic

Name: Tania Yuki Mustapic Title: Chief Executive Officer

Signature Page to Agreement and Plan of Merger

In Witness Whereof, Parent, Parent Holdco LLC, Parent Merger Sub, the Company, Shareablee Holdco, Shareablee Merger Sub and the Stockholder Representative have caused this Agreement to be signed, all as of the date first written above.

SHAREHOLDER REPRESENTATIVE SERVICES LLC,

solely in its capacity as the Stockholder Representative

By: /s/ Corey Quinlan

Name: Corey Quinlan

Title: Director

Signature Page to Agreement and Plan of Merger

ANNEX A CERTAIN DEFINED TERMS

- "2022 Statement" shall have the meaning given in Section 2.6(d)(i).
- "Accredited Investor" shall have the meaning given in Rule 501 of Regulation D promulgated under the Securities Act.
- "Adjustment Amounts" shall have the meaning given in Section 2.10(a).
- "Adjustment Resolution Period" shall have the meaning given in Section 2.10(b).
- "Adjustment Review Period" shall have the meaning given in Section 2.10(b).
- "Affiliate" shall mean, with respect to any Person, any other Person directly or indirectly controlling, directly or indirectly controlled by, or under direct or indirect common control with, such Person or, in the case of an individual, a member of such Person's immediate family; or, if such Person is a partnership or a limited liability company, any general partner or managing member, as applicable, of such Person or a Person controlling any such general partner or managing member. For purposes of this definition, "control" (including the correlative terms "controlling", "controlled by" and "under common control with") shall mean the power, directly or indirectly, to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by Contract or otherwise.
 - "Aggregate Option Exercise Price" shall have the meaning given in Section 3.2(d).
 - "Agreement" shall have the meaning given in the opening paragraph.
 - "Allocation Schedule" shall have the meaning given in Section 2.4(a).
- "Anti-Corruption Laws" shall mean any Law governing financial recordkeeping and reporting requirements or for the prevention or punishment of public or commercial corruption and bribery, including the U.S. Currency and Foreign Transaction Reporting Act of 1970, the U.S. Money Laundering Control Act of 1986, the U.S. Foreign Corrupt Practices Act of 1977, 18 U.S.C. §201, the UK Bribery Act 2010 and any applicable money laundering-related or anti-corruption Laws of any other jurisdiction.
- "Approved Data Consents" shall mean for each Data Partner, a consent that satisfies each of the "Data Consent Requirements" set forth opposite the name of such Data Partner on Schedule C-2.
 - "As-Converted Stockholders" shall mean the holders of the As-Converted Shares.
- "As-Converted Shares" shall mean a number, in each case as of immediately prior to the Initial Merger Effective Time, equal to (a) the total number of shares of Shareablee Holdco Common Stock outstanding, *plus* (b) the total number of shares of Shareablee Holdco Series A Preferred Stock outstanding, *plus* (c) the total number of shares of Shareablee Holdco Common Stock issuable upon the exercise of the Shareablee Holdco Warrant.
 - "Balance Sheet Date" shall have the meaning given in Section 3.7(b).
 - "Beneficiary" shall have the meaning given in Section 7.4(a).
- "Business Day" shall mean each day that is not a Saturday, Sunday or other day on which the Federal Reserve Bank New York is closed.
- "CARES Act" shall mean the Coronavirus Aid, Relief, and Economic Security Act (Public Law 116-136), signed into law on March 27, 2020.
- "Certificate of Incorporation" shall mean the second amended and restated certificate of incorporation of the Company, dated May 18, 2018.

"Certificates" shall mean certificates or other instruments delivered by the Company in connection with the Closing of the transactions contemplated by this Agreement, including the FIRPTA Compliance Certificate and the Estimated Closing Statement.

"Change of Control Event" means an event or series of events by which any Person or "group" (within the meaning of Section 13(d)(3) of the Exchange Act, but excluding any employee benefit plan of such Person and its subsidiaries, and any Person acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan) (a) becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act), directly or indirectly, of 50% or more of the equity securities of Parent on a fully-diluted basis or (b) has the power to elect or appoint a majority of the directors constituting the full Parent Board.

"Charter Documents" shall have the meaning given in Section 3.1(b).

"Claimant" shall have the meaning given in Section 3.11(j)(i).

"Claims" shall have the meaning given in Section 9.14(a).

"Closing" shall have the meaning given in Section 1.6.

"Closing Cash Amount" shall mean, as of immediately prior to Closing, all unrestricted cash, cash equivalents (including money market accounts, money market funds, money market instruments, certificates of deposit and demand deposits) and marketable securities of the Company, determined on a consolidated basis in accordance with GAAP. For the avoidance of doubt, "Cash" shall (x) exclude Restricted Cash, (y) be calculated net of all issued but uncleared checks and drafts issued by the Company to the extent the related accounts payable is not included in the calculation of Closing Net Working Capital Amount and shall include all checks and wire transfers and drafts deposited or available for deposit for the account of the Company and (z) exclude the PPP Loan Deposit Amount. For the avoidance of doubt, the calculation of the Closing Cash Amount shall be determined without giving effect to the transactions contemplated by this Agreement.

"Closing Date" shall have the meaning given in Section 1.6.

"Closing Indebtedness Amount" shall mean, as of immediately prior to Closing, the Indebtedness of the Company.

"Closing Net Working Capital Amount" shall mean the difference between (expressed as a positive or negative number) (a) (x) the Company's consolidated total current assets immediately prior to Closing (as defined by and determined in accordance with GAAP) less (y) the Company's consolidated total current liabilities (including accrued liabilities) immediately prior to Closing (as defined by and determined in accordance with GAAP) and (b) [***]. For purposes of calculating the Closing Net Working Capital Amount, (i) the Company's current assets shall exclude all cash and Tax assets and (ii) the Company's current liabilities shall exclude deferred revenue, all Indebtedness included in the final calculation of the Closing Indebtedness Amount and any Transaction Expenses included in the final calculation of Closing Transaction Expense Amount.

"Closing Transaction Expense Amount" shall mean the Transaction Expenses of the Company that have not been fully and finally satisfied as of the Closing.

"COBRA" shall mean the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended.

"Code" shall have the meaning given in the recitals.

"Combination Transactions" shall have the meaning given in the recitals.

"Company" shall have the meaning given in the opening paragraph.

"Company Authorizations" shall have the meaning given in Section 3.14.

"Company Board" shall have the meaning given in the recitals.

"Company Board Resolutions" shall have the meaning given in Section 3.4(a).

- "Company Capital Stock" shall mean the Company Common Stock and the Company Preferred Stock.
- "Company Common Stock" shall mean shares of common stock, par value \$0.001 per share, of the Company.
- "Company Employee Plan" shall mean any plan, program, policy, practice, contract, agreement or other arrangement providing for compensation, severance, termination pay, deferred compensation, performance awards, stock or stock related awards, welfare benefits, fringe benefits or other employee benefits or remuneration of any kind, including each Employee Agreement, whether written, unwritten or otherwise, funded or unfunded, including each "employee benefit plan," within the meaning of Section 3(3) of ERISA (whether or not subject to ERISA) that is maintained, sponsored, contributed to, or required to be contributed to, by the Company or any ERISA Affiliate for the benefit of any Employee, or with respect to which the Company or any ERISA Affiliate has or may have any Liability.
 - "Company Equity Plan" shall mean the Shareablee, Inc. 2013 Stock Option/Stock Issuance Plan.
 - "Company Indemnitees" shall have the meaning given in Section 5.3(a).
- "Company Indemnitors" shall mean (a) when used with respect to the Deferred Amount, the Deferred Amount Holders and (b) when used in any other context, the Shareablee Holdco Stockholders and Shareablee Holdco Warrantholder.
 - "Company Indemnitors Deductible" shall have the meaning given in Section 7.3(a).
 - "Company IP" shall have the meaning given in Section 3.11(a)(i).
- "Company Option" shall mean an option (whether or not vested) to purchase or otherwise acquire shares of Company Capital Stock that is outstanding and unexercised immediately prior to the Shareablee Merger Effective Time.
- "Company Optionholder" shall mean any Person holding any Company Option that is outstanding and unexercised immediately prior to the Shareablee Merger Effective Time.
 - "Company Owned IP" shall have the meaning given in Section 3.11(a)(ii).
 - "Company Parties" shall have the meaning given in Article III.
- "Company Preferred Stock" shall mean, collectively, the Company Series A Preferred Stock and the Company Series A-2 Preferred Stock.
 - "Company Products" shall have the meaning given in Section 3.11(a)(iii).
 - "Company Registered IP" shall have the meaning given in Section 3.11(b)(i).
- "Company RSU Award" shall mean an unvested restricted stock unit that is (i) subject solely to time-based vesting and (ii) payable in shares of Company Capital Stock.
- "Company RSU Awardholder" shall mean any Person holding any Company RSU Award that is outstanding immediately prior to the Shareablee Merger Effective Time.
- "Company Securities" shall mean all (a) options, warrants or other rights or Contracts, arrangement or commitments of any character relating to the capital stock of the Company, (b) shares of capital stock of or other voting securities or ownership interests in the Company, and (c) restricted shares, restricted share units, stock appreciation rights, performance shares, contingent value rights, "phantom" stock or similar securities or rights that are derivative of or provide economic benefits based, directly or indirectly, on the value or price of, any capital stock or other voting securities (including any bonds, debentures, notes or other indebtedness having voting rights or convertible into or exchangeable for securities having voting rights) or ownership interests in the Company.

- "Company Series A Preferred Stock" shall mean the Series A Preferred Stock of the Company, par value \$0.001 per share.
- "Company Series A Preferred Stockholder" shall mean any holder of Company Series A Preferred Stock immediately prior to the Shareablee Merger Effective Time.
- "Company Series A-2 Preferred Stock" shall mean the Series A-2 Preferred Stock of the Company, par value \$0.001 per share.
- "Company Series A-2 Preferred Stockholder" shall mean any holder of Company Series A-2 Preferred Stock immediately prior to the Shareablee Merger Effective Time.
 - "Company Software" shall have the meaning given in Section 3.11(a)(iv).
 - "Company Source Code" shall have the meaning given in Section 3.11(a)(y).
- "Company Stockholder" shall mean any holder of Company Capital Stock immediately prior to the Shareablee Merger Effective Time.
 - "Company Stockholder Indemnified Parties" shall have the meaning given in Section 7.2(b).
 - "Company Stockholder Released Claims" shall have the meaning given in Section 9.14(a).
- "Company Stockholder Releasing Party" and "Company Stockholder Releasing Parties" shall have the meaning given in Section 9.14(a).
- "Company Warrant" shall mean the warrant, dated March 25, 2015, by and among the Company and Silicon Valley Bank, to purchase or otherwise acquire shares of Company Common Stock that is outstanding and unexercised immediately prior to the Shareablee Merger Effective Time.
- "Company Warrantholder" shall mean the holder of the Company Warrant immediately prior to the Shareablee Merger Effective Time.
 - "Company Written Consent" shall have the meaning given in the recitals.
 - "Confidentiality Agreement" shall have the meaning given in Section 6.5.
 - "Confidential Information" shall have the meaning given in Section 6.5.
 - "Contaminant" shall have the meaning given in Section 3.11(a)(vi).
- "Continuing Employees" shall mean the current Employees of the Company who remain or become employees or contractors of Parent or any of its Subsidiaries (including the Final Surviving Company) immediately following the Final Merger Effective Time.
- "Contract" shall mean any written or oral agreement, contract, subcontract, settlement agreement, lease, sublease, instrument, permit, concession, franchise, binding understanding, note, option, bond, mortgage, indenture, trust document, loan or credit agreement, license, sublicense, insurance policy or other legally binding commitment or undertaking of any nature.
 - "Conversion Cap" shall have the meaning given in Section 2.3(c).
 - "Converted Option" and "Converted Options" shall have the meaning given in Section 2.3(b)(i).
 - "Converted RSU Award" and "Converted RSU Awards" shall have the meaning given in Section 2.3(b)(ii).
 - "COVID-19 Measures" shall have the meaning given in Section 3.19(j).

- "*Critical Representations*" shall mean the representations and warranties contained in Sections 3.11(d)(iii) (Data Partners), $\underline{3.11(j)}$ (Non-Infringement), $\underline{3.11(n)}$ (Privacy), $\underline{3.11(q)}$ (Security Measures), and $\underline{3.11(r)}$ (Effect of Transaction).
 - "Current Balance Sheet" shall have the meaning given in Section 3.7(b).
 - "Data Partners" shall mean each of the Persons set forth on Schedule C-1 (including clauses (a) and (b)).
- "*Deferred Amount*" shall mean, without interest and, if applicable, as may be adjusted in accordance with <u>Section 2.6(d)</u>, (a) if in cash, \$8,600,000, (b) if in shares, a number of shares of Parent Common Stock with an aggregate value equal to (i) \$8,600,000, *divided* by (ii) the Parent Stock Price as of the applicable date, or (c) if in a combination of cash and shares of Parent Common Stock, at Parent's sole discretion, as set forth in (a) and (b); *provided* that the value set forth in clause (b) (i) of this definition shall be reduced by the corresponding amount of cash paid in respect of the Deferred Amount; *provided further* that the Deferred Amount shall be reduced by any costs allocated to the Stockholder Representative (on behalf of the Deferred Holders) pursuant to <u>Section 2.6(d)</u>.
 - "Deferred Amount Holders" shall mean the As-Converted Holders and the Equity Award Holders.
 - "Deferred Amount Prohibition Period" shall have the meaning given in Section 5.4(d).
- "*Deferred Payroll Taxes*" shall mean any payroll Taxes in respect of a Pre-Closing Tax Period that are payable by the Company following the Closing by reason of a deferral under the CARES Act or the Payroll Tax Executive Order, calculated without giving effect to any Tax credits afforded under the CARES Act, the Families First Coronavirus Response Act (Public Law 116-127), the Payroll Tax Executive Order or any similar Law to reduce the amount of such Taxes.
 - "Determination Date" shall have the meaning given in Section 2.10(c).
 - "DGCL" shall have the meaning given in Section 1.1(a).
 - "Disclosure Schedule" shall have the meaning given in Article III.
 - "DLLCA" shall have the meaning given in Section 1.1(b).
 - "DOL" shall mean the United States Department of Labor.
 - "Dollars" or "\$" shall mean United States Dollars.
 - "Downward Adjustment Amount" shall have the meaning given in Section 2.10(e).
 - "Downward Adjustment Consideration" shall have the meaning given in Section 2.10(e).
 - "D&O Indemnifiable Matters" shall have the meaning given in Section 5.3(a).
 - "D&O Tail Policy" shall have the meaning given in Section 5.3(b).
 - "Embargoed Countries" shall have the meaning given in Section 3.22.
- "*Employee*" shall mean any current or former employee, consultant, advisor, independent contractor, officer, manager or director of the Company.
- "Employee Agreement" shall mean each management, employment, severance, separation, consulting, advisory, contractor, relocation, repatriation, expatriation, loan, visa, work permit or other written agreement, or contract (including, any offer letter or any agreement providing for acceleration of Company Options, Company RSU Awards, or Company Capital Stock subject to a right of repurchase in favor of the Company) between the Company and any Employee, and pursuant to which the Company has or may have any Liability.

- *"Employee Non-Competition and Non-Solicitation Agreement"* shall mean each Employee Non-Competition and Non-Solicitation Agreement executed and delivered concurrently with the execution of this Agreement by the Key Employees.
- "Environmental, Health and Safety Requirements" shall mean all applicable Laws concerning or relating to worker/occupational health and safety, or pollution or protection of the environment, including those relating to the presence, use, manufacturing, refining, production, generation, handling, transportation, treatment, recycling, transfer, storage, disposal, distribution, importing, labeling, testing, processing, discharge, release, threatened release, control or other action or failure to act involving cleanup of any hazardous materials, substances or wastes, chemical substances or mixtures, pesticides, pollutants, contaminants, toxic chemicals, petroleum products or byproducts, asbestos, polychlorinated biphenyls, noise or radiation.
 - "Equity Award Holders" shall mean the Company Optionholders and Company RSU Awardholders.
- "*Equity Award Shares*" shall mean a number, as of immediately prior to the Initial Merger Effective Time, equal to the total number of shares of Shareablee Holdco Common Stock issuable upon the exercise of the Shareablee Holdco Options and upon the settlement of the Shareablee Holdco RSU Awards.
- "*Equity Interests*" shall mean, with respect to any Person, any capital stock of, or other ownership, membership, partnership, joint venture or equity interest in, such Person or any indebtedness, securities, options, warrants, call, subscription or other rights or entitlements of, or granted by, such Person or any of its Affiliates that are convertible into, or are exercisable or exchangeable for, or giving any Person any right or entitlement to acquire any such capital stock or other ownership, partnership, joint venture or equity interest, in all cases, whether vested or unvested.
 - "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.
- "*ERISA Affiliate*" shall mean any entity that together with the Company would be deemed a "single employer" for purposes of Section 4001(b)(1) of ERISA and/or Sections 414(b), (c) and/or (m) of the Code.
 - "Escrow Payments" shall have the meaning given in Section 1.8(b).
 - "Estimated Adjustment Amounts" shall have the meaning given in Section 2.10(a).
 - "Estimated Closing Statement" shall have the meaning given in Section 2.10(a).
 - "Excess Escrow Payments" shall have the meaning given in Section 1.8(b).
- "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, including the rules and regulations promulgated thereunder.
 - "Exchange Agent" shall have the meaning given in Section 2.5(a).
- "Exchange Agent Agreement" shall mean that exchange agent agreement, dated as of December 16, 2021, entered into by and between Parent and the Exchange Agent.
- *"Exchange Documentation"* shall mean, with respect to each Company Stockholder and the Company Warrantholder, any documents (including a properly completed IRS Form W-9, or the appropriate version of IRS Form W-8, as applicable) that Parent or the Exchange Agent may reasonably require.
 - "Export Approvals" shall have the meaning given in Section 3.22.
 - "Family Group" shall have the meaning given in Section 5.4(d).
 - "Final Adjustment Amounts" shall have the meaning given in Section 2.10(b).
 - "Final Closing Statement" shall have the meaning given in Section 2.10(c).
 - "Final Merger" shall have the meaning given in Section 1.1(d).

- "Final Merger Certificate" shall have the meaning given in Section 1.5(a).
- "Final Merger Effective Time" shall have the meaning given in Section 1.5(a).
- "Final Surviving Company" shall have the meaning given in Section 1.1(d).
- "Financials" shall have the meaning given in Section 3.7(a).
- "FIRPTA Compliance Certificate" shall have the meaning given in Section 1.7(a)(y).
- "First Release Date" means March 15, 2023 (provided if on March 15, 2023, the Reduction Amount has not been finally determined pursuant to Section 2.6(d), the "First Release Date" shall mean the date that is seven (7) Business Days following final determination of the Reduction Amount pursuant to Section 2.6(d)); provided if Parent receives all of the Approved Data Consents prior to the Outside Delivery Date, the "First Release Date" shall be the one (1) year anniversary of the Closing Date.
 - "Forgiven PPP Loan Amount" shall have the meaning given in Section 1.8(b).
- "Fraud" shall mean, with respect to any Party, actual common law fraud (and not constructive fraud, negligent misrepresentation or negligent omission, or any form of fraud premised on recklessness or negligence) under the Laws of the State of Delaware.
- "*Fully Diluted Shares*" shall mean a number, as of immediately prior to the Initial Merger Effective Time, equal to (a) the As-Converted Shares, *plus* (b) the Equity Award Shares.
- "Fundamental Representations" shall mean (i) with respect to the Company Parties, the representations and warranties contained in Section 3.1(a) (Organization of the Company), Section 3.2 (Company Capital Structure), Section 3.3 (No Other Subsidiaries; Ownership Interests), Section 3.4 (Authority and Enforceability), Section 3.5(a)(i) (No Conflicts), Section 3.6 (No Prior Activities of Shareablee Holdco and Shareablee Merger Sub), and Section 3.18 (Brokers' and Finders' Fees) and (ii) with respect to the Parent Parties, the representations and warranties contained in Section 4.1 (Organization), Section 4.2 (Authority and Enforceability), Section 4.3 (Capitalization), Section 4.4(a) (No Conflicts), Section 4.8 (NASDAQ Compliance), and Section 4.9 (Brokers' and Finders' Fees).
 - "GAAP" shall mean United States generally accepted accounting principles.
- "Governmental Entity" means (including non-US) (a) any supranational, national, federal, state, county, municipal, local, or foreign government or any entity exercising executive, legislative, judicial, regulatory, taxing, or administrative functions of or pertaining to government, (b) any public international governmental organization or (c) any agency, division, bureau, department, or other political subdivision of any government, entity or organization described in the foregoing clauses (a) or (b) of this definition (including patent and trademark offices and self-regulatory organizations).
 - "Health Plan" shall have the meaning given in Section 3.19(h).
 - "Inbound Licenses" shall have the meaning given in Section 3.11(d)(i).
- "Indebtedness" of any Person at any time shall mean, without duplication: (a) all liabilities of such Person for borrowed money; (b) all obligations evidenced by bonds, debentures, notes (convertible or otherwise) or similar instruments, and all liabilities in respect of mandatorily redeemable or purchasable share capital or securities convertible into share capital; (c) all liabilities of such Person for the deferred purchase price of property or services (including any milestone, earnout or similar payments); (d) all Unpaid Pre-Closing Taxes (other than Transaction Payroll Taxes and withholding Taxes described in clause (d) of the definition of "Unpaid Pre-Closing Taxes"); (e) all liabilities in respect of any lease (or other arrangement conveying the right to use) which are required to be classified and accounted for under GAAP as capital leases; (f) all long-term liabilities in respect of deferred rent amounts payable under any lease agreement; (g) all liabilities of such Person for the reimbursement of any obligor on any letter of credit, banker's acceptance or similar credit transaction securing obligations of a type described in clauses (a), (b), (c), (d) or (e) above to the extent of the obligation secured; (h) all accrued interest, fees and prepayment penalties on the items described in clauses (a) through (g) above; (i) all obligations of such Person under swaps, collars, caps, hedges, derivatives of any kind

or similar instruments; (j) all guarantees by such Person of any liabilities of a third party of a nature similar to the types of liabilities described in the foregoing clauses (a), (b), (c), (d), (e), (f), (g), (h) or (i), to the extent of the obligation guaranteed; and (k) any unpaid bonus, commission, severance, accrued vacation, or nonqualified deferred compensation obligations accrued, due to or owing at or prior to the Closing (or solely as a result of the consummation of the Closing and the transactions contemplated by this Agreement) in respect of any Employee, together with the employer portion of any applicable FICA, state, local or foreign payroll Taxes or similar Taxes imposed on the Company in respect of any such payments described in this clause (k).

- "Indemnification Claim Notice" shall have the meaning given in Section 7.4(a).
- "Indemnification Claim Objection Notice" shall have the meaning given in Section 7.4(b).
- "Indemnified Party" and "Indemnified Parties" shall have the meaning given in Section 7.2(b).
- "*Indemnifying Party*" shall have the meaning given in <u>Section 7.4(a)</u>.
- "Independent Accountant" shall have the meaning given in Section 2.10(b).
- "Initial Merger" shall have the meaning given in Section 1.1(c).
- "Initial Merger Certificate" shall have the meaning given in Section 1.4(a).
- "Initial Merger Effective Time" shall have the meaning given in Section 1.4(a).
- "Initial Merger Surviving Corporation" shall have the meaning given in Section 1.1(c).
- "Initial Merger Surviving Corporation Common Stock" shall have the meaning given in Section 2.1(e).
- "Integrated Transaction" shall have the meaning given in the recitals.
- "Intellectual Property Rights" shall have the meaning given in Section 3.11(a)(vii).
- "Intended Tax Treatment" shall have the meaning given in the recitals.
- "Interested Party" shall have the meaning given in Section 3.13(a).
- "IRS" shall mean the U.S. Internal Revenue Service.
- "IT Systems" shall have the meaning given in Section 3.11(a)(viii).
- "Joinder Agreement" shall mean a Joinder Agreement in the forms agreed by the Parties.
- "Key Employees" shall mean Tania Yuki Mustapic, Greg Dale and Jon Lieberman.
- "Knowledge" (or any derivation thereof) shall mean, with respect to the Company, the actual knowledge of Tania Yuki Mustapic, Greg Dale, and Jonathan Lieberman (but solely with respect to the representations and warranties set forth in Section 3.11 (Intellectual Property)), in each case, after reasonable inquiry of such Person's direct reports, and with respect to the Parent Parties, the actual knowledge of Bill Livek, Mary Margaret Curry, and Ashley Wright, in each case, after reasonable inquiry of such Person's direct reports.
 - "Labor Agreements" shall have the meaning given in Section 3.19(1).
 - "Labor Organization" shall have the meaning given in Section 3.19(1).
- "Law" shall mean, with respect to any Person, any U.S. or non-U.S. federal, state, local, national, provincial, or other constitution, law, statute, ordinance, rule, regulation, published administrative position, policy or principle of common law, or any Order or award, in any case issued, enacted, adopted, promulgated, implemented or otherwise put into legal effect by or under the authority of any Governmental Entity applicable to such Person.

- "Lease Agreements" shall have the meaning given in Section 3.10(b).
- "Leased Real Property" shall have the meaning given in Section 3.10(b).
- "Legal Proceeding" shall mean any action, suit, charge, complaint, litigation, arbitration, grievance, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), hearing, inquiry, audit, investigation examination or other similar proceeding by or before, or otherwise involving, any court or other Governmental Entity or arbitrator or arbitration body.
 - "Letter of Transmittal" shall have the meaning given in Section 2.5(a).
- "Liability" shall mean, with respect to any Person, all debts, liabilities, commitments and obligations of any kind (whether known or unknown, contingent, accrued, due or to become due, secured or unsecured, matured or otherwise), including accounts payable, royalties payable, and other reserves, accrued bonuses, accrued vacation, employee expense obligations and all other liabilities of such Person or any of its Subsidiaries, including those arising under applicable Law or any Legal Proceeding or any Order of a Governmental Entity and those arising under any Contract, regardless of whether such liabilities are required to be reflected on a balance sheet in accordance with GAAP, including, with respect to the Company, Transaction Expenses incurred by the Company on or prior to the Shareablee Merger Effective Time.
 - "Liability Claim" shall have the meaning given in Section 7.4(a).
 - "Lien" shall mean any lien, pledge, charge, claim, mortgage, security interest or other encumbrance of any sort.
 - "*Litigation Conditions*" shall have the meaning given in <u>Section 7.5</u>.
- "Losses" shall mean any and all deficiencies, judgments, settlements, losses, damages, interest, fines, penalties, Taxes, costs and expenses (including reasonable out-of-pocket legal, accounting and other costs and expenses of professionals incurred in connection with investigating, defending, settling or satisfying any and all demands, claims, actions, causes of action, suits, proceedings, assessments, judgments or appeals, and in seeking indemnification, compensation or reimbursement therefor); provided, that "Losses" shall not include any lost profits or special, consequential, exemplary or punitive damages (except to the extent (a) any such lost profits or special, consequential, exemplary or punitive damages are the natural, probable and reasonable result of the underlying event(s) or matter(s) or (b) paid or payable by a Parent Indemnified Party to a third party in connection with a third-party claim).
- "made available" shall mean that a correct and complete copy of the document (including any amendments, exhibits and schedules thereto) referenced in such statement has been posted in the electronic data site managed by the Company at docsend.com for the express purpose of facilitating the Combination Transactions prior to 5:00 p.m. (New York time) on the day that is three (3) Business Days prior to the Closing and retained at all times from the date of posting through Closing in such electronic data site.
 - "Material Contract" and "Material Contracts" shall have the meaning given in Section 3.12(b).
- "*Maximum Adjustment Consideration*" shall mean a number of shares of Parent Common Stock with an aggregate value equal to the Purchase Price Holdback Amount, *divided* by the Parent Stock Price as of the Closing Date.
- "Maximum Available Fully Diluted Shares Closing Consideration" shall mean a number of shares of Parent Common Stock equal to (a) (i) the Total Merger Consideration, minus (ii) the Series A-2 Preference Amount, plus (iii) the Aggregate Option Exercise Price, minus (iv) the Deferred Amount, minus (v) the Purchase Price Holdback Amount, plus/minus, (vi) the Estimated Adjustment Amounts, divided by (b) the Parent Stock Price as of the Closing Date. For the avoidance of doubt, the total amount of "Maximum Available Fully Diluted Shares Closing Consideration" issued at the Closing shall be reduced by the Warrant Exercise Price by virtue of Section 2.2(b).
 - "Notice of Adjustment Disagreement" shall have the meaning given in Section 2.10(b).
 - "Obligated Party" shall have the meaning given in Section 6.5.

- "Objection Notice" shall have the meaning given in Section 2.6(d)(ii).
- "OFAC" shall have the meaning given in Section 3.22.
- "Open Source Licenses" shall have the meaning given in Section 3.11(a)(ix).
- "Open Source Material" shall have the meaning given in Section 3.11(a)(ix).
- "Order" shall mean any order, judgment, injunction, ruling, edict, or other decree, whether temporary, preliminary or permanent, enacted, issued, promulgated, enforced or entered by any Governmental Entity.
- "Ordinary Course of Business" means, with respect to the Company, the ordinary course of business consistent with the Company's past practice; *provided*, that, deviations from such ordinary course of business consistent with the Company's past practice shall not be deemed outside the Ordinary Course of Business to the extent such deviations were reasonably necessary with respect to actions taken in response to the COVID-19 pandemic to protect the health and safety of the Employees and other individuals having business dealings with the Company.
 - "Outbound Licenses" shall have the meaning given in Section 3.11(d)(ii).
 - "Outside Delivery Date" shall have the meaning given in Section 2.6(b).
 - "Parent" shall have the meaning given in the opening paragraph.
 - "Parent Board" shall have the meaning given in the recitals.
 - "Parent Common Stock" shall mean shares of Common Stock of Parent, par value \$0.001 per share.
 - "Parent Holdco LLC" shall have the meaning given in the opening paragraph.
 - "Parent Indemnified Parties" shall have the meaning given in Section 7.2(a).
 - "Parent Merger Sub" shall have the meaning given in the opening paragraph.
- "*Parent Organizational Documents*" shall mean the Amended and Restated Certificate of Incorporation of Parent, dated July 2, 2007, as amended to date, the Amended and Restated Bylaws of Parent, dated May 30, 2018, and the Certificate of Designations of Series B Convertible Preferred Stock, dated March 10, 2021.
 - "Parent Parties" shall have the meaning given in Article IV.
 - "Parent Parties Deductible" shall have the meaning given in Section 7.3(a).
 - "Parent Plans" shall have the meaning given in Section 6.6(a).
 - "Parent Prepared Returns" shall have the meaning given in Section 5.2(a).
 - "Parent SEC Documents" shall have the meaning given in Section 4.7(a).
 - "Parent Stock Issuance" shall have the meaning given in the recitals.
- "Parent Stock Price" shall mean (a) when used with respect to the Closing Date, \$4.12 per share of Parent Common Stock and (b) when used with respect to any other date, the numerical average of the volume-weighted average trading price on the Nasdaq Global Select Market for one (1) share of Parent Common Stock for the ten (10) full trading days ending on and including the full trading day that is the last Business Day prior to the applicable date.
 - "Party" and "Parties" shall have the meaning given in the opening paragraph.
- "*Payroll Tax Executive Order*" shall mean the Presidential Memorandum on Deferring Payroll Tax Obligations in Light of the Ongoing COVID-19 Disaster, as issued on August 8, 2020 and including any

administrative or other guidance published with respect thereto by any Taxing Authority (including IRS Notice 2020-65).

- "*Pension Plan*" shall mean each Company Employee Plan that is an "employee pension benefit plan" within the meaning of Section 3(2) of ERISA.
- "*Per Fully Diluted Share Closing Consideration*" shall mean (a) the Maximum Available Fully Diluted Shares Closing Consideration, *divided* by (b) the Fully Diluted Shares.
- "Per Share Deferred Amount" shall mean an amount equal to (a) the Deferred Amount, divided by (b) the Fully Diluted Shares.
- "Per Share First Deferred Amount" shall mean, subject to Article VII, an amount equal to (a) with respect to each Deferred Amount Holder except for Tania Yuki Mustapic, fifty percent (50%) of the Per Share Deferred Amount, and (b) with respect to Tania Yuki Mustapic, one third (1/3) of the Per Share Deferred Amount.
- "Per Share Purchase Price Holdback Amount" shall mean (a) the Maximum Adjustment Consideration, divided by (b) the As-Converted Shares.
- "Per Share Second Deferred Amount" shall mean, subject to Article VII, an amount equal to (a) with respect to each Deferred Amount Holder except for Tania Yuki Mustapic, fifty percent (50%) of the Per Share Deferred Amount, and (b) with respect to Tania Yuki Mustapic, one third (1/3) of the Per Share Deferred Amount.
- "*Per Share Series A-2 Consideration*" shall mean (a) the Series A-2 Consideration, *divided* by (b) the number, as of immediately prior to the Initial Merger Effective Time, of issued and outstanding shares of Shareablee Holdco Series A-2 Preferred Stock.
- "*Per Share Third Deferred Amount*" shall mean, subject to <u>Article VII</u>, an amount equal to with respect to Tania Yuki Mustapic, one third (1/3) of the Per Share Deferred Amount.
- "Permitted Liens" shall mean (a) statutory liens for Taxes that are not yet due and payable or that are being contested in good faith through appropriate proceedings and for which adequate reserves have been established in the Financials in accordance with GAAP; (b) statutory liens to secure obligations to landlords, lessors or renters under leases or rental agreements; (c) deposits or pledges made in connection with, or to secure payment of, workers' compensation, unemployment insurance or similar programs mandated by applicable Law; (d) inchoate statutory liens in favor of carriers, warehousemen, mechanics and materialmen, to secure claims for labor, materials or supplies and other like liens; (e) any minor imperfection of title or similar liens, charges or encumbrances which individually or in the aggregate with other such liens, charges and encumbrances does not impair the value of or the ability to use or transfer the property subject to such lien, charge or encumbrance or the use of such property in the conduct of the Company's business; (f) non-exclusive licenses of Intellectual Property Rights made in the Ordinary Course of Business and (g) any liens created in favor of the PPP Lender under the PPP Loan Escrow Agreement.
- "*Person*" shall mean an individual or entity, including a partnership, a limited liability company, a corporation, an association, a joint stock company, a trust, a joint venture, an unincorporated organization, or a Governmental Entity.
 - "*Personal Information*" shall have the meaning given in Section 3.11(a)(x).
 - "Personnel Agreements" shall have the meaning given in Section 3.11(f)(i).
 - "Post-Closing Statement" shall have the meaning given in Section 2.10(b).
 - "PPACA" shall have the meaning given in Section 3.19(h).
 - "PPP Lender" shall have the meaning given in Section 1.8(a).

- "PPP Loan" shall mean the loan issued to the Company pursuant to that certain note, dated as of May 1, 2020, in the original principal amount of \$833,202, entered into by and between the Company and the PPP Lender in connection with the U.S. Small Business Administration's Paycheck Protection Program.
 - "PPP Loan Deposit Amount" shall have the meaning given in Section 1.8(a).
 - "PPP Loan Escrow Agreement" shall have the meaning given in Section 1.8(a).
- "*Pre-Closing Tax Period*" shall mean any taxable period ending on or prior to the Closing Date and the portion of any Straddle Period ending on the Closing Date.
 - "Privacy and Security Requirements" shall have the meaning given in Section 3.11(a)(xi).
 - "Privacy Statements" shall have the meaning given in Section 3.11(a)(xii).
- "Pro Rata Share" shall mean, with respect to a particular Company Stockholder or Shareablee Holdco Stockholder, Company Warrantholder or Shareablee Holdco Warrantholder, or Company Indemnitor, (a) when used in the context of the Deferred Amount or the Deferred Amount Holders, a fraction: (i) whose numerator is the aggregate number of outstanding Fully Diluted Shares held by such person as of immediately prior to the Initial Merger Effective Time (and excluding, for the avoidance of doubt, any shares cancelled pursuant to Section 2.1(b)), and (ii) whose denominator is the aggregate number of Fully Diluted Shares outstanding immediately prior to the Initial Merger Effective Time (and excluding, for the avoidance of doubt, any shares cancelled pursuant to Section 2.1(b)), (b) when used in the context of the As-Converted Stockholders, a fraction: (i) whose numerator is the aggregate number of outstanding As-Converted Shares held by such person as of immediately prior to the Initial Merger Effective Time (and excluding, for the avoidance of doubt, any shares cancelled pursuant to Section 2.1(b)), and (ii) whose denominator is the aggregate number of As-Converted Shares outstanding immediately prior to the Initial Merger Effective Time (and excluding, for the avoidance of doubt, any shares cancelled pursuant to Section 2.1(b)) and (c) when used in any other context, a fraction (i) whose number is the aggregate consideration that such Person is entitled to receive under this Agreement (including the full amount of any Deferred Amount and ignoring the Adjustment Amounts), in its capacity as such and (ii) the Total Merger Consideration. For the avoidance of doubt the total of all Pro Rata Shares in any context shall equal one (1).
 - "Process," "Processing" and "Processed" shall have the meaning given in Section 3.11(a)(xiii).
 - "Prohibited Party Lists" shall have the meaning given in Section 3.22.
 - "Purchase Price Holdback Amount" shall mean an amount equal to \$500,000.
 - "Reduction Amount" shall mean (a) the Deferred Amount multiplied by (b) the Revenue Miss.
 - "Registered IP" shall have the meaning given in Section 3.11(a)(xiv).
- "Related Agreements" shall mean the Employee Non-Competition and Non-Solicitation Agreements, the Stockholder Non-Competition and Non-Solicitation Agreement, the Joinder Agreements, the Letters of Transmittal, and all other agreements and certificates entered into by the Company or the Company Stockholders in connection with the Closing and the transactions contemplated herein.
 - "Release Date" shall have the meaning given in Section 2.6(b).
 - "Released Parties" shall have the meaning given in Section 9.14(a).
 - "Representative Losses" shall have the meaning given in Section 7.6(b).
- "*Representatives*" shall mean, with respect to a Person, such Person's officers, directors, Affiliates, stockholders or employees, or any investment banker, attorney, accountant, auditor or other advisor or representative retained by any of them.
 - "Resolved Matters" shall have the meaning given in Section 2.10(b).

- "Restricted Cash" shall mean all cash of the Company that is not freely usable following the Closing because such cash is subject to restrictions or limitations on use or distribution by Law or Contract, or Taxes imposed on repatriation to the United States.
 - "Restrictive Legends" shall have the meaning given in Section 5.4(b).
- "Revenue" shall mean gross income of Shareablee LLC from all sources including the sale and license of goods and services, determined in accordance with Parent's normal and customary accounting methods, but subject to Schedule E and Schedule F, for the calendar year ended December 31, 2022.
- "*Revenue Policies and Procedures*" shall mean, for purposes of <u>Section 2.6(d)</u>, the Company's methods of revenue recognition prior to the Closing Date attached hereto as <u>Schedule F</u>.
 - "Revenue Hurdle" shall have the meaning given in Schedule E.
- "*Revenue Miss*" shall mean, expressed as a percentage, (a)(i) 100% *minus*, (ii)(I) Revenue *divided* by, (II) the Revenue Hurdle, *multiplied* by (b) 4, subject to <u>Schedule E</u>.
 - "SEC" shall mean the Securities and Exchange Commission.
 - "Section 1542" shall have the meaning given in Section 9.14(c).
 - "Securities Act" shall mean the Securities Act of 1933, as amended.
 - "Security Policies" shall have the meaning given in Section 3.11(q)(iii).
 - "Security Incident" shall have the meaning given in Section 3.11(q)(ii).
 - "Series A-2 Preference Amount" shall mean \$7,636,900.
- "*Series A-2 Consideration*" shall mean a number of shares of Parent Common Stock equal to (a) the Series A-2 Preference Amount, *divided* by (b) the Parent Stock Price as of the Closing Date.
 - "Shareablee Conversion Certificate" shall have the meaning given in Section 1.3(a).
 - "Shareablee Conversion Effective Time" shall have the meaning given in Section 1.3(a).
 - "Shareablee Holdco" shall have the meaning given in the opening paragraph.
- "Shareablee Holdco Capital Stock" shall mean the Shareablee Holdco Common Stock and the Shareablee Holdco Preferred Stock.
- "Shareablee Holdco Common Stock" shall mean shares of common stock, par value \$0.001 per share, of Shareablee Holdco.
- "Shareablee Holdco Common Stockholder" shall mean any holder of Shareablee Holdco Common Stock immediately prior to the Initial Merger Effective Time.
 - "Shareablee Holdco Options" shall have the meaning given in Section 2.3(a)(ii).
- "Shareablee Holdco Preferred Stock" shall mean, collectively, the Shareablee Holdco Series A Preferred Stock and the Shareablee Holdco Series A-2 Preferred Stock
 - "Shareablee Holdco RSU Awards" shall have the meaning given in Section 2.3(a)(ii).
- "Shareablee Holdco Series A Preferred Stock" shall mean the Series A Preferred Stock of Shareablee Holdco, par value \$0.001 per share.
- "Shareablee Holdco Series A Preferred Stockholder" shall mean any holder of Shareablee Holdco Series A Preferred Stock immediately prior to the Initial Merger Effective Time.

- "Shareablee Holdco Series A-2 Preferred Stock" shall mean the Series A-2 Preferred Stock of Shareablee Holdco, par value \$0.001 per share.
- "Shareablee Holdco Series A-2 Preferred Stockholder" shall mean any holder of Shareablee Holdco Series A-2 Preferred Stock immediately prior to the Initial Merger Effective Time.
 - "Shareablee Holdco Stockholder" shall mean any holder of Shareablee Holdco Capital Stock.
 - "Shareablee Holdco Warrant" shall have the meaning given in Section 2.2(a).
- "Shareablee Holdco Warrantholder" shall mean the holder of the Shareablee Holdco Warrant immediately prior to the Initial Merger Effective Time.
 - "Shareablee LLC" shall have the meaning given in Section 1.1(b).
 - "Shareablee LLC Conversion" shall have the meaning given in Section 1.1(b).
 - "Shareablee Merger" shall have the meaning given in Section 1.1(a).
 - "Shareablee Merger Certificate" shall have the meaning given in Section 1.2(a).
 - "Shareablee Merger Effective Time" shall have the meaning given in Section 1.2(a).
 - "Shareablee Merger Sub" shall have the meaning given in the opening paragraph.
 - "Shareablee Merger Surviving Corporation" shall have the meaning given in Section 1.1(a).
 - "Shrink-Wrap Software" shall have the meaning given in Section 3.11(a)(xvi).
 - "Standard Form Agreements" shall have the meaning given in Section 3.11(d)(iv).
- "Stockholder Non-Competition and Non-Solicitation Agreement" shall mean the Stockholder Non-Competition and Non-Solicitation Agreement executed and delivered concurrently with the execution of this Agreement by Tania Yuki Mustapic.
 - "Stockholder Representative" shall have the meaning given in the opening paragraph.
 - "Stockholder Representative Group" shall have the meaning given in Section 7.6(b).
 - "Straddle Period" shall mean any taxable period beginning on or prior to, and ending after, the Closing Date.
- "Subsidiary" shall mean, with respect to any party, any corporation or other organization or Person, whether incorporated or unincorporated, of which (a) such party or any other subsidiary of such party is a general partner (excluding such partnerships where such party or any subsidiary of such party does not have a majority of the voting interest in such partnership) or (b) at least a majority of the securities or other interests having by their terms ordinary voting power to elect a majority of the board of directors or others performing similar functions with respect to such corporation or other organization is directly owned or controlled by such party or by any one or more of its subsidiaries or affiliates.
 - "Suitability Documentation" shall mean the accredited stockholder questionnaire agreed upon by the Parties.
- "*Tax*" or, collectively, "*Taxes*" shall mean any and all taxes and charges, levies or other assessments in the nature of a tax, together with all interest, penalties and additions thereto, imposed by a Governmental Entity.
 - "Tax Contest" shall have the meaning given in Section 5.2(e).
 - "Tax Forms" shall mean a properly completed IRS Form W-9, or the appropriate version of IRS Form W-8, as applicable.

"Tax Representations" shall have the meaning given in Section 7.1.

"*Tax Return*" shall mean any return, declaration, notice, statement, claim for refund, report or form (including estimated Tax returns and reports, withholding Tax returns and reports, any schedules or attachments, and information returns and reports) or other document filed or required to be filed in connection with the determination, assessment, collection or administration of any Taxes, and including any amendment or supplement thereof.

"*Taxing Authority*" means, with respect to any Tax, the Governmental Entity or political subdivision thereof that imposes such Tax, and the agency (if any) charged with the collection of such Tax for such entity or subdivision, including any Governmental Entity or agency that imposes, or is charged with collecting, social security or similar charges or premiums.

"Third Party Claim" shall have the meaning given in Section 7.5.

"Third Party Data" shall have the meaning given in Section 3.11(a)(xvii).

"Total Merger Consideration" shall mean an amount equal to \$45,000,000.

"Transaction Expenses" shall mean (a) all costs, fees and expenses, including all legal, accounting, financial advisory, consulting and all other costs, fees and expenses of third parties (including Company Stockholders) incurred by or on behalf of the Company in connection with the negotiation, consummation or effectuation of the terms and conditions of this Agreement and the transactions contemplated hereby, including any payments made or anticipated to be made by the Company as a brokerage or finders' fee, agents' commission or any similar charge, in connection with the transactions contemplated by this Agreement; (b) any bonus, severance, change-in-control payments or other payment (including in lieu of any previously promised but ungranted equity award), or similar payment obligations of the Company that becomes due and payable in connection with the consummation of the transactions contemplated by this Agreement (other than any such payments that arise out of, or result form, any "double trigger" change of control obligations (i.e., obligations that have, as a second trigger, any termination effected by Parent during a specified period following the Closing)); (c) the cost of the D&O Tail Policy; and (d) any Transaction Payroll Taxes; (e) the costs and expenses payable to the Stockholder Representative as of the Closing and (f) 50% of the costs and expenses of outside counsel of Parent in connection with tax restructuring items.

"Transaction Payroll Taxes" shall mean the employer's portion of any employment, payroll or similar Taxes with respect to any bonuses, assumptions or conversion of Company Options or other payments, whether in cash or securities or other property, in respect of payments on Company Capital Stock (to the extent applicable), severance, change-in-control or other compensatory payments made in connection with the transactions contemplated by this Agreement, whether such Taxes relate to amounts paid or accrued at, prior to or following the Closing Date and whether payable by Parent, the Final Surviving Company or any of their respective Affiliates.

"*Transfer Prohibition Period*" shall have the meaning given in <u>Section 5.4(d)</u>.

"*Transfer Taxes*" shall mean any transfer, stamp, documentary, sales, use, registration, goods and services, harmonized sales, recording, filing, value-added and other similar Taxes incurred in connection with the transactions contemplated by this Agreement.

"Unpaid Pre-Closing Taxes" shall mean, without duplication, (a) any Taxes of the Company relating or attributable to any Pre-Closing Tax Period that are unpaid as of the Closing Date (including Deferred Payroll Taxes and any other Taxes in respect of circumstances existing or occurring prior to the Closing but for which payment is not required until after the Closing by reason of an election, agreement, deferral or other arrangement, or by operation of Law), (b) any Taxes of a Person for which the Company is liable (i) under U.S. Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or non-U.S. Tax Law) as a result of having been a member of an affiliated, consolidated, combined, unitary or similar group (including any arrangement for group or consortium relief or similar arrangement) before the Closing or (ii) as a result of an express or implied obligation to indemnify such Person, as a transferee or successor, by Contract, by operation of Law or otherwise as a result of an event or transaction occurring before the Closing, (c) any Taxes of any of the Company Stockholders and their respective Affiliates, and (d) any Taxes attributable to the transactions contemplated by this Agreement, including any Transaction Payroll Taxes, Transfer Taxes and withholding

Taxes in respect of any amounts payable pursuant to this Agreement. For this purpose, in the case of Taxes based on income, sales, proceeds, profits, receipts or similar items and all other Taxes that are not imposed on a periodic basis, the amount of such Taxes that have accrued through the Closing Date for a Straddle Period shall be deemed to be the amount that would be payable if the taxable year or period ended at the end of the day on the Closing Date based on an interim closing of the books, except that exemptions, allowances or deductions that are calculated on an annual basis (including depreciation and amortization deductions, other than with respect to property placed in service after the Closing) shall be allocated on a per diem basis. In the case of any other Taxes that are imposed on a periodic basis for a Straddle Period, the amount of such Taxes that have accrued through the Closing Date shall be the amount of such Taxes for the relevant period (or, in the case of such Taxes determined on an arrears basis, the amount of such Taxes for the immediately preceding period) *multiplied* by a fraction the numerator of which shall be the number of calendar days from the beginning of the period up to and including the Closing Date and the denominator of which shall be the number of calendar days in the entire period.

"Unresolved Matters" shall have the meaning given in Section 2.10(b).

"Upward Adjustment Amount" shall have the meaning given in Section 2.10(d).

"Upward Adjustment Consideration" shall have the meaning given in Section 2.10(d).

"WARN Act" shall mean the Worker Adjustment and Retraining Notification Act, 29 U.S.C. Section 2101 et seq. (1988), as amended, and any similar Law under national, provincial, state, or local law.

"Warrant Exercise Price" shall have the meaning given in Section 2.2(b).

"willful breach" shall have the meaning given in Section 8.2.

Annex A-16





Exhibit 99.1

Comscore Announces Acquisition of Leading Social Media & Technology Provider Shareablee

Integrated offering to create world's most complete view of consumer engagement and reach across digital platforms

RESTON, Va., December 17, 2021 – Comscore (NASDAQ: SCOR), a trusted partner for planning, transacting and evaluating media across platforms, today announced that it has acquired Shareablee, an industry-leading marketing analytics and measurement company in the social media space. With the acquisition, which closed on December 16, 2021, Comscore plans to expand its Media Metrix® and Video Metrix® currencies to include Shareablee's social media engagement and video insights, which will bridge the industry gap of traditional digital and social measurement services that exists today.

The combined audience insights of Comscore and Shareablee will give media companies, brands and advertisers an unmatched understanding of cross-platform consumer engagement and reach. Integration plans call for Comscore digital products to benefit from Shareablee data, allowing clients to broaden their digital footprint in products like Video Metrix® with advanced social video insights, and for Shareablee clients to have additional advantages with curated views of Comscore digital data embedded in their Shareablee dashboards. Comscore plans to retain the Shareablee team after closing, with key members of Shareablee management, including Tania Yuki, Greg Dale and Jonathan Lieberman, continuing in leadership roles.

Shareablee, a Meta for Media Measurement Partner, provides analytics and intelligence to many of the world's largest brands, publishers and agencies including Mondelez, GroupM, ESPN, NBCUniversal, and Vox Media. Shareablee has built a proprietary taxonomy of over twenty million publishers, advertisers and content creators in over seventy global markets that publish millions of videos, images and articles each day across the major social media platforms – delivering a vast array of performance-based insight that is a natural complement to Comscore's currency TV, CTV, OTT, movie and digital measurement solutions. Comscore and Shareablee have joint clients that will benefit from a true cross-platform view of their marketing reach and effectiveness.

"Comscore is committed to the future and helping customers have a comprehensive view of the consumer across platforms, while understanding the unique engagement on premium video. The future of measurement would not be complete without including the impressions delivered by the social media platforms," said Bill Livek, CEO and Executive Vice Chair, Comscore. "For marketers, this inclusion into our currencies will present great opportunity to maximize their outcomes on advertising investments. We see many opportunities to help our clients grow their viewing audiences and maximize their impact using social platforms."

"At our core, we believe that content matters, and that data and insights can make storytelling more impactful. We are thrilled to join forces with Comscore and look forward to finding new ways to jointly drive innovation," said Tania Yuki, CEO and Founder of Shareablee. "Our large-scale social measurement capabilities provide critical insight into changing content consumption and consumer behaviors, particularly when it comes to shedding light on Gen Z, and fit perfectly alongside Comscore's comprehensive cross-platform intelligence. Bringing the resources and measurement sophistication of Comscore to our product development capabilities will be very exciting for the whole market."

About Comscore

Comscore (NASDAQ: SCOR) is a trusted partner for planning, transacting and evaluating media across platforms. With a data footprint that combines digital, linear TV, over-the-top and theatrical viewership intelligence with advanced audience insights, Comscore allows media buyers and sellers to quantify their multiscreen behavior and make business decisions with confidence. A proven leader in measuring digital and TV audiences and advertising at scale, Comscore is the industry's emerging, third-party source for reliable and comprehensive cross-platform measurement.

About Shareablee

Shareablee is a leading strategic marketing intelligence platform for brands, content publishers and agencies looking to turn real-time data from the social media landscape into actionable insights for the business. Founded in 2013 by data scientist/entrepreneur Tania Yuki, Shareablee today counts many of the world's top brands as customers for its Data Explorer product suite. Shareablee's database tracks over 20 million brands, publishers and creators in over 70 countries, providing unprecedented scale and scope across all major social media platforms.

Cautionary Note Regarding Forward-Looking Statements

This press release contains forward-looking statements within the meaning of federal and state securities laws, including, without limitation, our expectations, forecasts, plans and opinions regarding Comscore's acquisition of Shareablee and related integration plans; evolving industry trends; management retention; and the impact of the acquisition on product development and innovation, client relationships, business prospects and future growth. These statements involve risks and uncertainties that could cause actual events to differ materially from expectations, including, but not limited to, changes in our integration plans and product roadmap; degradation of employee, customer, partner and vendor relationships; external market conditions and competition; evolving privacy and regulatory standards; the continuing impact of the Covid-19 pandemic and related government mandates; and our ability to achieve our expected strategic, financial and operational plans. For additional discussion of risk factors, please refer to our Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q, and other filings that we make from time to time with the U.S. Securities and Exchange Commission (the "SEC"), which are available on the SEC's website (www.sec.gov).

Investors are cautioned not to place undue reliance on our forward-looking statements, which speak only as of the date such statements are made. We do not intend or undertake, and expressly disclaim, any duty or obligation to publicly update any forward-looking statements to reflect events, circumstances or new information after the date of this press release, or to reflect the occurrence of unanticipated events.

Press

Neil Ripley Comscore, Inc. 646-746-0579 press@comscore.com

